## STRUCTURAL PROPERTIES AND ESTIMATION OF DELAY SYSTEMS

by

Raymond Hon-Sing Kwong

S.B., Massachusetts Institute of Technology (1971)

S.M., Massachusetts Institute of Technology (1972)

SUBMITTED IN PARTIAL FULFILLMENT OF THE

REQUIREMENTS FOR THE DEGREE OF

DOCTOR OF PHILOSOPHY

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY August, 1975

Signature of Author. Department of Electrical Engineering and Computer Science, August 11, 1975

Certified by....

Thesis Supervisor

. . . . . .

Accepted by..... Chairman, Departmental Committee on Graduate Students



Ъy

Raymond Hon-Sing Kwong

Submitted to the Department of Electrical Engineering and Computer Science on August 11, 1975 in partial fulfillment of the requirements for the Degree of Doctor of Philosophy

#### ABSTRACT

Two areas in the theory of delay systems are studied: structural properties and their applications to feedback control, and optimal linear and nonlinear estimation. First, we study the concepts of controllability, stabilizability, observability and detectability. The property of pointwise degeneracy of linear time-invariant delay systems is then considered. Necessary and sufficient conditions for three dimensional linear systems to be made pointwise degenerate by delay feedback are obtained, while sufficient conditions for this to be possible are given for higher dimensional linear systems. These results are then applied to obtain solvability conditions for the minimum time output zeroing control problem by delay feedback. Next, we turn our attention to optimal linear and nonlinear estimation. A representation theorem is given for conditional moment functionals of general nonlinear stochastic delay systems. From this, stochastic differential equations are derived for conditional moment functionals satisfying certain smoothness properties. We then give a complete solution to the estimation problem for general linear delay systems. Stability properties of the infinite-time linear optimal control system with quadratic cost and the optimal linear filter for systems without delays in the observations are studied. When appropriate structural properties hold, the optimal control system and the optimal filter are both shown to be asymptotically stable. Finally, the cascade of the optimal filter and the optimal control system is shown to be asymptotically stable as well.

THESIS SUPERVISOR: Alan S. Willsky TITLE: Assistant Professor of Electrical Engineering and Computer Science

#### ACKNOWLEDGEMENTS

I would like to express my gratitude to my thesis supervisor, Professor Alan Willsky, not only for his guidance and encouragement throughout my research, but also for his general interest in my education and future career. He is more than a supervisor; he is a good friend.

I also wish to express my sincere thanks to Professors Sanjoy Mitter and Pravin Varaiya. Their many helpful suggestions and discussions have been invaluable.

Thanks are also due to my friends and colleagues at M.I.T. for many useful discussions and for making my life as a graduate student a very enjoyable one.

I wish to thank Miss Elyse Wolf for her superb typing of the manuscript and Mr. Arthur Giordani for the excellent drafting.

Finally I would like to express my most sincere and grateful thanks to my wife Shirley for her encouragement and understanding.

This research was carried out at the Electronic Systems Laboratory with partial support provided by the National Science Foundation under grant NSF-GK-41647 and by the NASA Ames Research Center under grant NGL-22-009-124.

-3-

## TABLE OF CONTENTS

			Page	
ABSTRACT				
ACKNOWLEDGEMENTS				
TABLE OF CONTENTS				
CHAPTER 1:	INTRO	DDUCTION	6	
	1.1	Mathematical Description of Delay Systems	6	
	1.2	Dynamical Processes Modeled by Delay Systems	8	
	1.3	Outline of Thesis	12	
CHAPTER 2:	STRU	CTURAL PROPERTIES OF LINEAR TIME-INVARIANT Y SYSTEMS	16	
	2.1	Introduction	16	
	2.2	Controllability	17	
	2.3	Stabilizability	25	
	2.4	Observability	27	
	2.5	Detectability	32	
CHAPTER 3:	POIN	IWISE DEGENERACY OF LINEAR DELAY SYSTEMS	35	
	3.1	Introduction	35	
	3.2	Some Existing Results on the Pointwise Degeneracy of Linear Delay Systems	36	
	3.3	A Criterion for Pointwise Degeneracy for Linear Delay Systems	43	
CHAPTER 4: TDELAY FEEDBACK CONTROL OF LINEAR SYSTEMS				
	4.1	Introduction	56	
	4.2	Construction of the Matrix B in 3 Dimensional Pointwise Degenerate Systems	57	
	4.3	Delay Feedback Control of 3 Dimensional Linear Systems	64	
	4.4	Delay Feedback Control of Higher Dimensional Linear Systems	75	
	4.5	Sensitivity of Pointwise Degenerate Systems	84	
	4.6	Some Examples	88	

TABLE OF CONTENTS (CONT'D)

			Page	
CHAPTER 5:	OPTIMAL FILTERING OF LINEAR AND NONLINEAR STOCHASTIC DELAY SYSTEMS			
	5.1	Introduction	94	
	5.2	Existence, Uniqueness and Other Properties of Solutions of Stochastic Functional Differential Equations	97	
	5.3	A Representation Theorem for Conditional Moment Functionals	103	
	5.4	Stochastic Differential Equations for Non- Linear Filtering of Delay Systems	114	
	5.5	Optimal Filtering of Linear Stochastic Delay Systems	122	
CHAPTER 6:	STABILITY OF LINEAR OPTIMAL FILTERS AND CONTROL SYSTEMS			
	6.1	Introduction	131	
	6.2	Optimal Control of Linear Delay Systems with Quadratic Cost	132	
	6.3	Duality Between Estimation and Control for Linear Delay Systems	138	
	6.4	Stability of the Optimal Filter for Linear Systems with Delays in the Dynamics	144	
	6.5	Convergence of Estimation Error Covariance Operator	154	
	6.6	Stochastic Control of Linear Delay Systems and Stability	159	
	6.7	Filter Stability for Systems with Delays in the Observations	166	
	6.8	Perspective	173	
CHAPTER 7: CONCLUSIONS AND SUGGESTIONS FOR FUTURE RESEARCH				
APPENDIX A: PROOF OF LEMMAS 4.1 AND 4.2				
APPENDIX B: DERIVATION OF EQUATIONS (5.53) - (5.55)				
APPENDIX C: PROOF OF LEMMA 6.6.1				
REFERENCES				
BIOGRAPHICAL NOTE				

#### CHAPTER 1

#### INTRODUCTION

# 1.1 Mathematical Description of Delay Systems

The research reported in this thesis deals with certain aspects in the theory of delay systems. Delay systems constitute a class of hereditary systems, dynamical systems whose future behavior depends on past events in a fundamental way. Mathematically, the simplest delay system can be described by the following differential equation

$$\dot{x}(t) = f[x(t), x(t-\tau), t]$$
 (1.1)

where  $x(t) \in \mathbb{R}^{n}$ , and  $\tau$ , a positive constant, is a delay. A moment's reflection shows that in order to determine the behavior of the system for  $t \geq t_{o}$ , one must specify not only  $x(t_{o})$ , but the function  $x(\theta)$  on the interval  $[t_{o}^{-\tau}, t_{o}]$ . If we define the function

 $x_t: [-\tau, 0] \rightarrow R^n$ 

Ъy

$$x_{t}(\theta) = x(t+\theta) \qquad \qquad \theta \varepsilon[-\tau, 0] \qquad (1.2)$$

we see that knowledge of the function  $x_{\sigma}$  is necessary and sufficient to determine the behavior of the system for  $t \geq \sigma$ . Thus  $x_t$  is the true state of the system (1.1), and being an element of some function space defined on the interval  $[-\tau, 0]$ , it is infinite dimensional. This is a fundamental difference between delay systems and finite dimensional ordinary differential systems. By a slight abuse of language, we shall still call x(t) the "state" of the system (1.1), and call  $x_t$  the "complete state."

Equation (1.1) is often called a differential-difference equation of retarded type [1]. From the control theory point of view, we can have a controlled delay differential equation

$$\dot{x}(t) = f[x(t), x(t-\tau), u(t), u(t-\tau), t]$$
 (1.3)

If the system trajectory is not directly measured, but is observed through output variables z(t), one can have systems with delays in the observations

$$z(t) = h[x(t), x(t-\tau), t]$$
 (1.4)

In actual applications, a combination of the above basic types of equations may, of course, arise. Furthermore, these equations can also be generalized to include systems with several delays  $\tau_i$ , with time-varying delays  $\tau(t)$ , or with delays  $\tau(x(t))$  depending on the state.

So far, we have described delay systems which have a finite number of delays. A more general type of controlled delay systems is described by the retarded functional differential equation [2]

$$\dot{x}(t) = f[x_t, u(t), t]$$
 (1.5)

where  $x_t$  is defined by (1.2). This includes differential equations with point delays (1.1) and equations with distributed delays

$$\dot{\mathbf{x}}(t) = \int_{-\tau}^{0} f[\mathbf{x}(t+\theta)] d\theta + u(t)$$
(1.6)

While even more general types of delay systems can be defined [3], they will not play a role in this thesis and hence will not be discussed. Finally, delay systems may also be subject to random disturbances. One formulation of a stochastic delay equation is given by

$$dx(t) = f[x(t), x(t-\tau), t]dt + dw(t)$$
(1.7)

where w(t) is a Wiener process, and the equation is understood in the Ito sense. Usually associated with the stochastic delay equation (1.7) is an observation process

$$dz(t) = h[x(t), x(t-\tau), t]dt + dv(t)$$
(1.8)

As in the deterministic case, the function  $x_t$  plays an important role in the theory of stochastic delay systems.

#### 1.2 Dynamical Processes Modeled by Delay Systems

Euler was the first mathematician to study delay differential equations [4]. While there were a number of mathematical investigations of such equations after Euler, notably Volterra [5], the basic mathematical foundations for delay systems were established in the nineteen forties, fifties and early sixties [1], [6]. Since then, the theory of delay systems and, more generally, that of hereditary systems have been the subject of intensive investigation. While it has long been known by control theorists that delay systems are the appropriate models for a wide variety of process control systems [7], recent applications of system theory have shown that some biological, ecological, economical and social processes also take the form of delay systems [8], [9]. This has generated a great deal of interest

-8-

in various aspects of the theory of delay systems in the control as well as mathematical community. We shall discuss some examples of such processes.

## Example 1. Chemical Control Processes

A basic configuration in many chemical process control systems is the following [10]:



Here u usually represents a control of the feed rate of raw materials, T represents a transformation which changes the relative composition of the reactants. A typical example in refineries is that T is composed of a cascade of a reaction cooler, a decanter, and a distillation column. The recycle loop feeds back a certain percentage of the raw materials and some byproducts of the chemical reaction. This recycling introduces a significant time delay, often of the order of a few minutes, into the system. To build a suitable mathematical model for the analysis of such chemical control plants, a delay system must be used.

# Example 2. <u>Estimation in Radar-Sonar Problems</u> We consider the simplest model of a radar system [11]:



Because of the distance of the target, the received signal is a delayed (possibly distorted) version of the transmitted signal. If we assume the target is a slowly fluctuating point target, the reflection is linear, and there are K interfering targets, the received signal can be written in the form [11]

$$\mathbf{r}(t) = \sqrt{2} \operatorname{Re} \left\{ \left[ \sqrt{E}_{t} \sum_{i=0}^{K} \tilde{b}_{i} \tilde{f}(t-\tau_{i}) e^{j\omega_{i}t} + \tilde{n}(t) \right] e^{j\omega_{c}t} \right\}$$

where  $\tilde{f}(t)$  is the complex envelope of the transmitted signal,  $E_t$  the energy of the signal,  $\omega_c$  the carrier frequency,  $\tilde{n}(t)$  the complex representation of an additive Gaussian noise process, and for the i<sup>th</sup> target,  $\tilde{b}_i$  summarizes the amplitude information,  $\tau_i$  is the round trip delay time, and  $\omega_i$  is the Doppler shift frequency. These multipletarget situations are often referred to as communication over multipath channels. A basic estimation problem for these radar-sonar systems is to find optimal estimates for the quantities  $\tau_i$  and  $\omega_i$ ,  $i = 0, \ldots, K$ , since these give useful information about the ranges and velocities of the various targets. This is clearly an important application of the estimation theory of stochastic delay systems.

-10-

## Example 3. Modeling of Epidemics

Delay differential equations have often been used in the modeling of epidemics. Let  $x_1(t)$ ,  $x_2(t)$ ,  $x_3(t)$ , and  $x_4(t)$  denote the number of susceptible but unexposed individuals; infective individuals, removed individuals, and exposed but not infective individuals respectively. Since the infective individuals are those who were infected some time earlier, this introduces delays in the dynamics. The following equation has been proposed for the spread of measles [12]:

$$\dot{x}_{1}(t) = -\beta(t)x_{1}(t)[2\gamma + x_{1}(t-14) - x_{1}(t-12)] + \gamma$$

where  $\beta(t)$  is a proportionality coefficient that is seasonally dependent,  $\gamma$  is the (constant) rate at which individuals enter the population, and a unit of time is a day.

A more complex model of controlled epidemics has also been proposed [13]. Letting  $u_1$  represent active immunization,  $u_2$  passive immunization, the model proposed is

$$\dot{x}_{1}(t) = -\beta x_{1}(t) x_{2}(t) - x_{1}(t) u_{1}(t-\tau) - x_{1}(t) u_{2}(t)$$

$$\dot{x}_{2}(t) = a(t) - r(t) - x_{2}(t) u_{2}(t)$$

$$\dot{x}_{3}(t) = r(t)$$

$$\dot{x}_{4}(t) = k(t)\beta x_{1}(t) x_{2}(t)$$

Here  $\beta$  is an "effective contact" rate, a(t) the normalized arrival rate of infectives, r(t) the normalized removal rate of infectives, and  $\tau$  is a delay time.

Many more examples of processes modeled by delay, or more generally, hereditary systems can be found in [8] and [9].

## 1.3 Outline of Thesis

From the above discussion, it is clear that delay systems arise in many practical applications. This provides ample justification for studying the theory of delay systems in depth. There is a large literature on the existence and uniqueness of solutions to delay differential equations without control and their qualitative properties [2], [14]. However, from the system theory point of view, a large number of system-theoretic concepts remain to be explored and examined. In this thesis, we study two areas of interest: structural properties of delay systems and their applications, and the estimation of stochastic delay systems. In the first area, we are interested in seeing what are the roles played by the structural properties in control and estimation problems, and how properties peculiar to linear delay systems can be exploited to solve control problems not handled by finite dimensional linear systems. In the second area, we are motivated by the desire to develop an estimation theory suitable for multipath communication problems such as that described in Example 2 of this chapter. Of course, these by no means exhaust the theoretical implications or the practical applications of the ideas and results developed in our work. An outline of the thesis is given below.

In Chapter 2, we study the concepts of controllability, stabilizability, observability, and detectability in connection with

-12-

delay systems. These are properties of great interest in any study of dynamical systems. While they merit an in-depth study in their own right, our interest in them stems mainly from their relevance to the stability properties of the linear optimal control system with quadratic cost and the linear optimal filter. The various notions of controllability and observability are therefore examined and compared from the standpoint of these two problems. The results in this chapter will be used in a crucial way in Chapter 6.

While Chapter 2 deals with standard system-theoretic concepts, in Chapter 3 we study a property which is peculiar to delay systems: the notion of <u>pointwise degeneracy</u>. This property has been studied previously by various authors [15] - [20] as an interesting aspect in the theory of delay systems. Our viewpoint, on the other hand, is to apply it to the control of linear systems. This motivates the development of a useful characterization of systems which can be made pointwise degenerate by delay feedback. These results serve as the foundation for the construction of delay feedback controllers.

In Chapter 4, we discuss the potential applications of the pointwise degeneracy property in delay feedback control for the minimum time output zeroing problem for linear systems. The results of Chapter 3 are then applied to obtain conditions under which the control problem can be solved. The sensitivity of such a control system under perturbations of its parameters is also studied.

-13-

Starting with Chapter 5, we turn our attention to the filtering problem for stochastic delay systems. We first give a general discussion and show how previous results are inadequate for solving the filtering problem for delay systems. We next give a representation theorem for conditional moment functionals of nonlinear stochastic delay systems. Under suitable conditions, stochastic differential equations for the conditional moment functionals can be derived from the representation theorem. We then specialize these results to linear systems and give a complete solution to the filtering problem in this case. The similarities as well as differences between the optimal linear filter for delay systems and that for ordinary differential systems are discussed.

In Chapter 6, we study the stability properties of the linear optimal delay system with quadratic cost and the linear optimal filter. When appropriate system-theoretic properties of controllability, stabilizability, observability, and detectability hold, asymptotic stability of these optimal systems can be established. The striking duality between control and estimation is also demonstrated. This gives what we believe is the first instance other than the ordinary linear differential systems case where the control and the filtering problems can both be satisfactorily solved.

Finally, we summarize our findings in Chapter 7 and suggest some directions for future research. We shall see that many more theoretical problems associated with delay systems remain to be

-14-

solved. It is hoped that the research reported in this thesis will serve as a useful step in establishing a full-fledged theory for these systems.

#### CHAPTER 2

# STRUCTURAL PROPERTIES OF LINEAR TIME-INVARIANT DELAY SYSTEMS

#### 2.1 Introduction

In this chapter, we study four of the basic structural properties of linear time-invariant delay systems: controllability, stabilizability, observability, and detectability. In contrast to finite dimensional linear time-invariant systems, where simple necessary and sufficient for these properties to hold are known [21], [22], the situation in delay systems is considerably more complicated. For example, there is more than one meaningful notion of controllability or observability, and the one that is more appropriate depends on the application we have in mind. The property that has been studied most extensively is that of pointwise controllability [23] - [28] (also known as relative controllability or Euclidean space controllability). Necessary and sufficient conditions have been given for this property to hold, and various algebraic criteria have been devised [23] - [28]. Stabilizability and functional controllability have also been studied by some authors [23], [25], [29] - [34], but the results are far from complete. On the other hand, observability and detectability have hardly been touched upon in the literature [29]. Our motivation in studying these properties is to see how they can be utilized to solve the stability problems for the linear optimal control system and linear optimal filter (see Chapter 6). The appropriate notions of controllability and observability for these problems turn out to be different from previously given definitions, and we shall explore

-16-

the relations of these various notions. We make no claim to completeness or depth of our investigations here. In fact, a great deal of work is needed before these structural properties of delay systems can be clarified.

#### 2.2 <u>Controllability</u>

We shall be concerned with the controllability properties of the system

$$\dot{\mathbf{x}}(t) = A\mathbf{x}(t) + B\mathbf{x}(t-\tau) + C\mathbf{u}(t)$$
$$\mathbf{x}(\theta) = \phi(\theta) \qquad \theta \varepsilon[-\tau, 0] \qquad (2.1)$$

where A, B, C are nxn, nxn, and nxm matrices respectively. We shall denote the solution to (2.1) by  $x(t,\phi,u)$ . We shall also write  $x_t(\phi,u)$ for the complete state of the system (2.1) due to the control u, and it is defined by

$$\mathbf{x}_{t}(\phi,\mathbf{u})(\theta) = \mathbf{x}(t+\theta,\phi,\mathbf{u}) \qquad \theta \in [-\tau,0].$$

As indicated in Chapter 1, the state space of a delay system is a function space. Several choices for the state space are possible. The most common one is the space of continuous functions  $C([-\tau, 0]; \mathbb{R}^n)$ equipped with the sup norm which we will abbreviate by  $\mathscr{C}[2]$ . Other choices are  $M^2$  (which is isomorphic to  $\mathbb{R}^n x L^2$ ) [29] and  $W_2^{(1)}$ , the Sobolev space of absolutely continuous function on  $[-\tau, 0]$  with derivatives in  $L_2[-\tau, 0]$  [33]. Each has its own advantages and disadvantages. For a discussion of this, see [35]. In this thesis, we will usually take  $\mathscr{C}$  as our state space. Occasionally, we will discuss other spaces when  $\mathscr{C}$  proves to be inappropriate for the

problem at hand or when other choices shed light on the problem under consideration.

Since the state space is a function space, there are more than one notion of controllability that makes sense. The first systematic study of controllability for delay systems is that by Kirillova and Churakova [23] and is concerned with pointwise controllability. For simplicity, we will take our set of admissible controls U to be all  $R^{m}$ -valued piecewise continuous functions on some interval [0,T], which we shall denote by PC([0,T];  $R^{m}$ ). The same basic arguments can be used for U = L<sub>2</sub>([0,T];  $R^{m}$ ), the space of  $R^{m}$ -valued square integrable functions on [0,T].

<u>Definition 2.1</u> The delay system (2.1) is called pointwise controllable if for any  $\phi$  and any  $x_1 \in \mathbb{R}^n$ , there exists a time T and an admissible control u on [0,T] such that  $x(T,\phi,u) = x_1$ . We also speak of pointwise controllability on [0,T] or at time T if the terminal time T is fixed.

By the variation of constants formula [2], [36], the solution to (2.1) can be written as

$$\mathbf{x}(t,\phi,\mathbf{u}) = \Phi(t,0)\phi(0) + \int_{-\tau}^{0} \Phi(t,s+\tau)B\phi(s)ds + \int_{0}^{t} \Phi(t,s)Cu(s)ds$$
(2.2)

where  $\Phi(t,s)$  is the fundamental matrix associated with (2.1) and satisfies

-18-

$$\frac{d}{dt} \Phi(t,s) = A\Phi(t,s) + B\Phi(t-\tau,s)$$

$$\Phi(s,s) = I$$

$$\Phi(t,s) = 0 \qquad t \le s \qquad (2.3)$$

We will also write (2.2) as

$$x(t,\phi,u) = x(t,\phi,0) + F_t u$$

where for any fixed t  $\geq 0$ ,  $F_t: U \rightarrow R^n$ 

$$F_t^u = \int_0^t \Phi(t,s)Cu(s)ds$$

A straightforward argument shows that the following is true [37], [26].

Lemma 2.1 System (2.1) is pointwise controllable if and only if there exists a time T such that  $\mathscr{R}(F_T) = R^n$  where  $\mathscr{R}(X)$  of an operator X denotes the range. An equivalent condition is

rank 
$$\int_0^T \Phi(T,s)CC'\Phi'(T,s)ds = n \qquad (2.4)$$

In contrast to finite dimensional linear differential systems, the fundamental matrix  $\Phi(t,s)$  defined by (2.3) can be singular [17], [35]. Furthermore, pointwise controllability to any point in  $\mathbb{R}^n$  is <u>not</u> equivalent to pointwise controllability to the null vector in  $\mathbb{R}^n$ . To distinguish these cases, we make the following

<u>Definition 2.2</u> The delay system (2.1) is called pointwise null controllable if for any  $\phi$ , there exists a time T and an admissible

-19-

control u on [0,T] such that  $x(T,\phi,u) = 0$ .

It is clear that pointwise controllability implies pointwise null controllability. The converse, however, is not true. To discuss necessary and sufficient conditions for pointwise null controllability, we need the notions of pointwise completeness and pointwise degeneracy.

<u>Definition 2.3</u> System (2.1) is said to be pointwise complete if for each t, there exists a set of initial functions  $\phi_i^t \in \mathscr{C}$ , i = 1, ..., nsuch that the vectors  $x(t, \phi_i^t, 0)$  i = 1, ... n form a basis for  $\mathbb{R}^n$ . If the system is not pointwise complete, i.e., if there exists a proper subspace V of  $\mathbb{R}^n$  such that at some time t,  $x(t, \phi, 0) \subseteq V$  for all  $\phi \in \mathscr{C}$ , the system is said to be pointwise degenerate (at time t).

The relevance of these notions is shown by (see [26])

Lemma 2.2 If the system (2.1) is pointwise complete, then the condition

rank 
$$\int_{0}^{T} \Phi(T,s)CC'\Phi(T,s)ds = n$$

is necessary and sufficient for pointwise null controllability at time T.

The point is that if the system is pointwise degenerate, we do not need  $\mathscr{R}(F_T) = R^n$  for pointwise null controllability at T. We only need  $\mathscr{R}(F_T) \supset \{x(T,\phi,0): \phi \in \mathscr{C}\}$ , which is a <u>proper</u> subspace of  $R^n$ . The notion of pointwise degeneracy has other useful applications,

and we shall be studying it in much more depth in Chapters 3 and 4.

The condition (2.4) can be replaced by a number of algebraic criteria [23], [26], [28]. These can be thought of as generalizations of the well-known controllability results for finite dimensional linear systems. While pointwise controllability is useful in linear control problems with target sets in Euclidean space [35], [38], it is too weak for other problems. For control problems with target sets in function space, we need a much stronger notion of controllability. Here, following Banks et al., [33], we adopt the space  $W_2^{(1)}$ as our state space and  $L_2$  as our set of admissible controls U. The reason for switching the state space to  $W_2^{(1)}$  is that for  $u \in L_2$ and  $\phi \in W_2^{(1)}$ , eq. (2.1) implies that  $\dot{x}(t) \in L_2$ . Hence  $W_2^{(1)}$  is a natural choice for the state space when the control space is  $L_2$ . For further discussions on this point, see [33].

<u>Definition 2.4</u> The system (2.1) is said to be functionally controllable if for any functions  $\phi$  and  $\psi \in W_2^{(1)}$ , there exists a time T and a control  $u \in L_2[0,T]$  such that  $x_T(\phi, u) = \psi$ . If the terminal time T is fixed, we say that the system is functionally controllable on [0,T].

The following result of Banks, et al., [33] shows that functional controllability on [0,T] where  $T > \tau$  imposes extremely stringent conditions on the system which are rarely met in practice.

<u>Lemma 2.3</u> The system (2.1) is functionally controllable on [0,T]where T >  $\tau$  if and only if rank C = n.

-21-

Functional controllability is too strong for many control problems of interest, for example, the stability properties of linear optimal feedback systems and linear optimal filters. We now develop a new notion of controllability which is applicable to these problems. We revert to  $\mathscr C$  and piecewise continuous functions as our state space and control space.

Let us first define the function

$$W_{\pi}$$
: PC([0,T]; R<sup>n</sup>)  $\rightarrow$  C([T- $\tau$ ,T]; R<sup>n</sup>)

Ъy

$$(W_{T}v)(t) = \int_{0}^{T} \Phi(t,s)v(s)ds \qquad t \in [T-\tau,T]$$

Similarly, define

$$W_{T}C: U (= PC([0,T]; R^{m})) \rightarrow C([T-\tau,T]; R^{n})$$

by

$$(W_{T}Cu)(t) = \int_{0}^{T} \Phi(t,s)Cu(s)ds \qquad t \in [T-\tau,T]$$

Let us denote the space of all  $\mathbb{R}^{n}$ -valued functions  $\psi$  on [t,t+ $\tau$ ], which are of bounded variation on [t,t+ $\tau$ ] and continuous from the left on (t,t+ $\tau$ ), by B<sub>0</sub>([t,t+ $\tau$ ];  $\mathbb{R}^{n}$ ). We equip B<sub>0</sub>([t,t+ $\tau$ ];  $\mathbb{R}^{n}$ ) with the total variation norm on [t,t+ $\tau$ ], denoted by Var . For [t,t+ $\tau$ ]  $\psi \in B_{0}([t,t+\tau]; \mathbb{R}^{n}), \phi \in C([t-\tau,t]; \mathbb{R}^{n}),$  define the bilinear form  $\langle \psi, \phi \rangle_{t}$  by (see [2])

$$\langle \psi, \phi \rangle_{t} = \psi'(t)\phi(t) + \int_{t-\tau}^{t} \psi'(s+\tau)B\phi(s)ds$$
 (2.5)

We shall refer to  ${}^{\langle\psi}, \phi_t^{\rangle}$  as the hereditary product at time t. We adopt the following

<u>Definition 2.5</u> The system (2.1) is said to be controllable on [0,T] if there exists no  $\psi \in B_0([T,T+\tau]; \mathbb{R}^n)$  such that  $\langle \psi, \phi \rangle_T = 0$ for all  $\phi \in \widehat{\mathscr{R}(W_TC)}$ , but that  $\langle \psi, \phi_1 \rangle_T \neq 0$ , for some  $\phi_1 \in \widehat{\mathscr{R}(W_T)}$ .

This definition of controllability may not be very wellmotivated at the moment since  $PC([0,T]; R^n)$  is not the space of controls. It is obtained strictly as the dual notion of observability, which we will discuss in section 2.4. The physical meaning of  $\mathscr{R}(W_T^C)$ is clear and corresponds to functions which can be attained at T by suitable control. On the other hand, it is not clear to what  $\mathscr{R}(W_T)$ corresponds. However, this definition of controllability is easily seen to be stronger than pointwise controllability and weaker than functional controllability. For if we take t = T, and take  $\psi$  to be such that  $\psi(T) \neq 0$ ,  $\psi(s) = 0$ ,  $T < s \leq T + \tau$ , we get that

$$\langle \psi, \phi \rangle_{T} = \psi'(T) \phi(T)$$

Controllability then says that there exists no  $\psi(T)$  such that  $\psi'(T) \int_{0}^{T} \Phi(T,s)v(s)ds \neq 0$ , for some  $v \in PC([0,T]; \mathbb{R}^{n})$ , but that  $\psi'(T) \int_{0}^{T} \Phi(T,s)Cu(s)ds = 0$ ,

for all  $u \in PC([0,T]; \mathbb{R}^{m})$ . Since  $\Phi(T,s) = e^{A(T-s)}$  for  $s \in [T-\tau,T]$ , for arbitrary  $x_1 \in \mathbb{R}^{n}$ , we can define

$$v(t) = e^{A'(T-t)} \left[ \int_{T-\tau}^{T} e^{A(T-t)} e^{A'(T-t)} dt \right]^{-1} x_{1} \qquad t \varepsilon [T-\tau, T]$$
$$= 0 \qquad \qquad t \varepsilon [0, T-\tau)$$

This yields 
$$\int_{0}^{T} \Phi(T,s)v(s)ds = x_{1}$$
 and shows that  $\{\int_{0}^{T} \Phi(T,s)v(s)ds: v \in PC([0,T]; \mathbb{R}^{n})\} = \mathbb{R}^{n}$ . If  $\{\int_{0}^{T} \Phi(T,s)Cu(s)ds: u \in PC([0,T]; \mathbb{R}^{n})\}$ 

is a proper subspace of  $\mathbb{R}^n$ , then we can find  $\psi(T)$  such that

$$\psi'(T)\int_0^T \Phi(T,s)Cu(s)ds = 0,$$

for all u  $\in$  PC([0,T]; R<sup>m</sup>). Choosing v such that  $\int_0^T \Phi(T,s)v(s)ds = \psi(T)$ ,

we see that system (2.1) will not be controllable. Hence controllability implies  $\mathscr{R}(F_T) = R^n$  which is precisely pointwise controllability. On the other hand, if rank C = n, then  $\overline{\mathscr{R}(W_T)}$  is clearly the same as  $\overline{\mathscr{R}(W_TC)}$ , from which controllability follows. Hence functional controllability implies controllability.

Finally, there is a fourth notion of controllability, that of approximate functional controllability.

<u>Definition 2.6</u> System (2.1) is called approximately functionally controllable if for any  $\phi$  and  $\psi \in \mathscr{C}$ , there exists a time T and a sequence  $\{u_n\}$  in U such that  $x_T(\phi, u_n)$  converges to  $\psi$ .

Again approximate functional controllability is weaker than functional controllability, but stronger than pointwise controllability. Its relation to our definition of controllability is not known at present. Its use in delay system problems seems to be rather limited. However, it is one of the standard notions of controllability in infinite dimensional linear systems [39], and we have stated it for completeness.

-24-

## 2.3 <u>Stabilizability</u>

We start with the definitions of stability and asymptotic stability of unforced delay systems.

Consider the system

$$\dot{\mathbf{x}}(t) = \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{x}(t-\tau)$$
$$\mathbf{x}(\theta) = \phi(\theta) \qquad \theta \ \varepsilon[-\tau, 0] \qquad (2.6)$$

where  $\phi \in \mathscr{C}$ .

We shall denote the solution to (2.6) by  $x(t,\phi)$  and the complete state of (2.6) by  $x_{t}(\phi)$ .

<u>Definition 2.7</u> System (2.6) is said to be stable if for every  $\varepsilon > 0$ , there exists a  $\delta(\varepsilon) > 0$  such that  $|\phi| < \delta$  implies the solution of (2.6) satisfies  $|\mathbf{x}_{t}| < \varepsilon$ , for all  $t \ge 0$ .

<u>Definition 2.8</u> System (2.6) is said to be asymptotically stable if it is stable, and there is a  $\delta_0$  such that  $|\phi| < \delta_0$  implies the solution of (2.6) satisfies  $\lim_{t\to\infty} |\mathbf{x}_t(\phi)| = 0$ .

It is known that asymptotic stability of linear delay systems is equivalent to exponential stability [2], [36]. In other words, (2.6) is asymptotically stable if and only if there are positive constants K and  $\alpha$  such that  $|\mathbf{x}_{t}(\phi)| \leq \mathrm{Ke}^{-\alpha t} |\phi|$ ,  $t \geq 0$ , for all  $\phi \in \mathscr{C}$ .

We are now ready to give the definition of stabilizability.

<u>Definition 2.9</u> System (2.1) is said to be stabilizable if there exists a matrix function L:  $[-\tau, 0] \rightarrow R^{n \times n}$ , of bounded variation on  $[-\tau, 0]$ , such that the closed-loop system

$$\dot{\mathbf{x}}(t) = \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{x}(t-\tau) + \mathbf{C} \int_{-\tau}^{0} d\mathbf{L}(\theta)\mathbf{x}(t+\theta)$$
(2.7)

is asymptotically stable. We also say (A,B,C) is stabilizable to mean this definition.

As in the control of finite dimensional linear systems, stabilizability is crucial for the well-posedness of infinite-time control problems for delay systems (see Chapter 6). Necessary and sufficient conditions for stabilizability can be given in terms of controllability of a finite dimensional system obtained by projecting the complete state  $x_t$  onto the generalized eigenspaces associated with the eigenvalues with nonnegative real parts [30], [31]. This result is difficult to use, however, because it is necessary to compute the eigenvalues and find a basis for the generalized eigenspaces associated with them. One would like to have conditions expressed explicitly in terms of the system parameters A, B and C. Since the decomposition of the complete state  $x_t$  has been given in detail by Hale [2], this appears to be a hopeful, but still open task.

In finite dimensional systems, controllability is a sufficient condition for stabilizability. Since the controllability condition is very easy to check, this is an extremely useful sufficient condition. It is important to see, therefore, if some notion of controllability implies stabilizability in infinite dimensional linear

-26-

systems. This is an aspect of the theory of infinite dimensional linear systems that has received quite a bit of attention recently [40], [41]. We have seen in the above that for delay systems, there is indeed a notion of controllability that is necessary and sufficient for stabilizability. This notion, however, does not correspond to any of those discussed in section 2.2. It is worthwhile to give a few remarks concerning the relationships between those notions and that of stabilizability.

It is clear that functional controllability is sufficient for stabilizability, since functional controllability enables us to move x<sub>t</sub> to the zero function in finite time. Approximate functional controllability is also sufficient since this certainly implies that finite dimensional projections are controllable. On the other hand, pointwise controllability is not sufficient for stabilizability. An example illustrating this has been given by Morse [42]. The relationship between controllability in the sense of Definition 2.5 and stabilizability is not known.

#### 2.4 Observability

Here we are concerned with the system

$$\mathbf{x}(t) = A\mathbf{x}(t) + B\mathbf{x}(t-\tau)$$
$$\mathbf{x}(\theta) = \phi(\theta) \qquad \qquad \theta \ \varepsilon[-\tau, 0] \qquad (2.8)$$

$$z(t) = Cx(t)$$
  $t > 0$  (2.9)

We would like to know when it is possible to deduce the trajectory of the system x(t),  $0 \le t \le T$ , from the observations z(t),  $0 \le t \le T$ .

In contrast to controllability, pointwise observability has very little meaning. This is because knowledge of  $x(\sigma)$  is in general not sufficient in determining the evolution of the system for  $t \ge \sigma$ . We shall define two notions of observability.

<u>Definition 2.10</u> The system (2,8) - (2.9) is said to be observable if for any  $\phi$ , z(t) = 0,  $t \ge 0$ , implies x(t) = 0,  $t \ge 0$ . It is observable on [0,T] if z(t) = 0,  $0 \le t \le T$ , implies x(t) = 0,  $0 \le t \le T$ .

Since (2.8) - (2.9) are time-invariant, an equivalent definition is: z(t) = 0,  $t \ge s$ , implies x(t) = 0,  $t \ge s$ , any  $s \ge 0$ . It is useful to rephrase Definition 2.10 in terms of the initial function  $\phi$ .

<u>Definition 2.10</u>' The system (2.8) - (2.9) is said to be observable if there exists no function  $\phi \in \mathscr{C}$  such that z(t) = 0,  $t \ge 0$ , and such that x(t) is not identically zero for  $t \ge 0$ .

Let us express Definition 2.10' in terms of certain operators. By the variation of constants formula

$$z(t) = C\Phi(t,0)\Phi(0) + \int_{-\tau}^{0} C\Phi(t,s+\tau)B\Phi(s)ds \qquad (2.10)$$

Define the operator  $M_T: \mathscr{C} \to PC([0,T]; \mathbb{R}^n)$  by

$$(M_{T}\phi)(t) = \Phi(t,0)\phi(0) + \int_{-\tau}^{0} \Phi(t,s+\tau)B\phi(s)ds$$
 (2.11)

for any t  $\varepsilon$  [0,T]. Similarly, define  $CM_T: \mathscr{C} \rightarrow PC([0,T]; \mathbb{R}^m)$  by

$$(CM_{T}\phi)(t) = C\Phi(t,0)\phi(0) + \int_{-\tau}^{0} C\Phi(t,s+\tau)B\phi(s)ds$$
 (2.12)

for any t  $\varepsilon$  [0,T]. Then the system (2.8) - (2.9) is observable on [0,T] if and only if

$$N(M_{T}) = N(CM_{T})$$
(2.13)

where N(X) of an operator X denotes the nullspace of X. We will now prove the following duality result which relates the notions of controllability and observability.

Theorem 2.1 The delay system

 $\dot{\mathbf{x}}(t) = \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{x}(t-\tau) + \mathbf{C}\mathbf{u}(t)$ 

$$\mathbf{x}(\theta) = \phi(\theta) \qquad \theta \in [-\tau, 0]$$
 (2.14)

is controllable on [0,T] if and only if its hereditary adjoint system [2], [36]

$$\dot{\mathbf{y}}(t) = -\mathbf{A'}\mathbf{y}(t) - \mathbf{B'}\mathbf{y}(t+\tau)$$

$$\mathbf{y}(\theta) = \psi(\theta) \qquad \theta \ \varepsilon[\mathbf{T},\mathbf{T}+\tau], \ \psi\varepsilon\mathbf{B}_{o}([\mathbf{T},\mathbf{T}+\tau]; \ \mathbf{R}^{n}) \qquad (2.15)$$

$$\mathbf{z}(t) = \mathbf{C'}\mathbf{y}(t) \qquad (2.16)$$

is observable on [0,T].

<u>Remark</u> We shall usually simply say (2.15) - (2.16) is the adjoint or dual system to (2.14). Of course, the adjoint system (2.15) evolves backwards in time starting at t = T.

<u>Proof</u>: Applying the variation of constants formula to the adjoint system [2], [36], we obtain, for t  $\varepsilon$  [0,T]

$$y(t) = \Phi'(T,t)\psi(T) + \int_{T}^{T+T} \Phi'(s-\tau,t)B'\psi(s)ds$$

$$\stackrel{\Delta}{=} (H_{T}^{\star}\psi)(t) \qquad (2.17)$$

Hence observability of the adjoint system is equivalent to

$$N(C'H_T^*) = N(H_T^*)$$

Relative to the hereditary product, the adjoint of  $\overset{H\star}{\operatorname{T}}$  is given by

$$H_{T}: PC([0,T]; \mathbb{R}^{n}) \rightarrow C([T-\tau,T]; \mathbb{R}^{n})$$

$$\leq H_{T}^{*}\psi, v_{pc}^{*} = \leq \psi, H_{T}^{*}v_{T}^{*}, \text{ where } \leq u, v_{pc}^{*} = \int_{0}^{T} u'(t)v(t)dt.$$

We calculate

Hence 
$$(H_T v)(t) = \int_0^T \Phi(t,s)v(s)ds$$
  $t \in [T-\tau,T]$ 

The reader may note here that  $H_T = W_T$ , with  $W_T$  defined in section 2.2. Similarly, the adjoint of C'H<sup>\*</sup><sub>T</sub> is given by

$$H_{T}C: PC([0,T]; \mathbb{R}^{m}) \rightarrow C([T-\tau,T]; \mathbb{R}^{n})$$

$$(H_{T}Cv)(t) = \int_{0}^{T} \Phi(t,s)Cv(s)ds \qquad t\varepsilon[T-\tau,T]$$

(Note that  $H_T C = W_T C$ ). Now  $\psi \in N(H_T^*)$  if and only if

$$\langle H_{T}^{*}\psi, v \rangle_{pc} = 0$$
 for all  $v \in PC([0,T]; R^{n})$ ,

i.e.,  $\psi$  is in the annihilator of  $\overline{\mathscr{R}}(\mathrm{H}_{\mathrm{T}})$  relative to the hereditary product. Hence the condition N(H<sup>\*</sup><sub>T</sub>) = N(C'H<sup>\*</sup><sub>T</sub>) says that there exists no  $\psi$  such that

$$\langle \psi, \phi \rangle_{T} = 0 \quad \forall \phi \in \overline{\mathscr{R}(H_{T}C)}$$

but  $\langle \psi, \phi_1 \rangle_T \neq 0$  for some  $\phi_1 \in \overline{\mathscr{R}(H_T)}$ 

which is precisely the condition required for controllability of the original system.

This duality result will be used in a crucial way in Chapter 6. Since it is based on the use of the hereditary product, it is not the same as the functional analytic duality between the various operators and spaces involved. It would be interesting to see what duality result would be obtained if we use functional analytic adjoints for delay systems, as expounded in Hale [2].

Another definition of observability which is stronger than Definition 2.10 is that of strong observability.

<u>Definition 2.11</u> The system (2.8) - (2.9) is said to be strongly observable if z(t) = 0,  $t \ge 0$  implies the initial function  $\phi \equiv 0$ .

Clearly, if the system is strongly observable, it is observable. The converse is not true. Indeed, the matrix B must necessarily be nonsingular if strong observability is to hold. Otherwise we can choose an initial function  $\phi \in \mathscr{C}$  such that  $\phi(0) = 0$  and  $0 \neq \phi(s) \in N(B)$ ,  $s \in [-\tau, 0)$ . This yields z(t) = 0,  $t \ge 0$ , but  $\phi \ne 0$ .

By exactly the same arguments as in Theorem 2.1, we can show that the dual of strong observability is the following notion of controllability: there exists no  $\psi \in B_0([T,T+\tau]; \mathbb{R}^n)$  such that  $\langle \psi, \overline{\mathscr{R}(\mathbb{H}_T C)} \rangle_T = 0$ . Again B must necessarily be nonsingular. This may be taken as still another notion of controllability if desired.

#### 2.5 Detectability

As in finite dimensional linear systems, this is basically the dual notion of stabilizability.

<u>Definition 2.12</u> The system (2.8) - (2.9) is said to be detectable if there exists a matrix function K:  $[-\tau, 0] \rightarrow R^{n\times m}$ , of bounded variation on  $[-\tau, 0]$ , such that

$$\dot{\mathbf{x}}(t) = \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{x}(t-\tau) + \int_{-\tau}^{0} d\mathbf{K}(\theta)\mathbf{z}(t+\theta)$$
(2.18)

is asymptotically stable.

Theorem 2.2 The delay system (2.8) - (2.9) is detectable if and only if the adjoint system

$$\dot{y}(t) = 'A'y(t) - B'y(t+\tau) - C'u(t)$$
 (2.19)

is stabilizable (again this system runs backward in time).

<u>Proof</u>: Stabilizability of (2.19) requires the existence of a matrix function K':  $[-\tau, 0] \rightarrow R^{mxn}$ , of bounded variation on  $[-\tau, 0]$ , such that

$$-\dot{y}(t) = A'y(t) + B'y(t+\tau) + C' \int_{-\tau}^{0} dK'(\theta)y(t-\theta)$$
 (2.20)

is asymptotically stable. The right hand side of (2.20) defines a Stieltjes integral of the form

-32-

where 
$$\eta'(\theta) = \begin{cases} 0 & d\eta'(\theta)y(t-\theta) \\ C'K'(\theta) & \theta \ge 0 \\ -A'+C'K'(\theta) & -\tau < \theta < 0 \\ -A'-B'-C'K'(\theta) & \theta \le -\tau \end{cases}$$

The hereditary adjoint system to (2.20) is given by [2]

$$\dot{\mathbf{x}}(t) = \int_{-\tau}^{0} d\eta(\theta) \mathbf{x}(t+\theta)$$
(2.21)

Substituting for  $\eta(\theta)$ , we get that (2.21) is

$$\dot{\mathbf{x}}(t) = \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{u}(t-\tau) + \int_{-\tau}^{0} d\mathbf{K}(\theta) C\mathbf{x}(t+\theta)$$
$$= \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{x}(t-\tau) + \int_{-\tau}^{0} d\mathbf{K}(\theta) \mathbf{z}(t+\theta)$$
(2.22)

Since the stability properties of the adjoint system are the same as those for the original system [2], stability of (2.20) is equivalent to stability of (2.22). Hence the theorem follows.

The remarks following the proof of Theorem 2.1 concerning the use of adjoint systems also applies to this case.

Analogous to the situation for stabilizability, it would be interesting to see what notion of observability implies detectability. Presumably this would be the dual of that notion of controllability which implies stabilizability.

From the above discussion, it is clear that a great deal of work remains to be done in studying the concepts of controllability, stabilizability, observability, and detectability. We have merely

-33-

formulated definitions which appear to be useful in control and estimation problems, and pointed out some of the relationships between these concepts. Our discussion has not even exhausted the definitions of controllability and observability which have appeared in the literature. Motivated by the algebraic approach to finite dimensional linear systems, Kamen and Morse [42], [43] have given some algebraic definitions of controllability and observability which are useful in realization theory for delay systems. It is not clear what the physical interpretations of these definitions are, or what their relationships to our definitions are. We shall leave this entire area as a subject for future research.

## CHAPTER 3

#### POINTWISE DEGENERACY OF LINEAR DELAY SYSTEMS

## 3.1 Introduction

In Chapter 2, we studied some structural properties of linear delay systems and introduced the concept of pointwise degeneracy. In this chapter, we shall study the degeneracy property in much more depth, with a view towards applying it to feedback control. The degeneracy property of delay systems has recently attracted the attention of many researchers [15] - [20], partly because of its connections with other system-theoretic concepts (see Chapter 2), and partly because it is peculiar to delay equations and has no counterpart in ordinary differential equations. Popov [15] is the first to make a systematic study of the subject. His fundamental results have laid the foundations from which our results are developed. Since the publication of his paper, many other results have been obtained on pointwise degeneracy. It is fair to say, however, that this peculiar property is still far from being completely understood. Our viewpoint differs somewhat from these previous investigations in that we are primarily interested in applying the property in the construction of delay feedback controls. We do not therefore study pointwise degeneracy as an intrinsic property of delay systems. Rather, we study conditions under which a linear system can be made pointwise degenerate by delay feedback. Since we rely heavily on the techniques developed by Popov, in

-35-

Section 3.2, we shall first review his results and others which are germaine to our subsequent development. In Section 3.3, we give a characterization of systems which can be made pointwise degenerate by delay feedback. We shall then explore some of the implications of this characterization. This will prepare the way for the discussions in Chapter 4 where these results are used to obtain delay feedback controls for certain problems in linear system theory.

## 3.2 <u>Some Existing Results on the Pointwise Degeneracy of</u> Linear Delay Systems

We shall be concerned primarily with linear, constant delay systems of the form

$$\dot{x}(t) = Ax(t) + Bx(t-\tau)$$
 (3.1)

with an initial function  $\varepsilon \mathscr{C}$ . We first give an alternative definition of pointwise degeneracy of Eq. (3.1).

<u>Definition 3.1</u> The linear delay system (3.1) is called pointwise degenerate if there exist an n-vector  $q \neq 0$  and a number  $t_1 > 0$ such that every continuous function x:  $[-\tau, t_1] \rightarrow \mathbb{R}^n$  satisfying (3.1) in the open interval (0,  $t_1$ ) satisfies also  $q'x(t_1) = 0$ . If the system is not pointwise degenerate, it is called pointwise complete.

<u>Notation</u>. We will abbreviate pointwise degenerate by p.d. and pointwise complete by p.c. We will also say (A, B, q,  $\tau$ ) is p.d. at t<sub>1</sub> or (A, B,  $\tau$ ) is p.c. to mean the above definition. If the
time of pointwise degeneracy is of no concern, we will drop the qualification "at  $t_1$ " from our statements.

Definition 3.1 is readily seen to be equivalent to Definition 2.3 of pointwise degeneracy. However, Definition 3.1 has proved to be much more useful in obtaining results. We will adopt this definition of pointwise degeneracy from now on.

Let us note the following important consequence of Definition 3.1:

If (A, B, q,  $\tau$ ) is p.d. at  $t_1$ , then it will be p.d. on an interval of the form  $[t_1, \infty)$ , which we will call the degeneracy interval. To see this, take any number  $t_2 > t_1$ . For any continuous function  $\tilde{x}$ :  $[-\tau, t_2] \rightarrow \mathbb{R}^n$  satisfying (3.1) on (0,  $t_2$ ), define the shifted function  $\tilde{x}$ :  $[-\tau, t_1] \rightarrow \mathbb{R}^n$  by  $\tilde{x}(t) = x(t + t_2 - t_1)$ . Then  $\tilde{x}$  is continuous and satisfies (3.1) in (0,  $t_1$ ). Hence  $q'\tilde{x}(t_1) = q'x(t_2) = 0$ and x is p.d. at  $t_2$  also.

This simple fact is of central importance in the application of the p.d. property in delay feedback control, as will be explained later. In fact, a stronger result holds for the structure of the degeneracy interval. This will be given in Theorem 3.1.

Definition 3.1 admits a simple geometric interpretation. For each initial function  $\phi$ , there corresponds a solution of (3.1), which we will denote by  $x(t, \phi)$ . Then pointwise degeneracy means that there is a fixed vector q such that the solutions  $x(t, \phi)$  evolve

in a space perpendicular to q, for any choice of  $\varphi \in \mathscr{C}.$ 



The first example of a p.d. system was given by Popov [15], which has

$$A = \begin{bmatrix} 0 & 2 & 0 \\ 0 & 0 & -1 \\ 0 & 0 & 0 \end{bmatrix}$$
$$B = \begin{bmatrix} 0 & 0 & 0 \\ 1 & 0 & 0 \\ 0 & 2 & 0 \end{bmatrix}$$

q' = [1 -2 -1]

and  $\tau = 1$  with a degeneracy interval of  $[2, \infty)$ . This example sparked a surge of interest in the p.d. property. By now, much more is known about this example. We shall return to it in Chapter 4. Most of the basic results concerning pointwise degeneracy have been given by Popov. The following theorems are fundamental.

<u>Theorem 3.1</u> (Popov [15]). Assume that (3.1) is p.d. for the vector q. Then the largest set of points  $t_1$  at which (3.1) is p.d. for q is the interval [ $\ell \tau$ ,  $\infty$ ), where  $\ell$  is the smallest integer with the property

$$q'S(\sigma)^{\ell+1} = 0 \tag{3.2}$$

and  $S(\sigma)$  is the nxn polynomial matrix given by

$$(\sigma I - A)S(\sigma) = B \det(\sigma I - A)$$

Moreover, 2  $\leq$   $\ell$   $\leq$  n-1

<u>Theorem 3.2</u> (Popov [15]). The delay system (3.1) is p.d. for q at time  $t_1 > 0$  if and only if there exist an integer m > 0, k matrices  $P_j$ , mxn (k is the largest integer such that  $kT \le t_1$ ), an mxm matrix V, and an m-vector v such that

$$P_1 B = 0$$
 (3.3)

$$P_{j}A + P_{j+1}B = VP_{j}$$
  $j = 1, 2...k-1$  (3.4)

$$P_k A = V P_k$$
(3.5)

$$v'e^{VT}P_1 = 0$$
 (3.6)

$$v'e^{Vt}P_{j+1}-v'P_{j} = 0$$
  $j = 1, 2..., k-1$  (3.7)

$$\mathbf{v'P}_{\mathbf{k}} = \mathbf{q'} \tag{3.8}$$

Moreover, if the above quantities exist, one can always choose them so that rank  $(P_1 \dots P_k) = m$ .

From these theorems, one can deduce the following corollaries, some of which were obtained by other authors.

Corollary 3.1 (Popov [15]). System (3.1) is p.c. if rank B = 1.

Corollary 3.2. System (3.1) is p.c. if rank B = n.

<u>Proof</u>: If rank B = n, (3.3) implies  $P_1 = 0$ . Using (3.4) repeatedly, we immediately see that  $P_2$ , ...,  $P_k$  are all equal to zero, and the corollary follows.

This result was apparently established earlier by E.B. Lee.

<u>Corollary 3.3.</u> System (3.1) is p.c. if  $n \leq 2$ .

<u>Proof</u>: If n = 1, Corollary 3.1 immediately shows that (3.1) is p.c. Assume therefore n = 2. In this case, rank B is either 1 or 2. But then Corollary 3.1 and 3.2 together imply (3.1) is p.c.

This result was apparently established earlier by J.A. Yorke and J. Kato.

Although in the proof of Theorem 3.2, Popov gave a constructive procedure for finding the matrices  $P_1, \ldots, P_k$ , V, and the vector v, it is so complicated that in applications, Theorem 3.2 will be difficult to use. However, Popov was able to isolate a class of p.d. systems which have a remarkably simple structure. These are the so-called

-40-

### regular pointwise degenerate (r.p.d.) systems.

<u>Definition 3.2</u>. A p.d. system of the form (3.1) is called "regular" if the pair (A, B) is completely controllable and there exists an n-vector q such that the pair (q', A) is completely observable, and (3.1) is p.d. for q in the interval  $[2\tau, \infty)$ .

For such systems, the following theorem is true.

Theorem 3.3 (Popov [15]). Suppose (3.1) is r.p.d. with respect to q. Then there exists an nxn matrix Z such that

$$ZAZ = Z^2 A \tag{3.9}$$

$$q'Z^2 = 0$$
 (3.10)

$$q'Z = q'e^{AT}$$
 (3.11)

Equation (3.1) then takes the form

.

$$\dot{x}(t) = Ax(t) + (AZ-ZA)x(t-T)$$
 (3.12)

(that is, AZ-ZA = B). Conversely, every equation of the form (3.12), in which Z satisfies (3.9) - (3.11), is p.d. for q in  $[2\tau, \infty)$  (even though the other conditions for regularity may not be satisfied).

The following corollary can be deduced from Theorem 3.3.

Corollary 3.4. (Popov [15]). If B can be written as AZ-ZA where Z is given by

$$Z = rq'e^{AT}$$
(3.13)

and r is the solution to the equations

$$q'r = 1$$
 (3.14)

$$q'e^{AT}r = 0 \tag{3.15}$$

$$q'e^{AT}Ar = 0 (3.16)$$

then  $(A, B, q, \tau)$  is p.d. on  $[2\tau, \infty)$ .

Finally, Popov [15] has proved the following important result for 3 dimensional p.d. systems.

<u>Theorem 3.4</u>. Any 3 dimensional p.d. system is regular and can be written in the form (3.12) with Z given by (3.13) - (3.16).

For later reference, we emphasize here that our results on pointwise degeneracy and delay feedback are heavily based on Corollary 3.4 and Theorem 3.4. It is also worth pointing out that while (3.13) – (3.16) are considerably simpler than (3.3) – (3.8), they still involve the calculation of matrix exponentials. Given a triple  $(A,q,\tau)$  it is not obvious from (3.14) to (3.16) when a solution for r will exist. Our results given in section 3.3 will be concerned precisely with simple conditions on  $(A,q,\tau)$  under which a solution r to (3.14) – (3.16) exists.

Some additional properties of r.p.d. systems can be deduced from (3.9) - (3.12). For example, the Z matrix is in fact unique for regular p.d. systems. To see this, suppose  $Z_1$  and  $Z_2$  both satisfy (3.9) - (3.12). Then

$$q'(Z_1 - Z_2) = 0$$

-42-

Also

$$q'AZ_1 = q'(Z_1A + B) = q'Z_1A + q'B$$
  
=  $q'e^{AT}A + q'B$ 

this implies

$$q'A(Z_1 - Z_2) = 0$$

Similarly, it can be shown that

$$q'A^{1}(Z_{1} - Z_{2}) = 0$$
  $i = 0, ..., n-1$ 

By observability of (q', A),  $Z_1 = Z_2$  and uniqueness follows. It can also be shown that  $Z^{n-1} = 0$  and  $B^n = 0$  for r.p.d. systems. The significance of these properties are not known at the present time and we shall not pursue them further here.

We have summarized in the above some known results on pointwise degeneracy. We will apply these in our investigations in the next section.

## 3.3 A Criterion for Pointwise Degeneracy for Linear Delay Systems

Let us motivate our investigations by the following control problem (see also Popov [44]). Consider the linear control system

$$\dot{x}(t) = Ax(t) + u(t)$$
 (3.17)

$$y(t) = q'x(t)$$
 (3.18)

The objective is to find a linear state feedback law, possibly delayed with a fixed delay time  $\tau > 0$ , to drive the output y(t) to zero in

-43-

minimum time for all initial conditions, remaining zero thereafter. We shall call this minimum time output zeroing by delay feedback.

It is obvious that this problem cannot be solved by using instantaneous state feedback

$$u(t) = Kx(t)$$

However, if we can find a matrix B such that  $(A,B,q,\tau)$  is p.d. at 2 $\tau$ , then by definition of the p.d. property, y(t) = 0 for  $t \ge 2\tau$ . Furthermore, since  $2\tau$  is the smallest instant at which a delay system can be p.d. (see Theorem 3.1), this matrix B will be the solution to our minimum time control problem.

The first question to be examined concerning this approach is to find conditions on the matrix A, the vector q, and the delay time  $\tau$ , such that a matrix B will exist with the desired properties. Corollary 3.4 tells us that if we can find a solution r to (3.14) -(3.16), then on constructing the Z matrix as in (3.13) and setting B = AZ - ZA, we will obtain a p.d. system with degeneracy interval  $[2\tau, \infty)$ . Furthermore, Theorem 3.4 shows that for 3 dimensional systems, the above construction gives the unique B. Therefore, for 3 dimensional systems an equivalent problem is to find conditions on A, q, and T such that (3.14) - (3.16) admit a solution. We will exploit this in the proof of the following theorem.

<u>Theorem 3.5</u>. Given a 3x3 matrix A with eigenvalues  $\lambda_1$ ,  $\lambda_2$ ,  $\lambda_3$ , a 3-vector q, and a number  $\tau > 0$ , a matrix B with the property that

-44-

 $(A,B,q,\tau)$  is p.d. exists if and only if the following conditions are satisfied:

- (i) The geometric multiplicity [45] of each distinct eigenvalue of A is one;
- (ii) If there is a pair of complex eigenvalues, say  $\lambda_2 = \lambda_3^*$ ,  $\lambda_1$  real, then

 $\xi$  + e<sup>ητ</sup> (ηsinξτ - ξcosξτ)  $\neq$  0

where  $\eta = \lambda_1 - \operatorname{Re}\lambda_2$ ,  $\xi = -\operatorname{Im}\lambda_2$ ;

(iii) (q',A) is observable.

Before proving Theorem 3.5, we first state a lemma which is completely obvious. This lemma is not restricted to 3 dimensions.

Lemma 3.1. Suppose  $(A,B,q,\tau)$  is p.d. then for any nonsingular matrix F,  $(FAF^{-1}, FBF^{-1}, F'^{-1}q, \tau)$  is also p.d.

Lemma 3.1 implies that if we are free to choose B such that  $(A,B,q,\tau)$  is p.d., there is no loss of generality in assuming A to be in Jordan form. We exploit this in

<u>Proof of Theorem 3.5</u>: Let  $q' = [q_1 q_2 q_3]$ . We will apply the p.d. criterion established in Theorem 3.4. By the above remarks, we may assume, without loss of generality, that A is in Jordan form. Since n=3, the possible Jordan forms are as follows:

-45-

(I)  

$$J = \begin{bmatrix} \lambda_1 & 0 & 0 \\ 0 & \lambda_2 & 0 \\ 0 & 0 & \lambda_3 \end{bmatrix}$$

 $\lambda_{\mbox{i}}$  's may not be distinct, and there can only be a pair of complex eigenvalues.

(II)  

$$J = \begin{bmatrix} \lambda_1 & 1 & 0 \\ 0 & \lambda_1 & 0 \\ 0 & 0 & \lambda_2 \end{bmatrix}$$
or
$$\begin{bmatrix} \lambda_1 & 0 & 0 \\ 0 & \lambda_2 & 1 \\ 0 & 0 & \lambda_2 \end{bmatrix}$$

 $\lambda_1^{}$ ,  $\lambda_2^{}$  may not be distinct, but they are necessarily real.

(III)  

$$J = \begin{bmatrix} \lambda & \mathbf{1} & 0 \\ 0 & \lambda & \mathbf{1} \\ 0 & 0 & \lambda \end{bmatrix}$$

 $\lambda$  is necessarily real.

We first note that if there exists a solution r to (3.14) - (3.16), the vectors q', q'e<sup>AT</sup>, and q'e<sup>AT</sup>A are necessarily linearly independent (cf. Popov [15], p. 559). Theorem 3.4 then requires that

$$\det \begin{bmatrix} q' \\ q' e^{J\tau} \\ q' e^{J\tau} \end{bmatrix} \stackrel{\Delta}{=} \det W \neq 0$$

We consider each of these cases separately.

Case I. We require

$$det W = det \begin{bmatrix} q_{1} & q_{2} & q_{3} \\ \lambda_{1}\tau & \lambda_{2}\tau & \lambda_{3}\tau \\ q_{1}e^{-\lambda_{1}\tau} & q_{2}e^{-\lambda_{2}\tau} & q_{3}e^{-\lambda_{3}\tau} \\ q_{1}\lambda_{1}e^{-\lambda_{1}\tau} & q_{2}\lambda_{2}e^{-\lambda_{2}\tau} & q_{3}\lambda_{3}e^{-\lambda_{3}\tau} \end{bmatrix}$$
$$= q_{1}q_{2}q_{3}[(\lambda_{3}-\lambda_{2})e^{-(\lambda_{2}+\lambda_{3})\tau} + (\lambda_{1}-\lambda_{3})e^{-(\lambda_{1}+\lambda_{3})\tau} + (\lambda_{2}-\lambda_{1})e^{-(\lambda_{1}+\lambda_{2})\tau}] \neq 0$$

Clearly if any  $q_i = 0$ , then det W = 0 and (3.1) cannot possibly be degenerate for this q. We must also not have

$$(\lambda_{3}-\lambda_{2})e^{(\lambda_{2}+\lambda_{3})\tau} + (\lambda_{1}-\lambda_{3})e^{(\lambda_{1}+\lambda_{3})\tau} + (\lambda_{2}-\lambda_{1})e^{(\lambda_{1}+\lambda_{2})\tau} = 0$$
(3.20)

Dividing throughout by e  $(\lambda_2 + \lambda_3)^{\tau}$ , (3.20) becomes

$$(\lambda_3 - \lambda_2) + (\lambda_1 - \lambda_3) e^{(\lambda_1 - \lambda_2)\tau} + (\lambda_2 - \lambda_1) e^{(\lambda_1 - \lambda_3)\tau} = 0$$
(3.21)

Let  $\gamma_1 = \lambda_1 - \lambda_2$ ,  $\gamma_2 = \lambda_1 - \lambda_3$ , and let

$$f(\gamma_{1},\gamma_{2}) = (\gamma_{1}-\gamma_{2}) + \gamma_{2}e^{\gamma_{1}T} - \gamma_{1}e^{\gamma_{2}T}$$
(3.22)

Our problem now is reduced to finding conditions under which  $f(\gamma_1, \gamma_2) \neq 0$ . We first consider the case where all the  $\lambda_i$ 's are real (so that  $\gamma_1$  and  $\gamma_2$  are also real). Without loss of generality, we may assume  $\gamma_1 \geq 0$ (i.e.,  $\lambda_1 \geq \lambda_3$ ). If  $\gamma_1 = 0$ , we have immediately  $f(\gamma_1, \gamma_2) = 0$ . Suppose  $\gamma_1 > 0$  is fixed. We plot  $\gamma_2(e^{\gamma_1 T} - 1)$  and  $\gamma_1(e^{\gamma_2 T} - 1)$  as functions of  $\gamma_2$ .  $f(\gamma_1, \gamma_2) = 0$  at the points where these two functions are equal. Note that at  $\gamma_2 = 0$ , both functions are zero, and at  $\gamma_2 = \gamma_1$ , they are equal.

$$\frac{d}{d\gamma_2} \left[ \gamma_2 \left( e^{\gamma_1 \tau} - 1 \right) \right] = e^{\gamma_1 \tau} - 1$$

and  $\frac{d}{d\gamma_2} [\gamma_1(e^{\gamma_2 \tau}-1)] = \gamma_1 \tau e^{\gamma_2 \tau}$ 

In the region where  $\gamma_2 \leq 0$ ,  $\gamma_1^{\tau e^{\gamma_2^{\tau}}} \leq \gamma_1^{\tau} < e^{\gamma_1^{\tau}} - 1$ , for  $\gamma_1 > 0$ . In this region, therefore

$$\gamma_2(e^{\gamma_1^{T}}-1) \leq \gamma_1(e^{\gamma_2^{T}}-1)$$

We can now see that the graphs of the two functions look like



The functions can only intersect at two points. But we already know that they do intersect at  $\gamma_2 = 0$  and  $\gamma_2 = \gamma_1$ . Hence these are the only two points at which they can be equal. So in the region  $\gamma_1 > 0$ ,  $f(\gamma_1, \gamma_2) = 0$  if and only if  $\gamma_2 = \gamma_1$  or  $\gamma_2 = 0$ . We can now conclude that if all the  $\lambda_i$ 's are real,  $f(\gamma_1, \gamma_2) = 0$  if and only if  $\gamma_1 = 0$ , or  $\gamma_2 = 0$ , or  $\gamma_1 = \gamma_2$ . In other words, det W = 0 if and only if either  $\lambda_1 = \lambda_2$ , or  $\lambda_2 = \lambda_3$ , or  $\lambda_3 = \lambda_1$ , or any of the  $q_i$ 's = 0.

Next we consider the case when there is a pair of complex eigenvalues, say,  $\lambda_2 = \lambda_3^*$ . This implies  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$  are necessarily distinct. With  $\gamma_1$  and  $\gamma_2$  defined as before, we have that  $\gamma_1 = \gamma_2^*$ . Define  $\eta = \lambda_1 - \operatorname{Re}\lambda_2$ ,  $\xi = -\operatorname{Im}\lambda_2$  (i.e.,  $\gamma_1 = \eta + i\xi$ ). Then

$$f(\gamma_{1},\gamma_{2}) = 2i\xi + (\eta-i\xi)e^{(\eta+i\xi)\tau} - (\eta+i\xi)e^{(\eta-i\xi)\tau}$$
  
= 2i\xi + e^{\eta\tau} {(\eta-i\xi)(\cos\xi\tau + i\sin\xi\tau) - (\eta+i\xi)(\cos\xi\tau - i\sin\xi\tau)}  
= 2i[\xi+e^{\eta\tau}(\eta\sin\xi\tau - \xi\cos\xi\tau)] (3.23)

 $f(\gamma_1, \gamma_2) = 0$  if and only if  $\xi + e^{\eta \tau}(\eta \sin \xi \tau - \xi \cos \xi \tau) = 0$ . Thus, if there is a pair of complex eigenvalues, we require precisely condition (ii) of Theorem 3.5, in addition to no  $q_i = 0$ . This completes the consideration of Case I.

Case II. Since the same arguments hold for either

$$J = \begin{bmatrix} \lambda_{1} & 1 & 0 \\ 0 & \lambda_{1} & 0 \\ 0 & 0 & \lambda_{2} \end{bmatrix} \quad \text{or} \quad \begin{bmatrix} \lambda_{1} & 0 & 0 \\ 0 & \lambda_{2} & 1 \\ 0 & 0 & \lambda_{2} \end{bmatrix}$$

-49-

we only discuss the former situation. In this case,

$$e^{JT} = \begin{bmatrix} e^{\lambda_{1}T} & \lambda_{1}T & 0 \\ e^{\lambda_{1}T} & 0 \\ 0 & e^{\lambda_{1}T} & 0 \\ 0 & 0 & e^{\lambda_{2}T} \end{bmatrix}$$

Thus

$$det W = det \begin{bmatrix} q_{1} & q_{2} & q_{3} \\ \lambda_{1}^{\tau} & \lambda_{2}^{\tau} & \lambda_{2}^{\tau} \\ q_{1}e^{\lambda} & (q_{1}^{\tau}+q_{2})e^{\lambda} & q_{3}e^{\lambda} \\ q_{1}\lambda_{1}e^{\lambda} & (q_{1}^{\tau}+q_{1}\lambda_{1}^{\tau}+q_{2}\lambda_{1})e^{\lambda} & q_{3}\lambda_{2}e^{\lambda} \\ q_{1}\lambda_{1}e^{\lambda} & (q_{1}^{\tau}+q_{1}\lambda_{1}^{\tau}+q_{2}\lambda_{1})e^{\lambda} & q_{3}\lambda_{2}e^{\lambda} \\ = q_{1}^{2}q_{3}e^{\lambda_{1}^{\tau}} [(\lambda_{2}^{\tau}-1-\lambda_{1}^{\tau})e^{\lambda_{2}^{\tau}}+e^{\lambda_{1}^{\tau}}]$$

after some calculations. Hence

det W  $\neq$  0 if and only if q<sub>1</sub>  $\neq$  0, q<sub>3</sub>  $\neq$  0

and  $(\lambda_2^{\tau-1-\lambda_1^{\tau})}e^{\lambda_2^{\tau}}e^{\lambda_1^{\tau}} \neq 0.$ 

Let 
$$\gamma = \lambda_1 - \lambda_2$$
. Then

$$(\lambda_2 \tau - 1 - \lambda_1 \tau) e^{\lambda_2 \tau} + e^{\lambda_1 \tau} = 0$$

if and only if

$$\gamma \tau + 1 = e^{\gamma \tau} \tag{3.24}$$

The only real solution to (3.24) is  $\gamma = 0$ . Hence det W  $\neq 0$  if and only if  $q_1 \neq 0$ ,  $q_3 \neq 0$ , and  $\lambda_1 \neq \lambda_2$ . The same arguments show that if

$$J = \begin{bmatrix} \lambda_1 & 0 & 0 \\ 0 & \lambda_2 & 1 \\ 0 & 0 & \lambda_2 \end{bmatrix}$$

-50-

det W  $\neq$  0 if and only if  $q_1 \neq 0$ ,  $q_2 \neq 0$  and  $\lambda_1 \neq \lambda_2$ . This completes the consideration of Case II.

Case III. In this case

$$e^{J} = \begin{bmatrix} e^{\lambda \tau} & \tau e^{\lambda \tau} & \frac{1}{2} \tau^{2} e^{\lambda \tau} \\ 0 & e^{\lambda \tau} & \tau e^{\lambda \tau} \\ 0 & 0 & e^{\lambda \tau} \end{bmatrix}$$

$$det W = det \begin{bmatrix} q_1 & q_2 & q_3 \\ q_1 e^{\lambda \tau} & (q_1 \tau + q_2) e^{\lambda \tau} & (q_1 \frac{\tau^2}{2} + q_2 \tau + q_3) e^{\lambda \tau} \\ q_1 \lambda e^{\lambda \tau} & (q_1 + \lambda q_1 \tau + \lambda q_2) e^{\lambda \tau} & (q_1 \tau + q_2 + \lambda q_1 \frac{\tau^2}{2} + \lambda q_2 \tau + q_3 \lambda) e^{\lambda \tau} \end{bmatrix}$$
$$= q_1^3 \frac{\tau^2}{2} e^{2\lambda \tau}$$

Thus det  $W \neq 0$  if and only if  $q_1 \neq 0$ .

If we now combine all three cases, it is clear that the eigenvalues  $\lambda_i$  are required to satisfy conditions (i) and (ii) of Theorem 3.5. Furthermore, if a pair (q', A) is observable, then (q'T<sup>-1</sup>, TAT<sup>-1</sup>) is also observable for any nonsingular T. Take T such that

$$TAT^{-1} = J$$

It is known that for observability of  $(q'T^{-1}, TAT^{-1})$ , the components of  $q'T^{-1}$  must satisfy precisely the same conditions imposed on the  $q_i$ 's in the above discussion [46]. Hence observability of (q', A) is also re-

quired for the solution of (3.14) - (3.16). Since conditions (i), (ii) and (iii) of Theorem 3.5 are necessary and sufficient for the solvability of (3.14) - (3.16), they are also necessary and sufficient for the existence of the matrix B. The proof is finished.

As an easy corollary of Theorem 3.5, we have

<u>Corollary 3.5</u>. Suppose n > 3. Given an nxm matrix A, an n-vector q, and a number  $\tau > 0$ , if at least three of the eigenvalues satisfy conditions (i) and (ii) of Theorem 3.5, and if (q', A) is observable, then there exists a B such that (A,B,q, $\tau$ ) is p.d. on [ $2\tau$ ,  $\infty$ ).

<u>Proof</u>: Without loss of generality, we may assume that A is in Jordan form and in fact the first three eigenvalues satisfy conditions (i) and (ii) of Theorem 3.5. By observability of (q', A), the first three components of q satisfy the conditions imposed in the proof of Theorem 3.5. If we now consider a solution r to (3.14) - (3.16) in the form of

	r <sub>1</sub>				
	r <sub>2</sub>				
r =	r <sub>3</sub>				(3.25)
	0 :				
	0	-			

a little thought shows that the equations to be satisfied for the solvability of  $r_1$ ,  $r_2$ , and  $r_3$  are precisely the same as those in Theorem 3.5. By that theorem, we can solve for  $r_1$ ,  $r_2$ , and  $r_3$  with the assumptions in this corollary. Thus a solution to (3.14) - (3.16) exists in the form (3.25) and the corollary follows.

We make several remarks to clarify the contents of Theorem 3.5 and Corollary 3.5.

<u>Remark 3.1</u>. In contrast to Theorem 3.5, Corollary 3.5 gives only a set of sufficient conditions for the existence of the desired B matrix. The reason for this is that for n > 3, solvability of (3.14) - (3.16) constitutes only a set of sufficient conditions, generally not necessary, for the existence of the B matrix. To obtain necessary and sufficient conditions in the style of Theorem 3.5 for n > 3 seems to be rather difficult and remains an open problem. Nevertheless, it is worthwhile to make a few comments on how Corollary 3.5 can be extended. For example, if A is 4x4 and has two distinct pairs of complex eigenvalues,  $\lambda_1 = \lambda_2^*$ ,  $\lambda_3 = \lambda_4^*$ , then condition (ii) of Theorem 3.5 cannot be satisfied since there are no real eigenvalues. However, in this case, the modified condition

$$\xi' + e^{\eta' \tau} (\eta' \sin \xi' \tau - \xi' \cos \xi' \tau) \neq 0$$

where  $\xi' = \operatorname{Re}_1^1 - \operatorname{Re}_3^1$ ,  $\eta' = \operatorname{Im}_1^1 - \operatorname{Im}_3^1$ , can be used instead. We can also relax the requirement that (q', A) be observable. On the other hand, Corollary 3.5 is easy to state and apply, and is sufficient for our purposes later in Chapter 4. We shall therefore be contented with giving Corollary 3.5 as it stands.

<u>Remark 3.2</u>. Theorem 3.5 and Corollary 3.5 can be given the following interpretation:

Given A, q, and  $\tau$ , and  $n \ge 3$ , there "almost always" exists a B such that (A,B,q, $\tau$ ) is p.d. on [2 $\tau$ ,  $\infty$ ). This is because observability

-53-

of (q', A) is a generic property of a pair (q', A) [47] (conditions (i) and (ii) are also generic properties of square matrices). Thus, while pointwise degeneracy of the delay system (3.1) is a singular property, the existence of a B for which (A,B,q, $\tau$ ) is p.d. on [2 $\tau$ ,  $\infty$ ) is a generic property of A, q, and  $\tau$ .

<u>Remark 3.3</u>. The singular cases where no B exists are basically related to the eigenvalues of A. For example, if in the 3x3 case, A is symmetric and has two equal eigenvalues, then no B exists regardless of what q and  $\tau$  are. Furthermore, if B is obtained as in Corollary 3.4, it will depend on the eigenvalues of A. This remark will be very useful when we apply Theorem 3.5 and Corollary 3.5 to construct delay feedback controllers in the next chapter.

<u>Remark 3.4</u>. If the matrix B is obtained in the manner described in Corollary 3.4, the eigenvalues associated with the delay equation (3.1), i.e., those values of  $\lambda$  such that

$$\det(\lambda I - A - e^{-\lambda T} B) = 0$$
(3.26)

are precisely the same as the eigenvalues of A. To see this, first note that (3.14) - (3.16) imply  $Z^2 = ZAZ = 0$ . We obtain successively

$$det(\lambda I-A-e^{-\lambda T}B) = det[\lambda I-A-e^{-\lambda T}(AZ-ZA)]$$
  
=  $det[\lambda I-A-e^{-\lambda T}AZ+e^{-\lambda T}ZA+e^{-2\lambda T}ZAZ)$   
=  $det[\lambda I+(e^{-\lambda T}Z-I)A(e^{-\lambda T}Z+I)]$   
=  $det(I-e^{-\lambda T}Z)(\lambda I-A)(I+e^{-\lambda T}Z)$   
=  $det(\lambda I-A)$ 

-54-

By a result of Henry [48], the spectrum of the infinitesimal generator of the semigroup for the delay equation is finite, and the range of the semigroup is finite dimensional. Kappel [20] has proved a number of interesting results for this class of p.d. systems.

<u>Remark 3.5</u>. The peculiar condition (ii) in Theorem 3.5 relates the eigenvalues of A to the delay time  $\tau$ . Indeed, if  $\eta = 0$ ,  $\xi = \frac{2\pi}{\tau}$ ,  $\xi + e^{\eta \tau} (\eta \sin \xi \tau - \xi \cos \xi \tau) = 0$ . This suggests that for degeneracy to occur, the period of the natural frequency of A should not "match up" with the delay time. A more precise interpretation of (ii) is not known.

Theorem 3.5 and Corollary 3.5 give relatively simple and transparent conditions for the existence of the B matrix. Only simple calculations are necessary to check the conditions. More important, they give us a hold on when singular situations will arise. As such, they form the basis for the development of delay feedback controllers investigated in the next chapter.

#### CHAPTER 4

#### DELAY FEEDBACK CONTROL OF LINEAR SYSTEMS

#### 4.1 Introduction

In the previous chapter, we developed a theorem which allows us, under certain conditions, to assert the existence of a matrix B such that (A, B, q, T) is pointwise degenerate. This was motivated by the control problem described at the beginning of section 3.3. In this chapter, we would like to generalize the formulation of the control problem to include an input matrix C. Specifically, we would like to solve the problem of minimum time output zeroing by delay feedback for the linear system

$$\dot{x}(t) = Ax(t) + Cu(t)$$
 (4.1)

$$y(t) = q'x(t)$$
 (4.2)

Let us consider the 3 dimensional case. If we just use delay feedback of the form

$$u(t) = Lx(t-\tau)$$

Theorem 3.5 will impose certain conditions on A, q and  $\tau$  for the existence of a suitable matrix L. Furthermore, if we construct a matrix B for which (A,B,q, $\tau$ ) is p.d., we must also have the range of B included in the range of C in order for L to exist. These are rather severe restrictions. On the other hand, since the eigenvalues of A play such an important role (see remark 3.3), we may be able to solve the problem of minimum time output zeroing if we can modify the eigenvalues of A. This suggests using a feedback law of the form

-56-

$$u(t) = Kx(t) + Lx(t-\tau)$$
 (4.3)

We shall see that this choice of the feedback law enables us to solve the minimum time output zeroing problem for (4.1) and (4.2) under rather mild conditions. We first study the 3 dimensional case, as the required notation is relatively simple, and the techniques used can be readily extended to higher dimensions. In section 4.2, we establish the form of the matrix B for which  $(A,B,q,\tau)$  is p.d. in terms of the parameters A, q and  $\tau$ . We then prove a theorem in section 4.3 which gives necessary and sufficient conditions for the solvability of the minimum time output zeroing problem in 3 dimensions. In fact, the proof of the theorem gives a constructive procedure for finding the desired matrices K and L. For greater clarity and completeness, a summary of the algorithm is given after the proof of the theorem. Next, the techniques are adapted to obtain a solution to the control problem in higher dimensions. Again, we summarize the algorithm for this case at the end of section 4.4. In section 4.5, we study the properties of the feedback control system under perturbation of its parameters. Finally, we give some examples which illustrate the theory.

## 4.2 <u>Construction of the Matrix B in 3 Dimensional</u> Pointwise Degenerate Systems

In the development of delay feedback control using the p.d. property, it is very useful to know the explicit form of B. We shall carry out the necessary calculations in this section. Using these results, we show that in Popov's example (see section 3.2),

-57-

degeneracy in fact occurs for a unique  $\tau$  and a unique q (up to scalar multiples).

<u>Notation</u>. Since all our p.d. systems will be constructed using Corollary 3.4, they will have a degeneracy interval of  $[2\tau,\infty)$ . From now on, the term p.d. will mean p.d. on the interval  $[2\tau,\infty)$ .

As in section 3.3, we consider three cases:

- (I) A has three real distinct eigenvalues
- (II) A has two repeated eigenvalues

	$^{\lambda}$ 1	1	0
J =	0	$\lambda_1$	0
	0	0	λ <sub>2</sub>

 $\boldsymbol{\lambda}_1$  and  $\boldsymbol{\lambda}_2$  are distinct and necessarily real.

(III) A has three repeated real eigenvalues and has a Jordan form  $J = \begin{bmatrix} \lambda & 1 & 0 \\ 0 & \lambda & 1 \\ 0 & 0 & \lambda \end{bmatrix}$ 

Throughout this section, we assume (q',A) to be observable. By Theorem 3.5, we know that there exists a B such that  $(A,B,q,\tau)$  is p.d.

Case I: Let T be the matrix which diagonalizes A, i.e.,

$$\mathbf{T}^{-1}\mathbf{A}\mathbf{T} = \begin{bmatrix} \lambda_1 & 0 & 0 \\ 0 & \lambda_2 & 0 \\ 0 & 0 & \lambda_3 \end{bmatrix} \stackrel{\Delta}{=} \Lambda$$

with  $\lambda$  's real and distinct. Equations (3.14) to (3.16) can be rewritten as

$$q'TT^{-1}r = 1$$
 (4.5)

$$q'Te^{\Lambda \tau}T^{-1}r = 0$$
 (4.6)

$$q'Te^{\Lambda T}\Lambda T^{-1}r = 0 \qquad (4.7)$$

Letting  $\alpha^{\prime}$  = q'T,  $\eta$  =  $T^{-1}r,$  we obtain

$$\alpha'\eta = 1 \tag{4.8}$$

$$\alpha' e^{\Lambda \tau} \eta = 0 \tag{4.9}$$

$$\alpha' e^{\Lambda \tau} \Lambda \eta = 0 \tag{4.10}$$

Letting  $\beta_i = \alpha_i \eta_i$ , i = 1,2,3, we can write (4.8) - (4.10) as

$$\beta_1 + \beta_2 + \beta_3 = 1$$
 (4.11)

$$\beta_{1}e^{\lambda_{1}T} + \beta_{2}e^{\lambda_{2}T} + \beta_{3}e^{\lambda_{3}T} = 0$$
 (4.12)

$$\beta_1 \lambda_1 e^{\lambda_1 \tau} + \beta_2 \lambda_2 e^{\lambda_2 \tau} + \beta_3 \lambda_3 e^{\lambda_3 \tau} = 0$$
(4.13)

Solving for  $\boldsymbol{\beta}_1,~\boldsymbol{\beta}_2,~\text{and}~\boldsymbol{\beta}_3,~\text{we obtain}$ 

$$\beta_{1} = \frac{(\lambda_{2} - \lambda_{3})e^{\lambda_{2}\tau}}{D}$$
(4.14)

$$\beta_{2} = \frac{(\lambda_{3} - \lambda_{1})e^{\lambda_{1}\tau}}{D}$$

$$\beta_{3} = \frac{(\lambda_{1} - \lambda_{2})e^{(\lambda_{1} + \lambda_{2} - \lambda_{3})\tau}}{D}$$
(4.16)

where  $D = (\lambda_3 - \lambda_1) (e^{\lambda_1 \tau} - e^{\lambda_2 \tau}) + (\lambda_1 - \lambda_2) e^{(\lambda_2 - \lambda_3) \tau} (e^{\lambda_1 \tau} - e^{\lambda_3 \tau})$  (4.17)

Note that D is always nonzero since the  $\lambda_i$ 's are real and distinct (see section 3.3).

Next, we construct a matrix  $\overline{Z}$ .

$$\begin{split} \overline{z} &= \eta \alpha' e^{\Lambda \tau} \\ &= \frac{1}{D} \begin{bmatrix} (\lambda_2 - \lambda_3) e^{(\lambda_1 + \lambda_2) \tau} & \frac{\alpha_1}{\alpha_2} (\lambda_2 - \lambda_3) e^{(\lambda_2 - \lambda_3) e^{(\lambda_1 + \lambda_3) \tau}}} \\ (\lambda_3 - \lambda_1) \frac{\alpha_1}{\alpha_2} e^{(\lambda_1 - \lambda_2) \tau} & (\lambda_3 - \lambda_1) e^{(\lambda_1 - \lambda_2) \tau} \\ (\lambda_1 - \lambda_2) \frac{\alpha_1}{\alpha_2} e^{(\lambda_1 + \lambda_2 - \lambda_3) \tau} & (\lambda_1 - \lambda_2) \frac{\alpha_2}{\alpha_3} e^{(\lambda_1 + 2\lambda_2 - \lambda_3) \tau} & (\lambda_1 - \lambda_2) e^{(\lambda_1 + \lambda_2) \tau} \end{bmatrix} \end{split}$$

And 
$$B = \Lambda Z - Z\Lambda$$

$$= \frac{1}{D} \begin{bmatrix} 0 & \frac{\alpha_2}{\alpha_1} (\lambda_1 - \lambda_2) (\lambda_2 - \lambda_3) e^{2\lambda_2 T} & \frac{\alpha_3}{\alpha_1} (\lambda_1 - \lambda_3) (\lambda_2 - \lambda_3) e^{(\lambda_2 + \lambda_3)T} \\ \frac{\alpha_1}{\alpha_2} (\lambda_2 - \lambda_1) (\lambda_3 - \lambda_1) e^{2\lambda_1 T} & 0 & \frac{\alpha_3}{\alpha_2} (\lambda_2 - \lambda_3) (\lambda_3 - \lambda_1) e^{(\lambda_1 + \lambda_3)T} \\ \frac{\alpha_1}{\alpha_3} (\lambda_3 - \lambda_1) (\lambda_1 - \lambda_2) e^{(2\lambda_1 + \lambda_2 - \lambda_3)T} & \frac{\alpha_2}{\alpha_3} (\lambda_3 - \lambda_2) (\lambda_1 - \lambda_2) e^{(\lambda_1 + 2\lambda_2 - \lambda_3)T} & 0 \\ (4.19) \end{bmatrix}$$



$$\overline{Z} = \eta \alpha' e^{\Lambda \tau} = T^{-1} r q' T T^{-1} e^{\Lambda \tau} T$$
$$= T^{-1} Z T$$

and  $\overline{B} = \Lambda \overline{Z} - \overline{Z} \Lambda$ 

$$= T^{-1}ATT^{-1}ZT - T^{-1}ZTT^{-1}AT$$
$$= T^{-1}BT$$

Thus the desired matrix B is given by

$$B = T\overline{B}T^{-1}$$
(4.20)

We note that the range space of  $\overline{B}$  is spanned by the vectors

$$\begin{bmatrix} 0 \\ 1 \\ -\frac{\alpha_2}{\alpha_3} e^{(\lambda_2 - \lambda_3)\tau} \end{bmatrix} \text{ and } \begin{bmatrix} 1 \\ 0 \\ -\frac{\alpha_1}{\alpha_3} e^{(\lambda_1 - \lambda_3)\tau} \end{bmatrix}$$
(4.21)

Case II: Let T be the matrix which brings A to its Jordan form, i.e.,

$$\mathbf{T}^{-1}\mathbf{A}\mathbf{T} = \mathbf{J} = \begin{bmatrix} \lambda_1 & 1 & 0 \\ 0 & \lambda_1 & 0 \\ 0 & 0 & \lambda_2 \end{bmatrix}$$

Completely analogous arguments show that the desired matrix B is given by

$$B = T \begin{bmatrix} \lambda_{1}^{\tau} & [\lambda_{2}(\alpha_{1}^{\tau+\alpha_{2}})-\eta_{1}\alpha_{1}]e^{\lambda_{1}^{\tau}} & [(\lambda_{1}^{-\lambda_{2}})\eta_{1}\alpha_{3}^{+\eta_{2}}\alpha_{3}]e^{\lambda_{2}^{\tau}} \\ 0 & -\eta_{2}\alpha_{1}e^{\lambda_{1}^{\tau}} & [(\lambda_{1}^{-\lambda_{2}})\eta_{2}\alpha_{3}e^{\lambda_{2}^{\tau}} \\ (\lambda_{1}^{-\lambda_{2}})\eta_{2}\alpha_{3}e^{\lambda_{2}^{\tau}} \end{bmatrix} T^{-1} \\ (\lambda_{2}^{-\lambda_{1}})\eta_{3}\alpha_{1}e^{\lambda_{1}^{\tau}} & [(\lambda_{2}^{-\lambda_{1}})\eta_{3}(\alpha_{1}^{\tau+\alpha_{2}})-\eta_{3}\alpha_{1}]e^{\lambda_{1}^{\tau}} & 0 \end{bmatrix}$$

(4.22)

-

where  $\alpha' = q'T$  and the  $\eta_i$ 's satisfy

$$\eta_{1} = \frac{\left[\left(\lambda_{2} - \lambda_{1}\right)\tau + \frac{\alpha_{2}}{\alpha_{1}}\left(\lambda_{2} - \lambda_{1}\right) - 1\right]e^{\lambda_{2}\tau}}{\alpha_{1}^{D}}$$
(4.23)

$$n_2 = \frac{-(\lambda_2 - \lambda_1)e^{\lambda_2 \tau}}{\alpha_1 D}$$
(4.24)

$$\eta_3 = \frac{e^{\lambda_1 \tau}}{\alpha_3 D}$$
(4.25)

with D = 
$$(\lambda_2 - \lambda_1)\tau e^{\lambda_2 \tau} + e^{\lambda_1 \tau} - e^{\lambda_2 \tau}$$
 (4.26)

The range space of  $\overline{B}$  is spanned by the vectors.

$$\begin{bmatrix} \eta_2 \\ 0 \\ (\lambda_2 - \lambda_1)\eta_3 \end{bmatrix} \text{ and } \begin{bmatrix} \eta_1 \\ \eta_2 \\ \eta_3 \end{bmatrix}$$
(4.27)

# Case III: Let T be the transformation matrix such that

$$\mathbf{T}^{-1}\mathbf{A}\mathbf{T} = \mathbf{J} = \begin{bmatrix} \lambda & 1 & 0\\ 0 & \lambda & 1\\ 0 & 0 & \lambda \end{bmatrix}$$

$$\mathbf{B} = \mathbf{T} \begin{bmatrix} n_{2}\alpha_{1}e^{\lambda \tau} & [n_{2}(\alpha_{1}\tau + \alpha_{2}) - n_{1}\alpha_{1}]e^{\lambda \tau} & [n_{2}(\alpha_{1}\frac{\tau^{2}}{2} + \alpha_{2}\tau + \alpha_{3}) - n_{1}(\alpha_{1}\tau + \alpha_{2})]e^{\lambda \tau} \\ n_{3}\alpha_{1}e^{\lambda \tau} & [n_{3}(\alpha_{1}\tau + \alpha_{2}) - n_{2}\alpha_{1}]e^{\lambda \tau} & [n_{3}(\alpha_{1}\frac{\tau^{2}}{2} + \alpha_{2}\tau + \alpha_{3}) - n_{2}(\alpha_{1}\tau + \alpha_{2})]e^{\lambda \tau} \\ 0 & -n_{3}\alpha_{1}e^{\lambda \tau} & -n_{3}(\alpha_{1}\tau + \alpha_{2})e^{\lambda \tau} \end{bmatrix} \begin{bmatrix} \mathbf{T}^{-1} \\ \mathbf{T}^{-1} \\ \mathbf{T}^{-1} \end{bmatrix}$$

$$(4.28)$$

where  $\alpha' = q'T$  and the  $\eta_i$ 's satisfy

$$n_{1} = \frac{\alpha_{1}^{2}\tau^{2} + 2\alpha_{1}\alpha_{2}\tau + 2\alpha_{2}^{2} - 2\alpha_{1}\alpha_{3}}{\alpha_{1}^{3}\tau^{2}}$$
(4.29)

$$\eta_2 = \frac{-2(\alpha_1 \tau + \alpha_2)}{\alpha_1^2 \tau^2}$$
(4.30)

$$n_3 = \frac{2}{\alpha_1 \tau^2}$$
 (4.31)

The range space of  $\overline{B}$  is spanned by the vectors

$$\begin{bmatrix} n_2 \\ n_3 \\ 0 \end{bmatrix} \quad \text{and} \quad \begin{bmatrix} n_1 \\ n_2 \\ n_3 \end{bmatrix}$$
(4.32)

It is important to note that the transformation matrix T will in general depend on A. Thus the expressions for B given above do not show B as an explicit function of A. However, the form that B must assume will be useful in our subsequent analysis.

As an application of the above construction, let us re-examine Popov's example. Using the uniqueness of B, straightforward calculations show that for degeneracy to occur,  $\tau$  must be 1, and the q vector must be such that  $q_2 = -2q_1$ ,  $q_3 = -q_1$ . We can therefore conclude that Popov's example will be p.d. for a unique vector q, up to scalar multiples, and a unique  $\tau$ .

## 4.3 <u>Delay Feedback Control of 3 Dimensional</u> Linear Systems

In this section, we apply the results developed in sections 3.3 and 4.2 to obtain the solution of the minimum time output zeroing problem using the feedback law (4.3) for the system defined by (4.1) and (4.2). We shall assume throughout this section that the system (4.1) is 3 dimensional. As in every theorem related to pointwise degeneracy, the result is simple and elegant in three dimensions.

Theorem 4.1 For matrices K and L to exist such that the closedloop system

$$x(t) = (A + CK)x(t) + CLx(t-\tau)$$
 (4.33)

satisfies q'x(t) = 0,  $t \ge 2\tau$ , for all initial conditions, it is necessary and sufficient that rank  $C \ge 2$ , and (A,C) be controllable.

It is helpful to discuss the conditions of Theorem 4.1 before the proof. This will help to motivate the construction given below. Rank C  $\geq$  2 is clearly necessary since otherwise rank CL  $\leq$  1 and by Corollary 3.1, (4.33) is always p.c. Controllability of (A,C) is introduced so that we can choose K to shift the eigenvalues of A arbitrarily (of course, it is necessary for another reason; see the necessity proof of Theorem 4.1 below). This enables us to avoid the singular situations described in Theorem 3.5. For example, if C is 3x3 and of full rank, then obviously we can find a matrix K such that the conditions of Theorem 3.5 are satisfied. Furthermore, if we construct a matrix B such that (A+CK,B,q,T) is p.d., then since C is nonsingular, there always exists a matrix L such that B = CL. Of

more interest is the case where rank C = 2. We can still choose K such that (A+CK) has eigenvalues which satisfy conditions (i) and (ii) of Theorem 3.5. However, it is not clear <u>a priori</u> that there will always exist L such that B = CL, where B is such that (A+CK,B,q,T) is p.d. Indeed, most of the work involved in proving Theorem 4.1 is to show we can choose the eigenvalues of A+CK in such a way that the matrix B, constructed as in section 4.2, always lie in the range space of C.

We shall need the following lemmas whose proofs are given in Appendix A.

Lemma 4.1 Let  $\alpha_1$ ,  $\alpha_2$ ,  $\alpha_3$  be real numbers with the property  $\alpha_2^2 - 4\alpha_1\alpha_3 > 0$ . Then the function  $f(\lambda) \stackrel{\Delta}{=} (\alpha_1 + \alpha_2\lambda + \alpha_3\lambda^2)e^{\lambda\tau}$  for  $\lambda$  real has one maximum and one minimum. Furthermore there exist straight lines of the form  $g(\lambda) = \gamma_{\beta}(\lambda - \beta)$  and  $g(\lambda) = \omega$  each of which intersect  $f(\lambda)$  at three real distinct points, none of which is a zero of  $f(\lambda)$ . Here  $\beta$  is a finite real number not equal to any of the zeros of  $f(\lambda)$ ,  $\gamma_{\beta}$  is some real number dependent on  $\beta$ , and  $\omega$  is some real constant.

<u>Lemma 4.2</u> Suppose we are given a 3-vector q, a 3x3 matrix A, and a 3x2 matrix C of full rank, with (A,C) controllable. Suppose q'c<sub>i</sub>  $\neq 0$ where c<sub>i</sub> is the i<sup>th</sup> column of C. Then there exist matrices K and P such that P<sup>-1</sup>(A+CK)P is in companion form, q'P = ( $\alpha_1 \ \alpha_2 \ \alpha_3$ ) is such that  $\alpha_2^2 - 4\alpha_1\alpha_3 > 0$ , and that P<sup>-1</sup>C has as its i<sup>th</sup> column

-65-

0

Furthermore, if the other column can be written as



with  $\beta_1 \neq 0$ , then P can be chosen such that  $\frac{\beta_2}{\beta_1}$  is not a zero of  $\eta(\lambda) \stackrel{\Delta}{=} \alpha_1 + \alpha_2 \lambda + \alpha_3 \lambda^2$ .

<u>Proof of Theorem 4.1</u> (Necessity) That rank  $C \ge 2$  is necessary has already been discussed above. If (A,C) is not controllable, (A+CK,C) is also not controllable for any K since constant state feedback has no effect on controllability. Noting that

 $(CL ACL..., A^{n-1}CL) = (C AC..., A^{n-1}C)L$ 

we see that (A+CK,CL) is also not controllable for any L. Since by Theorem 3.4, all 3 dimensional p.d. systems are regular, this implies that the feedback system (4.33) is p.c. for any K and L. Thus controllability of (A,C) is necessary.

(Sufficiency) By the discussion following the statement of Theorem 4.1, it suffices to consider the case where C is a 3x2 matrix of full rank. We give a constructive procedure for finding the matrices K and L.

<u>Step 1</u>: We may assume, without loss of generality, that  $q'c_1 \neq 0$ . We choose matrices  $K_1$  and  $P_1$  such that  $P_1^{-1}(A+CK_1)P_1$  is in companion form,  $P_1^{-1}C$  is of the form

0	β1
0	<sup>β</sup> 2
[1	β <sub>3</sub>

-66-

and  $q'P_1 = (\alpha_1 \ \alpha_2 \ \alpha_3)$  with  $\alpha_2^2 - 4\alpha_1 \alpha_3 > 0$ . By Lemma 4.2, such matrices exist (see the proof of Lemma 4.2 in Appendix A for the constructive procedure for  $K_1$  and  $P_1$ ).

<u>Step 2</u>: Choose a matrix  $K_2$  such that the matrix

$$A_2 \stackrel{\triangle}{=} P_1^{-1} (A+CK_1) P_1 + P_1^{-1}CK_2$$

has three real distinct eigenvalues  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$ . The  $\lambda_i$ 's are not arbitrary. Their specification will be given later.

Step 3: Use the Vandemonde matrix

$$P_{2} = \begin{bmatrix} 1 & 1 & 1 \\ \lambda_{1} & \lambda_{2} & \lambda_{3} \\ \lambda_{1}^{2} & \lambda_{2}^{2} & \lambda_{3}^{2} \end{bmatrix}$$

to diagonalize A<sub>2</sub>, i.e.,

$$P_{2}^{-1}A_{2}P_{2} = \Lambda = \begin{bmatrix} \lambda_{1} & 0 & 0 \\ 0 & \lambda_{2} & 0 \\ 0 & 0 & \lambda_{3} \end{bmatrix}$$

If we define  $z(t) = P_2^{-1} P_1^{-1} x(t)$ ,  $v(t) = u(t) - (K_1 + K_2 P_1^{-1}) x(t)$ , we see that

$$\dot{z}(t) = P_2^{-1} P_1^{-1} [A + C(K_1 + K_2 P_1^{-1})] P_1 P_2 z(t) + P_2^{-1} P_1^{-1} Cv(t)$$
  
=  $\Lambda z(t) + P_2^{-1} P_1^{-1} Cv(t)$  (4.34)

<u>Step 4</u>: We now restrict  $\lambda_1$ ,  $\lambda_2$ ,  $\lambda_3$  to be such that they are not solutions of the equation

$$\alpha_1 + \alpha_2 \lambda + \alpha_3 \lambda^2 = 0 \tag{4.35}$$

This implies  $q'P_1P_2 = (\eta_1 \ \eta_2 \ \eta_3)$ , where  $\eta_i = \alpha_1 + \alpha_2\lambda_i + \alpha_3\lambda_i^2$ , has components all nonzero. By Theorem 3.5, we can construct a matrix B such that  $(\Lambda, B, q'P_1P_2, \tau)$  is p.d. In fact, the results in section 4.2 show that B is given by

$$B = \frac{1}{D} \begin{bmatrix} 0 & \frac{\eta_2}{\eta_1} (\lambda_1 - \lambda_2) (\lambda_2 - \lambda_3) e^{2\lambda_2 \tau} & \frac{\eta_3}{\eta_1} (\lambda_1 - \lambda_3) (\lambda_2 - \lambda_3) e^{(\lambda_2 + \lambda_3) \tau} \\ \frac{\eta_1}{\eta_2} (\lambda_1 - \lambda_1) (\lambda_3 - \lambda_1) e^{2\lambda_1 \tau} & 0 & \frac{\eta_3}{\eta_2} (\lambda_2 - \lambda_3) (\lambda_3 - \lambda_1) e^{(\lambda_1 + \lambda_3) \tau} \\ \frac{\eta_1}{\eta_3} (\lambda_3 - \lambda_1) (\lambda_1 - \lambda_2) e^{(2\lambda_1 + \lambda_2 - \lambda_3) \tau} & \frac{\eta_2}{\eta_3} (\lambda_3 - \lambda_2) (\lambda_1 - \lambda_2) e^{(\lambda_1 + 2\lambda_2 - \lambda_3) \tau} & 0 \end{bmatrix}$$

where 
$$D = (\lambda_3 - \lambda_1) (e^{\lambda_1 \tau} - e^{\lambda_2 \tau}) + (\lambda_1 - \lambda_2) e^{(\lambda_2 - \lambda_3) \tau} (e^{\lambda_1 \tau} - e^{\lambda_2 \tau})$$

<u>Step 5</u>: We now show that there exists a 2x3 matrix  $L_1$  such that  $P_2^{-1} P_1^{-1} CL_1 = B$ . We know from section 4.2 that  $\mathscr{R}(B)$ , the range space of B, is spanned by the vectors

$$\begin{bmatrix} 0 \\ 1 \\ -\frac{\eta_2}{\eta_3} e^{(\lambda_2 - \lambda_3)\tau} \end{bmatrix} \text{ and } \begin{bmatrix} 1 \\ 0 \\ -\frac{\eta_1}{\eta_3} e^{(\lambda_1 - \lambda_3)\tau} \\ -\frac{\eta_1}{\eta_3} e^{(\lambda_1 - \lambda_3)\tau} \end{bmatrix}$$

A necessary and sufficient condition for the existence of  ${\rm L}^{}_{1}$  is that

 $\mathscr{R}(B) \subset \mathscr{R}(P_2^{-1} P_1^{-1} C)$ . Since dim  $\mathscr{R}(B) = \dim \mathscr{R}(P_2^{-1} P_1^{-1} C) = 2$ , L<sub>1</sub> exists if  $\mathscr{R}(B) = \mathscr{R}(P_2^{-1} P_1^{-1} C)$ . Thus we only need to find a 2x2 nonsingular matrix

$$\mathbf{L}_{2} = \begin{bmatrix} \boldsymbol{\ell}_{11} & \boldsymbol{\ell}_{12} \\ & & \\ \boldsymbol{\ell}_{21} & \boldsymbol{\ell}_{22} \end{bmatrix}$$

such that

$$P_{2}^{-1} P_{1}^{-1} C \begin{bmatrix} \ell_{11} & \ell_{12} \\ & & \\ \ell_{21} & \ell_{22} \end{bmatrix} = \begin{bmatrix} 0 & 1 \\ 1 & 0 \\ -\frac{\eta_{2}}{\eta_{3}} e^{(\lambda_{2} - \lambda_{3})\tau} & -\frac{\eta_{1}}{\eta_{3}} e^{(\lambda_{1} - \lambda_{3})\tau} \end{bmatrix}$$

The first column of  $L_1$  is then given by

given by

given by

$$\frac{1}{D} \frac{\eta_3}{\eta_2} (\lambda_2 - \lambda_3) (\lambda_3 - \lambda_1) e^{(\lambda_1 + \lambda_3)\tau} \begin{bmatrix} \ell_{11} \\ \\ \\ \ell_{21} \end{bmatrix} + \frac{1}{D} \frac{\eta_3}{\eta_1} (\lambda_1 - \lambda_3) (\lambda_2 - \lambda_3) e^{(\lambda_2 + \lambda_3)\tau} \begin{bmatrix} \ell_{12} \\ \\ \\ \\ \ell_{22} \end{bmatrix}.$$

Using the particular forms of  $P_1^{-1}$  and  $P_2$  we obtain

$$\begin{bmatrix} \beta_{1} \ell_{21} & \beta_{1} \ell_{22} \\ \beta_{2} \ell_{21} & \beta_{2} \ell_{22} \\ \ell_{11} + \beta_{3} \ell_{21} & \ell_{12} + \beta_{3} \ell_{22} \end{bmatrix} = \\ \begin{bmatrix} 1 - \frac{n_{2}}{n_{3}} e^{(\lambda_{2} - \lambda_{3})\tau} & 1 - \frac{n_{1}}{n_{3}} e^{(\lambda_{1} - \lambda_{3})\tau} \\ \lambda_{2} - \lambda_{3} \frac{n_{2}}{n_{3}} e^{(\lambda_{2} - \lambda_{3})\tau} & \lambda_{1} - \lambda_{3} \frac{n_{1}}{n_{3}} e^{(\lambda_{1} - \lambda_{3})\tau} \\ \lambda_{2}^{2} - \lambda_{3}^{2} \frac{n_{2}}{n_{3}} e^{(\lambda_{2} - \lambda_{3})\tau} & \lambda_{1}^{2} - \lambda_{3}^{2} \frac{n_{1}}{n_{3}} e^{(\lambda_{1} - \lambda_{3})\tau} \end{bmatrix}$$

$$(4.36)$$

We consider three cases

Case A. 
$$\beta_1 = 0$$
.

This requires

$$1 - \frac{\eta_2}{\eta_3} e^{(\lambda_2 - \lambda_3)\tau} = 0$$

 $1 - \frac{\eta_1}{\eta_3} e^{(\lambda_1 - \lambda_3)\tau} = 0$ 

and

These can be rewritten as

$$\eta_1 e^{\lambda_1 \tau} = \eta_2 e^{\lambda_2 \tau} = \eta_3 e^{\lambda_3 \tau}$$
(4.37)

Since  $\alpha_2^2 - 4\alpha_1^2 \alpha_3 > 0$ , by Lemma 4.1, we conclude that there exists a constant  $\omega$  such that the line  $\lambda = \omega$  intersects  $(\alpha_1^2 + \alpha_2^2 \lambda + \alpha_3^2) e^{\lambda \tau}$ 

at three real distinct points  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$ . These will be the eigenvalues that we choose. We can solve for  $\ell_{11}$ ,  $\ell_{12}$ ,  $\ell_{21}$ , and  $\ell_{22}$  from (4.36) explicitly by

$$\begin{aligned} \lambda_{21} &= \frac{\lambda_2 - \lambda_3}{\beta_2} \\ \lambda_{22} &= \frac{\lambda_1 - \lambda_3}{\beta_2} \\ \lambda_{11} &= \lambda_2^2 - \lambda_3^2 - \beta_3 \lambda_{21} \\ \lambda_{12} &= \lambda_1^2 - \lambda_3^2 - \beta_3 \lambda_{22} \end{aligned}$$

<u>Case B</u>.  $\beta_2 = 0$ .

This requires

$$\frac{\eta_1 e^{\lambda_1 \tau}}{\lambda_1} = \frac{\eta_2 e^{\lambda_2 \tau}}{\lambda_2} = \frac{\eta_3 e^{\lambda_3 \tau}}{\lambda_3}$$
(4.38)

By appealing to Lemma 4.1, we conclude that real and distinct  $\lambda_1$ ,  $\lambda_2$ and  $\lambda_3$  satisfying (4.38) exist. With the eigenvalues thus chosen, the matrix L<sub>2</sub> can be obtained from (4.36).

<u>Case C</u>.  $\beta_1, \beta_2$  both nonzero.

Completely analogous considerations show that we must have

$$\left(\lambda_{1}-\frac{\beta_{2}}{\beta_{1}}\right)\frac{e^{-\lambda_{1}T}}{\eta_{1}} = \left(\lambda_{2}-\frac{\beta_{2}}{\beta_{1}}\right)\frac{e^{-\lambda_{2}T}}{\eta_{2}} = \left(\lambda_{3}-\frac{\beta_{2}}{\beta_{1}}\right)\frac{e^{-\lambda_{3}T}}{\eta_{3}}$$
(4.39)

Appealing to Lemma 4.1 once again, we conclude that desired eigenvalues  $\lambda_i$ , i = 1,2,3 exist, and the matrix  $L_2$  can be evaluated from (4.36).

We have now constructed a p.d. system of the form

$$\dot{z}(t) = \Lambda z(t) + P_2^{-1} P_1^{-1} CL_1 z(t-\tau)$$
 (4.40)

with  $q' P_1 P_2 z(t) = 0$   $t \ge 2\tau$ 

In terms of the original coordinate system, we have

$$\dot{\mathbf{x}}(t) = [A+C(K_1+K_2P_1^{-1})]\mathbf{x}(t) + CL_1P_2^{-1}P_1^{-1}\mathbf{x}(t-\tau)$$
(4.41)

is p.d. for q. The desired feedback matrices are thus  $K = K_1 + K_2 P_1^{-1}$ and  $L = L_1 P_2^{-1} P_1^{-1}$ . The theorem is proved.

For greater clarity and completeness, we summarize the algorithm for constructing the matrices K and L in the three dimensional case. We assume, without loss of generality, that  $q'c_1 \neq 0$ .

(i) Construct, as in the proof of Lemma 4.2 in Appendix A, P<sub>1</sub>, K<sub>1</sub> such that  $P_1^{-1}(A + CK_1)P_1$  is in companion form,  $P_1^{-1}C$  is of the form  $\begin{bmatrix} 0 & \beta_1 \\ 0 & \beta_2 \\ 1 & \beta_3 \end{bmatrix}$ , and  $q'P_1 = (\alpha_1 \alpha_2 \alpha_3)$  with  $\alpha_2^2 - 4\alpha_1\alpha_3 > 0$ .

(ii) Let  $\eta(\lambda) = \alpha_1 + \alpha_2 \lambda + \alpha_3 \lambda^2$ . If  $\beta_1 = 0$ , choose three real and distinct numbers  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$  such that  $\eta(\lambda_1) \neq 0$ , i=1,2,3, and that  $\eta(\lambda_1)e^{\lambda_1 T} = \eta(\lambda_2)e^{\lambda_2 T} = \eta(\lambda_3)e^{\lambda_3 T}$ . If  $\beta_2 = 0$ , choose real and distinct  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$  such that  $\eta(\lambda_1) \neq 0$ , i=1,2,3, and
$$\frac{\eta(\lambda_1)e^{\lambda_1^{\mathsf{T}}}}{\lambda_1} = \frac{\eta(\lambda_2)e^{\lambda_2^{\mathsf{T}}}}{\lambda_2} = \frac{\eta(\lambda_3)e^{\lambda_3^{\mathsf{T}}}}{\lambda_3}$$

If  $\beta_1$  and  $\beta_2$  are both nonzero, choose real and distinct  $\lambda_1$ ,  $\lambda_2$ ,  $\lambda_3$  such that  $\eta(\lambda_i) \neq 0$ , i=1,2,3, and

$$\left(\lambda_{1} - \frac{\beta_{2}}{\beta_{1}}\right) \frac{e^{-\lambda_{1}\tau}}{\eta(\lambda_{1})} = \left(\lambda_{2} - \frac{\beta_{2}}{\beta_{1}}\right) \frac{e^{-\lambda_{2}\tau}}{\eta(\lambda_{2})} = \left(\lambda_{3} - \frac{\beta_{2}}{\beta_{1}}\right) \frac{e^{-\lambda_{3}\tau}}{\eta(\lambda_{3})}$$

(iii) Once  $\lambda_1$ ,  $\lambda_2$ ,  $\lambda_3$  are chosen, choose a matrix  $K_2$  such that the matrix  $P_1^{-1}(A + CK_1)P_1 + P_1^{-1}CK_2$  has as its eigenvalues  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$ .

(iv) Compute the Vandemonde matrix

$$P_2 = \begin{bmatrix} 1 & 1 & 1 \\ \lambda_1 & \lambda_2 & \lambda_3 \\ \lambda_1^2 & \lambda_2^2 & \lambda_3^2 \end{bmatrix}$$

(v) Calculate the matrix  $L_1$  as in step 5 of the proof. In other words, find  $L_1$  such that  $(P_2^{-1} P_1^{-1} (A + CK_1 + CK_2 P_1^{-1}) P_1 P_2, P_2^{-1} P_1^{-1} CL_1, P_2' P_1' q, \tau)$  is p.d. The choice of  $\lambda_1, \lambda_2$ , and  $\lambda_3$  in (ii) guarantees that  $L_1$  exists.

(vi) Finally, the desired feedback matrices are given by

$$K = K_1 + K_2 P_1^{-1}$$
  
and 
$$L = L_1 P_2^{-1} P_1^{-1}$$

Note that once the eigenvalues  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$  are chosen, the calculations involved in steps (iii) to (vi) are completely explicit and straightforward. The calculations involved in step (i) are detailed in the proof of Lemma 4.2 in Appendix A. They are all explicit excepting perhaps the choice of  $f_1$  and  $f_2$ . The numbers  $f_1$ and  $f_2$  are chosen to satisfy  $\alpha_2^2 - 4\alpha_1'\alpha_3 > 0$  and  $\frac{d_{22}}{d_{12}}$  not being a zero of  $\alpha_1' + \alpha_2\lambda + \alpha_3\lambda^2$ . A simple search procedure will enable us to find an appropriate  $f_1$  and  $f_2$ . As for step (ii), we need to find eigenvalues  $\lambda_1$ ,  $\lambda_2$ ,  $\lambda_3$  which satisfy (4.37), or (4.38), or (4.39). One way of doing this is to plot, as in the proof of Lemma 4.1 in Appendix A, the function  $f(\lambda)$  and draw the appropriate straight line to intersect  $f(\lambda)$  at three real distinct points. While other methods for solving (4.37) - (4.39) can probably be devised, the computations involved in step (ii) will, unfortunately, be nonexplicit.

<u>Remark 4.1</u> The proof of Theorem 4.1 is rather clumsy and involved. Since we only assume controllability of (A,C) we only have freedom in assigning the eigenvalues. The lengthy construction shows that we can find a suitable set of eigenvalues so that the matrix B we constructed necessarily lies in the range space of the input matrix C. Certainly a simpler and more elegant proof is desirable. However, because of the nonlinear nature of the problem, a substantial simplification may not be possible.

<u>Remark 4.2</u> Remark 3.4 shows that the eigenvalues of the delay feedback system (4.41) are precisely those of  $[A + C(K_1 + K_2P_1^{-1})]$ , and these satisfy (4.44) or (4.45) or (4.46). In general, there are many solutions to these equations (see Appendix A). If we can choose the eigenvalues so that they are negative, we will have obtained an

asymptotically stable p.d. system. We have not been able to show as yet that such a choice is always possible. Thus, the closed-loop p.d. system (4.41) may be unstable.

<u>Remark 4.3</u> We have assumed that the delay time  $\tau$  is prescribed in advance and not available as a design parameter. However, if it is to be designed, the smaller  $\tau$  is, the larger the elements of the B matrix will be (see eqs. (4.19) and (4.17)). Hence we have a tradeoff in this situation. We shall not investigate this point further.

## 4.4 <u>Delay Feedback Control of Higher Dimensional</u> Linear Systems

Our goal in this section is to extend the feedback control results for 3 dimensional systems to higher dimensions n > 3. The method of proof is substantially the same as that of Theorem 4.1. Notationally however, it is much more involved.

<u>Theorem 4.2</u> For matrices K and L to exist such that the closed-loop system

$$\dot{\mathbf{x}}(t) = (\mathbf{A} + \mathbf{C}\mathbf{K})\mathbf{x}(t) + \mathbf{C}\mathbf{L}\mathbf{x}(t-\tau)$$

satisfies q'x(t) = 0,  $t \ge 2\tau$ , for all initial conditions, it is sufficient that rank  $C \ge 3$ , and (A,C) be controllable.

Proof: Consider the equations

$$q'r = 1$$
 (4.42)

$$q'e^{AT}r = 0$$
 (4.43)

$$q'e^{AT}Ar = 0 (4.44)$$

where  $q' = (q_1 q_2 \dots q_n)$ 

$$A = \begin{bmatrix} \lambda_{1} & 0 & \dots & \dots & 0 \\ 0 & \lambda_{2} & 0 & \dots & \ddots & \ddots \\ \vdots & 0 & \ddots & \ddots & \vdots \\ \vdots & 0 & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ \vdots & \vdots & \ddots & \ddots & \ddots \\ 0 & 0 & \dots & 0 & \lambda_{n} \end{bmatrix}$$

Suppose at least three of the eigenvalues  $\lambda_i$  are real and distinct with the corresponding  $q_i$ 's nonzero. Without loss of generality, we may take  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$  to be these eigenvalues with  $q_1$ ,  $q_2$ ,  $q_3$  nonzero. Under these conditions, (4.42) - (4.44) can be solved to give

$$r_{3} = \frac{1}{q_{3}w_{3}} \left\{ \frac{\begin{pmatrix} (\lambda_{1} + \lambda_{2})\tau \\ (\lambda_{2} - \lambda_{1})e \\ (\lambda_{2} - \lambda_{1})e \\ (e - e ) \end{pmatrix} - \sum_{i=4}^{n} q_{i}w_{i}r_{i} \right\}$$
(4.45)

$$\mathbf{r}_{2} = -\frac{1}{q_{2}(e^{\lambda_{2}\tau}-e^{\lambda_{1}\tau})} \left\{ e^{\lambda_{1}\tau} + \sum_{i=3}^{n} q_{i}r_{i}(e^{\lambda_{i}\tau}-e^{\lambda_{1}\tau}) \right\}$$
(4.46)

$$r_1 = 1 - \sum_{i=2}^{n} q_i r_i$$
 (4.47)

$$w_{i} = \frac{(\lambda_{i} - \lambda_{1})e^{\lambda_{i}^{T}}(e^{\lambda_{2}^{T}} - e^{\lambda_{1}^{T}}) - (e^{\lambda_{i}^{T}} - e^{\lambda_{1}^{T}})(\lambda_{2} - \lambda_{1})e^{\lambda_{2}^{T}}}{e^{\lambda_{2}^{T}} - e^{\lambda_{1}^{T}}}$$
(4.48)

where

Thus  $r_1$ ,  $r_2$  and  $r_3$  are completely determined in terms of the 2n-3 "free" variables  $\lambda_1$ ,  $\lambda_2$ ,..., $\lambda_n$ , and  $r_4$ ,  $r_5$ ,..., $r_n$ . Constructing  $\overline{Z} = rq'e^{AT}$  and  $\overline{B} = A\overline{Z} - \overline{Z}A$  we obtain



Note that  $\overline{B}$  is of rank 2, and its range space is spanned by the vectors

0		$r_1^{(\lambda_1 - \lambda_2)}$
$r_2(\lambda_2 - \lambda_1)$		0
	and	$r_3(\lambda_3 - \lambda_2)$
•		•
$r_n \left( \lambda_n - \lambda_1 \right)$		$r_n(\lambda_n-\lambda_2)$

We now proceed as in the proof of Theorem 4.1. We need only to consider the case where C is nx3 of full rank. Construct matrices K and  $P_1$  such that  $P_1^{-1}(A + BK)P_1$  is in companion form with distinct eigenvalues, and

$$P_{1}^{-1}C = \begin{bmatrix} 0 & d_{12} & d_{13} \\ 0 & \ddots & \ddots \\ 1 & d_{n2} & d_{n3} \end{bmatrix}$$

The restrictions on the eigenvalues will be given later. Next, we diagonalize  $P_1^{-1}(A + BK)P_1$  by a Vandemonde matrix

 $P_{2} = \begin{bmatrix} 1 & 1 & . & . & 1 \\ \lambda_{1} & \lambda_{2} & & \lambda_{n} \\ . & . & & \\ . & . & & \\ . & . & & \\ \lambda_{n-1}^{n-1} & \lambda_{2}^{n-1} & \lambda_{n}^{n-1} \\ 1 & 2 & n \end{bmatrix}$ 

so that

q'

$$P_{2}^{-1} P_{1}^{-1} (A + BK) P_{1} P_{2} \stackrel{\Delta}{=} \Lambda = \begin{bmatrix} \lambda_{1} & 0 & \cdots & 0 \\ 0 & \lambda_{2} & 0 & \cdots & 0 \\ \vdots & 0 & \vdots & \vdots \\ 0 & 0 & \cdots & 0 \\ \vdots & \vdots & \ddots & 0 \\ 0 & 0 & \cdots & 0 & \lambda_{n} \end{bmatrix}$$

$$P_{1} P_{2} = (\eta_{1} \cdots \eta_{n})$$

with 
$$\eta_i = \alpha_1 + \alpha_2^{\lambda_i} + \dots + \alpha_n^{\lambda_i^{n-1}}$$

and 
$$q'P_1 = (\alpha_1 \dots \alpha_n)$$

We now impose the restriction that three of the eigenvalues, say  $\lambda_1$ ,  $\lambda_2$  and  $\lambda_3$  are real and distinct, and that the resulting  $\eta_1$ ,  $\eta_2$ , and  $\eta_3$  are nonzero. Under these conditions, we may construct a matrix B

such that ( ,B,q'P<sub>1</sub>P<sub>2</sub>, ) is p.d. Analogous to step 5 of the proof of Theorem 4.1, we see that we need to find a matrix  $L_2$  such that

$$P_{1}^{-1} CL_{2} = P_{2} \begin{bmatrix} 0 & r_{1}(\lambda_{1}-\lambda_{2}) \\ r_{2}(\lambda_{2}-\lambda_{1}) & 0 \\ \vdots & r_{3}(\lambda_{3}-\lambda_{2}) \\ \vdots & \vdots \\ r_{n}(\lambda_{n}-\lambda_{1}) & r_{n}(\lambda_{n}-\lambda_{2}) \end{bmatrix}$$
$$= \begin{bmatrix} \sum_{i\neq 1} r_{i}(\lambda_{i}-\lambda_{1}) & \sum_{i\neq 2} r_{i}(\lambda_{i}-\lambda_{2}) \\ \sum_{i\neq 1} r_{i}\lambda_{i}(\lambda_{1}-\lambda_{1}) & \vdots \\ \vdots \\ \sum_{i\neq 1} r_{i}\lambda_{i}^{n-1}(\lambda_{i}-\lambda_{1}) & \sum_{i\neq 2} r_{i}\lambda_{i}^{n-1}(\lambda_{i}-\lambda_{2}) \\ \vdots \\ \sum_{i\neq 1} r_{i}\lambda_{i}^{n-1}(\lambda_{1}-\lambda_{1}) & \sum_{i\neq 2} r_{i}\lambda_{i}^{n-1}(\lambda_{1}-\lambda_{2}) \end{bmatrix}$$

Let

$$L_{2} = \begin{bmatrix} \ell_{11} & \ell_{12} \\ \ell_{21} & \ell_{22} \\ \ell_{31} & \ell_{32} \end{bmatrix}$$

Then

$$P^{-1}CL_{2} = \begin{bmatrix} d_{12}\ell_{21}+d_{13}\ell_{31} & d_{12}\ell_{22}+d_{13}\ell_{32} \\ d_{22}\ell_{21}+d_{23}\ell_{31} & d_{22}\ell_{22}+d_{23}\ell_{32} \\ \vdots & \vdots \\ d_{n-1,2}\ell_{21}+d_{n-1,3}\ell_{31} & d_{n-1,2}\ell_{22}+d \\ \ell_{11}+d_{n2}\ell_{21}+d_{n3}\ell_{31} & \ell_{12}+d_{n2}\ell_{22}+d_{n3}\ell_{32} \end{bmatrix}$$

We can solve for  $\ell_{11}$  and  $\ell_{12}$  independently. Thus we only need to show that we can solve for  $\ell_{21}$ ,  $\ell_{22}$ ,  $\ell_{31}$ , and  $\ell_{32}$  consistently from the equations

$$d_{j2}\ell_{21} + d_{j3}\ell_{31} = \sum_{i \neq 1} r_i \lambda_i^{j-1} (\lambda_i - \lambda_1) \quad j=1,...,n-1$$
 (4.50)

$$d_{j2}\ell_{22} + d_{j3}\ell_{32} = \sum_{i \neq 2} r_i \lambda_i^{j-1} (\lambda_i - \lambda_2) \qquad j=1,...,n-1$$
(4.51)

We now eliminate  $\ell_{21}$ ,  $\ell_{22}$ ,  $\ell_{31}$ ,  $\ell_{32}$  from (4.50) and (4.51). Using the fact that rank C = 3, a little thought shows that we will obtain 2n-6 equations relating  $\lambda_1, \ldots, \lambda_n$ ,  $r_1, \ldots, r_n$ . Since the number of "free" variables is 2n-3, we expect that it would be possible to choose the  $\lambda_i$ 's and  $r_i$ 's in such a way that all the requirements are fulfilled.

The simplest way to see this is to examine the special case where  $d_{j2} = 0$  for all j except j = m, and  $d_{j3} = 0$  for all j except j = k. This gives rise to equations of the form

$$\sum_{i \neq 1} r_i \lambda_i^{j-1} (\lambda_i - \lambda_1) = 0 \qquad j=1,\dots,n-1 \qquad (4.52)$$
$$j\neq m,k$$

and 
$$\sum_{i \neq 2} r_i \lambda_i^{j-1} (\lambda_i - \lambda_2) = 0 \qquad j=1,\ldots,n-1 \qquad (4.53)$$
$$j\neq m,k$$

If we take  $\lambda_1$ ,  $\lambda_2$ , and  $\lambda_3$  as fixed real distinct constants, these give rise to 2n-6 equations in the 2n-6 variables  $\lambda_4, \ldots, \lambda_n$ ,  $r_4, \ldots, r_n$ . Eliminate the variables  $r_4, \ldots, r_n$  in the above equations. This yields n-3 equations in the variables  $\lambda_4, \ldots, \lambda_n$ . Next we note that these equations are invariant under permutation of the indices in  $\lambda_i$ . Hence if we reduce these n-3 equations into a single equation involving  $\lambda_i$ for some i, say of the form  $f(\lambda_i) = 0$ , then  $f(\lambda_j) = 0$ ,  $j=4,\ldots$ n also. However, the form of equations (4.52) and (4.53) implies that f is an exponential polynomial [1] and it is known [1] that the zeros of these functions are distributed symmetrically about the real axis with asymptotic distribution separated at a certain distance. Since there is an infinite number of zeros of these exponential polynomials, it is possible to choose  $\lambda_4, \ldots, \lambda_n$  such that they form a symmetric set with the properties  $(\alpha_1 + \alpha_2 \lambda_1 + \ldots + \alpha_n \lambda_1^{n-1}) \neq 0$ ,  $\lambda_1$ 's distinct i=1,\ldots,n, and  $\lambda_1$ ,  $\lambda_2$ ,  $\lambda_3$  real.

With such a choice, the conditions for the existence of the matrix  $L_2$  are satisfied and we can construct, from  $L_2$ , a matrix  $L_1$  such that  $P_2^{-1} P_1^{-1} CL_1 = B$ . While we have only discussed the case where  $d_{j2} = 0$ ,  $j = 1, \ldots, n-1$ ,  $j \neq m$  and  $d_{j3} = 0$ ,  $j = 1, \ldots, n-1$ ,  $j \neq k$ , the other cases can be treated in exactly the same way. We can eliminate variables from (4.50) and (4.51) in a similar fashion to that discussed above and show that the desired  $\lambda_1, \lambda_2, \ldots, \lambda_n$  exist. The details are therefore omitted. The rest of the proof now proceeds

-81-

in exactly the same way as that of Theorem 4.1.

We summarize here the algorithm for constructing the matrices K and L in the higher dimensional case. Again, we assume  $q'c_1 \neq 0$ .

(i) Choose  $P_1$ ,  $K_1$  such that  $P_1^{-1}(A + CK_1)P_1$  is in companion form,  $P_1^{-1}C$  is of the form

$$\begin{bmatrix} 0 & d_{12} & d_{13} \\ 0 & d_{22} & d_{23} \\ \vdots & \vdots & \vdots \\ 0 & \vdots & \vdots \\ 1 & d_{n2} & d_{n3} \end{bmatrix}, \text{ and } q'P_1 = (\alpha_1 \ \alpha_2 \dots \ \alpha_n)$$

(ii) Choose eigenvalues  $\lambda_1$ ,  $\lambda_2$ ,..., $\lambda_n$  such that (4.50) - (4.51) can be solved consistently for  $\ell_{ij}$ , i = 1,2,3, j = 1,2. Such a choice is possible by the proof of Theorem 4.2.

(iii) Choose K<sub>2</sub> such that the matrix  $P_1^{-1}(A + CK_1)P_1 + P_1^{-1}CK_2$ has as its eigenvalues  $\lambda_1, \lambda_2, \dots, \lambda_n$ .

(iv) Compute the Vandemonde matrix

$$P_{2} = \begin{bmatrix} 1 & 1 & 1 \\ \lambda_{1} & \lambda_{2} & \lambda_{n} \\ \vdots & \vdots & \vdots \\ \lambda_{n-1}^{n-1} & \lambda_{2}^{n-1} & \lambda_{n}^{n-1} \end{bmatrix}$$

(v) Calculate the matrix  $L_1$  from  $\ell_{ij}$ , i = 1, 2, 3, j = 1, 2 such that

$$(P_2^{-1} P_1^{-1}(A + CK_1 + CK_2P_1^{-1})P_1P_2, P_2^{-1} P_1^{-1}CL_1, P_2'P_1' q, \tau)$$
 is p.d.

(vi) The matrices K and L are given by

$$K = K_1 + K_2 P_1^{-1}$$
$$L = L_1 P_2^{-1} P_1^{-1}$$

Except for step (ii), all the calculations involved in the algorithm are explicit. To choose the appropriate eigenvalues  $\lambda_1$ ,  $\lambda_2$ ,...,  $\lambda_n$  in step (ii), we can first eliminate  $\ell_{21}$ ,  $\ell_{22}$ ,  $\ell_{31}$ ,  $\ell_{32}$ ,  $r_1$ ,  $r_2$ ,...,  $r_n$  from (4.50) - (4.51). This elimination involves only algebraic operations. We are then left with n-3 coupled transcendental equations involving exponential polynomials in the variables  $\lambda_1$ ,  $\lambda_2$ ,...,  $\lambda_n$ . For fixed, real and distinct  $\lambda_1$ ,  $\lambda_2$ ,  $\lambda_3$ , an iterative procedure is now needed to find the appropriate values for  $\lambda_4$ ,  $\lambda_5$ ,...,  $\lambda_n$ . We have not had time to investigate the numerical aspects involved in solving these transcendental equations. However, quite a lot of work has been done on the solutions of transcendental equations involving exponential polynomials. The reader may, for example, consult [1] and the references therein.

<u>Remark 4.4</u> Theorem 4.2 gives only a sufficient condition for the solvability of the feedback problem. It is not possible to obtain necessary and sufficient conditions by our approach, since for n > 3, solvability of (4.42) - (4.44) is not a necessary condition for p.d.

<u>Remark 4.5</u> It may be possible to prove Theorem 4.2 under the assumption that rank C  $\geq$  2. A similar argument will then show that we have 2n-4 equations in 2n-3 variables. However, the algebraic difficulties in

-83-

showing that these equations can be consistently solved for suitable eigenvalues  $\lambda_1, \ldots, \lambda_n$  are much more formidable.

### 4.5 Sensitivity of Pointwise Degenerate Systems

In the last two sections, we have given conditions under which a delay feedback system will have its output identically zero after  $2\tau$ units of time. If these results are to be used in practice, we must ensure that the feedback system will behave reasonably well under perturbation of its parameters. Since we usually know which particular combination of state variables serves as the output, it is reasonable to assume that the output vector q is fixed, and we shall do so. There are two questions of interest. Suppose for a fixed set of parameters  $A_{o}$ ,  $C_{o}$ ,  $\tau_{o}$ , there exist matrices  $K_{o}$  and  $L_{o}$  which give rise to a p.d. system (A + C K, C L, q,  $\tau_0$ ). This yields a (nonunique) "design" function f: (A, C,  $\tau$ )  $\mapsto$  (K,L), and a "performance" function g(t):  $(A,C,\tau,f(A_o, C_o, \tau_o)) \mapsto q'x(t)$  for  $t \geq 2\tau$ . Continuity of f at  $(A_0, C_0, \tau_0)$  implies that the delay feedback design procedure is wellposed. Similarly, continuity of g(t) at (A<sub>o</sub>, C<sub>o</sub>,  $\tau_o$ , K<sub>o</sub>, L<sub>o</sub>) implies that small variations in the parameter values give rise to a small degradation in the system performance. Note that the continuity properties of the functions f and g(t) are related to those of  $f_1$ and  $g_1(t)$ , where  $f_1: (A, \tau) \rightarrow B$  with (A, B, q,  $\tau$ ) p.d., and  $g_1(t)$ : (A,  $\tau$ ,  $f_1(A_0, \tau_0)$ )  $\mapsto q'x(t)$ , for  $t \ge 2\tau$ . This is because in our construction of the matrices K and L, we first construct a matrix  $B_1$ such that (A + CK,  $B_1$ , q,  $\tau$ ), is p.d., and then we construct a matrix L such that  $B_1 = CL$ . If C is the identity matrix and K can be taken to be

-84-

zero, then the properties of the functions f and g(t) reduce to those of  $f_1$  and  $g_1(t)$ . We shall only study the behavior of  $f_1$  and  $g_1(t)$ . This section contains some simple results in this direction.

For the study of  $f_1$ , we will assume that A has distinct eigenvalues. This is the case of most interest to us since in our proof of Theorems 4.1 and 4.2, we assign distinct eigenvalues to the matrix A + CK. We shall concern ourselves only with perturbations of the form A +  $\varepsilon A_1 \stackrel{\Delta}{=} A(\varepsilon)$  and  $\tau + \varepsilon \tau_1 = \tau(\varepsilon)$ . The following perturbation result for matrices is known [49].

Lemma 4.3 If  $\lambda$  is a simple eigenvalue of A, then for sufficiently small  $|\varepsilon|$ , there is an eigenvalue  $\lambda(\varepsilon)$  of A( $\varepsilon$ ) with a power series expansion

$$\lambda(\varepsilon) = \lambda + \varepsilon \lambda^{(1)} + \varepsilon^2 \lambda^{(2)} + \dots$$

and there is a corresponding eigenvector  $\mathbf{x}(\boldsymbol{\epsilon})$  with a power series expansion

$$\mathbf{x}(\varepsilon) = \mathbf{x} + \varepsilon \mathbf{x}^{(1)} + \dots$$

where x is the eigenvector of A corresponding to  $\lambda$ .

A procedure for computing the  $\lambda^{(i)}$ 's and  $x^{(i)}$ 's from A and A may be found in [49]. We can now state

<u>Theorem 4.3</u> Suppose A has n distinct eigenvalues  $\lambda_1, \ldots, \lambda_n$ . The matrix  $B(\varepsilon) = f_1[A(\varepsilon), \tau(\varepsilon)]$  has a power series expansion in  $\varepsilon$  for sufficiently small  $|\varepsilon|$ 

$$B(\varepsilon) = B + \varepsilon B^{(1)} + \dots$$

where  $B = f_1(A,\tau)$ 

<u>Proof</u>: Since A has distinct eigenvalues, a transformation T that diagonalizes A consists of the n eigenvectors of A. By Lemma 4.3, the eigenvectors of A( $\epsilon$ ) can be expanded in a power series in  $\epsilon$  for sufficiently small  $|\epsilon|$ . We can therefore also expand T( $\epsilon$ ) and T<sup>-1</sup>( $\epsilon$ ) in a power series. This implies q'T( $\epsilon$ ) can be expanded in a power series. Equations (4.45) - (4.49) show that  $\overline{B}(\epsilon)$  can again be expanded in a power series in  $\epsilon$ . Finally, since B( $\epsilon$ ) = T( $\epsilon$ ) $\overline{B}(\epsilon)$ T<sup>-1</sup>( $\epsilon$ ), B( $\epsilon$ ) can also be expanded in a power series in  $\epsilon$  for sufficiently small  $|\epsilon|$ .

Next, we consider the behavior of  $g_1(t)$  again under perturbations of the form  $A(\varepsilon) = A + \varepsilon A_1$ ,  $\tau(\varepsilon) = \tau + \varepsilon \tau_1$ .

Theorem 4.4 Suppose the system

$$\dot{x}(t) = Ax(t) + Bx(t-\tau)$$
 (4.54)

is p.d. with respect to q for t  $\geq 2\tau$  . Then the system

$$\dot{\mathbf{x}}_{\varepsilon}(t) = \mathbf{A}(\varepsilon)\mathbf{x}_{\varepsilon}(t) + \mathbf{B}\mathbf{x}_{\varepsilon}(t-\tau(\varepsilon))$$
(4.55)

will have, for each t  $\geq 2\tau$ ,

$$q'x_{\varepsilon}(t) \stackrel{\Delta}{=} y_{\varepsilon}(t) = \varepsilon \gamma_1(t) + \varepsilon^2 \gamma_2(t) + \dots$$

<u>Proof</u>: Let  $z(t) = P_1 x_{\varepsilon}(t-(k-1)\tau(\varepsilon)) + \dots + P_k x_{\varepsilon}(t)$  where  $P_1, \dots, P_k$ satisfy Popov's condition (3.3) - (3.8). Then

$$\dot{z}(t) = P_1 A(\varepsilon) x_{\varepsilon} (t - (k - 1)\tau(\varepsilon)) + P_1 B x_{\varepsilon} (t - k\tau(\varepsilon))$$

$$+ \dots + P_k A(\varepsilon) x_{\varepsilon} (t) + P_k B x_{\varepsilon} (t - \tau(\varepsilon))$$

$$= V z(t) + P_1 A_1 x_{\varepsilon} (t - (k - 1)\tau(\varepsilon)) + \dots + \varepsilon P_k A_1 x_{\varepsilon} (t)$$

Thus,

$$v'z(t) = v'e^{V\tau(\varepsilon)}z(t-\tau(\varepsilon))$$

+ 
$$\varepsilon \int_{t-\tau(\varepsilon)}^{t} e^{V(t-s)} \{P_1 A_1 x_{\varepsilon} (s-(k-1)\tau(\varepsilon))\}$$

+ ... + 
$$P_{k}A_{1}x_{\varepsilon}(s)$$
 ds

$$\stackrel{\Delta}{=} v' e^{V_{\tau}(\varepsilon)} z(t-\tau(\varepsilon)) + \varepsilon \gamma_{\varepsilon}(t)$$
(4.56)

Using equations (3.6) - (3.8) in Chapter 3,

$$v'z(t) - v'e^{V\tau(\varepsilon)}z(t-\tau(\varepsilon))$$

$$= v'z(t)-v'e^{V\tau}z(t-\tau(\varepsilon))-v'(e^{V\tau(\varepsilon)}-e^{V\tau})z(t-\tau(\varepsilon))$$

$$= q'x_{\varepsilon}(t)-v'(e^{V\tau(\varepsilon)}-e^{V\tau})z(t-\tau(\varepsilon)) \qquad (4.57)$$

Combining (4.61) and (4.62), we get

$$q'x_{\varepsilon}(t) = v'(e^{V\tau(\varepsilon)} - e^{V\tau})z(t - \tau(\varepsilon)) + \varepsilon \gamma_{\varepsilon}(t)$$
(4.58)

The right hand side of (4.58) is precisely of the form  $\epsilon \gamma_1(t) + \epsilon^2 \gamma_2(t) + \dots$ , proving the theorem.

<u>Remark 4.6</u> Theorem 4.4 does not give any estimate of the magnitudes of  $\gamma_1(t)$ ,  $\gamma_2(t)$ , etc. Furthermore, no claim is made on the uniform behavior in t of the output  $y_{\varepsilon}(t)$ . In fact, if the system (4.55) is unstable,  $\gamma_1(t)$ ,  $\gamma_2(t)$ , etc., which are really derived from  $x_{\varepsilon}(t)$ , can become large as t gets large. However, if the original system (4.54) is asymptotically stable, then for small enough  $|\varepsilon|$ , the perturbed system (4.55) is also asymptotically stable [2]. In that case,  $\gamma_1(t)$ ,  $\gamma_2(t)$ , etc., will go to zero as t gets large.

To a certain extent, the above remark limits the applicability of the delay feedback approach we have explored so far. If the parameters of the system are perturbed, the errors in the output may grow, and hence the performance may become unacceptable after some time. However, if in the proofs of Theorems 4.1 and 4.2, we can choose the  $\lambda_i$ 's such that  $\Re e \lambda_i < 0$ , then the resulting feedback p.d. system will be asymptotically stable. In that case, the above mentioned problem does not arise. Whether we can always find feedback matrices K and L such that the resulting closed-loop system is asymptotically stable in addition to being p.d. remains an open problem.

## 4.6 Some Examples

We give two examples which illustrate the use of delay feedback in certain problems in linear system theory.

Example 1: "Deadbeat" output control.

Consider the system

$$\dot{\mathbf{x}}(t) = \mathbf{A}\mathbf{x}(t) + \mathbf{C}\mathbf{u}(t)$$

where

$$A = \begin{bmatrix} 1 & 1 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & 0 & 1 \\ \end{bmatrix}$$
$$C = \begin{bmatrix} 0 & 0 \\ 0 & 1 \\ 1 & 0 \end{bmatrix}$$
$$y(t) = q'x(t)$$

with q' = (2 -2 1)

The problem is to drive y(t) to zero in minimum time by delay feedback control of the form

$$u(t) = Kx(t) + Lx(t-1)$$

Let us note that (A,C) is controllable and rank C = 2. Also A violates condition (i) of Theorem 4.1. Thus we cannot solve this problem by using only delay feedback control (i.e., K = 0).

One convenient choice for K is

$$K = \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & 1 \end{bmatrix}$$

$$A + CK = \begin{bmatrix} 1 & 1 & 0 \\ 0 & 1 & 1 \\ 0 & 0 & 1 \end{bmatrix}$$

then

and a suitable B is

$$B = \begin{bmatrix} 0 & 0 & 0 \\ 2e & 0 & 0 \\ 0 & -2e & 0 \end{bmatrix}$$
  
If we take L = 
$$\begin{bmatrix} 0 & -2e & 0 \\ 2e & 0 & 0 \end{bmatrix}$$

we see that CL = B.

With these choices, one can easily verify that (A + CK, CL, q, 1) is p.d. The output y(t) has been controlled to zero and will remain zero for all t  $\geq 2$ .

The problem that is in a sense "dual" to the deadbeat control problem is that of a "deadbeat" observer. Consider the system

 $\dot{x}(t) = Ax(t)$ y(t) = Cx(t)

We would like to reconstruct the state of the system. The usual approach is, under the assumption of observability, to construct an observer of the form

$$\hat{\mathbf{x}}(t) = A\hat{\mathbf{x}}(t) + K[\mathbf{y}(t) - C\hat{\mathbf{x}}(t)]$$

such that the estimation error  $e(t) = x(t) - \hat{x}(t)$  goes to zero asymptotically. The theory developed here suggests the use of a delay observer of the form

$$\hat{\mathbf{x}}(t) = A\hat{\mathbf{x}}(t) + K[\mathbf{y}(t)-C\hat{\mathbf{x}}(t)] + L[\mathbf{y}(t-\tau)-C\hat{\mathbf{x}}(t-\tau)]$$

such that a linear combination of the error, q'e(t), is driven to zero in minimum time. If we can observe all but one component of the error, q'e(t), is driven to zero in minimum time. If we can observe all but one component of the error vector, this will enable us to reconstruct the estimation error and hence the state of the system in finite time. While finding necessary and sufficient conditions for the existence of such a delay observer is an open problem at the present, we can give an example illustrating the technique involved.

Example 2: "Deadbeat" observer.

Consider the system

$$\dot{\mathbf{x}}(t) = A\mathbf{x}(t)$$
  
 $\mathbf{y}(t) = C\mathbf{x}(t)$ 

where

$$A = \begin{bmatrix} -1 & 2 & 0 \\ 2 & -2 & -1 \\ 1 & 2 & 0 \end{bmatrix}$$
$$C = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 2 & 0 \end{bmatrix}$$

(A,C) is observable and rank C = 2. Since  $x_1$  and  $x_2$  are directly observed, we need only estimate  $x_3$ . Using the usual reduced order observer, we can asymptotically reconstruct the state by a 1 dimensional observer.

Here we shall use delay feedback. Consider an observer of the form

$$\dot{\hat{x}}(t) = A\hat{x}(t) - K[y(t)-C\hat{x}(t)] - L[y(t-1) - C\hat{x}(t-1)]$$
(4.59)

Then the estimation error  $e(t) = x(t) - \hat{x}(t)$  satisfies

$$\dot{e}(t) = (A + KC)e(t) + LCe(t-1)$$
 (4.60)

We want to find K and L such that (4.60) is p.d. One such choice is

$$K = \begin{bmatrix} 1 & 0 \\ -2 & 1 \\ -1 & -1 \end{bmatrix}$$
$$L = \begin{bmatrix} 0 & 0 \\ 1 & 0 \\ 0 & 1 \end{bmatrix}$$

Then

$$A + KC = \begin{bmatrix} 0 & 2 & 0 \\ 0 & 0 & -1 \\ 0 & 0 & 0 \end{bmatrix}$$
$$LC = \begin{bmatrix} 0 & 0 & 0 \\ 1 & 0 & 0 \\ 0 & 2 & 0 \end{bmatrix}$$

Equation (4.60) becomes precisely Popov's example, and it satisfies

$$e_1(t) - 2e_2(t) - e_3(t) = 0$$
  $t \ge 2$ 

But 
$$x_3(t) = \hat{x}_3(t) + e_3(t)$$
  

$$= \hat{x}_3(t) + e_1(t) - 2e_2(t) \qquad \text{for } t \ge 2$$

$$= \hat{x}_3(t) + y_1(t) - \hat{x}_1(t) - [y_2(t) - 2\hat{x}_2(t)] \qquad \text{for } t \ge 2$$

We know  $\hat{x}_1(t)$ ,  $\hat{x}_2(t)$ , and  $\hat{x}_3(t)$  from the solution of the observer equation (4.59). Hence  $x_3(t)$  is reconstructed after 2 units of time.

The tradeoff between using the delay feedback observer (4.59) and the usual reduced order observer is clear. Delay feedback allows us to determine the state exactly after 2 units of time. However, the delay feedback observer must be 3 dimensional.

#### CHAPTER 5

## OPTIMAL FILTERING OF LINEAR AND NONLINEAR STOCHASTIC DELAY SYSTEMS

## 5.1 Introduction

After studying various deterministic aspects of the theory of delay systems in the previous chapters, we turn our attention to stochastic problems. In this chapter, we study the filtering problem for stochastic delay systems of the form

$$dx(t) = f(x_t, t)dt + H'(t)dw_1(t)$$
$$x(\theta) = x_0(\theta) \qquad \theta \varepsilon[-\tau, 0]$$
(5.1)

$$dz(t) = h(x_t, t)dt + N(t)dw_2(t)$$
  

$$z(s) = w_2(s) = 0 \qquad s \le 0$$
(5.2)

We assume the reader to be familiar with the basic elements of stochastic processes [50], and we simply collect the basic definitions here for future reference. All stochastic processes are defined relative to a given probability space  $(\Omega, \mathscr{G}, \mathbb{P})$  and on an interval of the form [0,T]. The system process x(t) takes values in  $\mathbb{R}^n$ , the observation process z(t) in  $\mathbb{R}^p$ . For simplicity, we take  $w_1(t)$  and  $w_2(t)$  to be standard Wiener processes in  $\mathbb{R}^m$  and  $\mathbb{R}^p$  respectively, completely independent of each other. The initial function  $x_0$  is taken to be some random function on  $[-\tau,0]$ , completely independent of  $w_1(t)$  and  $w_2(t)$ . The maps f and h are functionals, possibly nonlinear, defined on  $\mathscr{C}x[0,T]$ . H(t) and N(t)are mxn and pxp matrix-valued continuous functions respectively. Furthermore N(t) is assumed to be symmetric and positive definite.

We shall write  $dw(t) = H'(t)dw_1(t)$  and  $dv(t) = N(t)dw_2(t)$ . Define Q(t) = H'(t)H(t), R(t) = N<sup>2</sup>(t). With this notation, w(t) is an "unnormalized" Wiener process in R<sup>n</sup>, with

cov[w(t); w(s)] = 
$$\int_{0}^{\min(t,s)} Q(u) du$$

Similarly, v(t) is an "unnormalized" Wiener process in  $\mathbb{R}^p$ , with  $cov[v(t); v(s)] = \int_0^{\min(t,s)} \mathbb{R}(u) du$ . This notation will be used

throughout this chapter.

The basic filtering problem is to estimate some function  $\phi$  of x(t) given the observations z(s),  $0 \le s \le t$ . It is well-known that the optimal estimate with respect to a large class of criteria, for example minimum mean square error, is the conditional expectation  $E\{\phi[x(t)]|z^t\}$ , where  $z^t$  denotes the  $\sigma$ -algebra generated by the observations z(s),  $0 \le s \le t$ . We shall also write  $E\{\phi(x(t))|z^t\}$  as  $\hat{\phi}[x(t)]$  or  $E^t\{\phi[x(t)]\}$ . Our objective in this chapter is to obtain formulas for  $\hat{\phi}[x(t)]$  in the case of delay systems.

The nonlinear filtering problem has been extensively studied by many authors. In particular, Fujisaki, et al., [51] have given a stochastic differential equation for the evolution of  $\hat{\phi}[\mathbf{x}(t)]$ for rather general stochastic systems, which includes our delay model. Specifically they showed that for the delay system (5.1) and (5.2)

$$d\hat{\phi}[\mathbf{x}(t)] = \widehat{\mathscr{L}_{t}\phi[\mathbf{x}(t)]}dt + [\widehat{\phi}[\mathbf{x}(t)]\widehat{h'}(\mathbf{x}_{t},t) - \widehat{\phi}[\mathbf{x}(t)]\widehat{h'}(\mathbf{x}_{t},t)] \cdot \mathbf{R}^{-1}(t)[dz(t) - \widehat{h}(\mathbf{x}_{t},t)dt]$$
(5.3)

where  $\mathscr{L}_t$  is a differential operator. If there are no delays in the system, the unknown terms on the right hand side of (5.3) are of the form  $\widehat{\phi[\mathbf{x}(t)]\mathbf{h}[\mathbf{x}(t),t]}$  and, assuming suitable differentiability conditions, one can write another stochastic differential equation for  $\widehat{\phi[\mathbf{x}(t)]\mathbf{h}[\mathbf{x}(t),t]}$ , leading, in general, to a countably infinite-order system of moment equations. While there is no known method of analyzing this infinite set of equations, at least the way in which they arise is clear. One can thus use approximation techniques to study these equations, and develop suboptimal filtering schemes.

Unfortunately, when there are delays in the system, the above procedure does not go through. The reason is that one of the terms on the right hand side of (5.3) is  $\widehat{\phi[\mathbf{x}(t)]\mathbf{h}(\mathbf{x}_t,t)}$ , and since h is a functional on the segment  $\mathbf{x}_t$ , it is not clear how one can develop a stochastic differential equation for  $\widehat{\phi[\mathbf{x}(t)]\mathbf{h}(\mathbf{x}_t,t)}$  using (5.3). These difficulties motivated the development of a representation theorem for a functional of  $\mathbf{x}_t$ . Using this representation, we derive stochastic differential equations for certain functionals f and h. The linear case with Gaussian distributions is then studied in detail. Stochastic differential equations for the conditional mean as well as partial differential equations for the conditional covariance are derived. This will prepare the way for the filter stability discussions in the next chapter. We will also discuss some of the basic difficulties in deriving stochastic differential equations for general nonlinear functionals  $\phi(\mathbf{x}_t)$ .

-96-

# 5.2 Existence, Uniqueness and Other Properties of Solutions of Stochastic Functional Differential Equations

In order for our estimation problem to be well-defined, we need conditions which guarantee existence and uniqueness of solutions to the functional stochastic differential equations (5.1) and (5.2). Since (5.1) can be solved independently of (5.2), we shall be mainly concerned with conditions to be imposed on (5.1). Conditions on (5.2) will be stated at the end of this section.

Existence and uniqueness of solutions to stochastic functional differential equations of the type (5.1) have been studied by Ito and Nisio [52], and Fleming and Nisio [53]. We state their results here. Assume

- (A.1)  $f(\phi,t)$  is measurable and continuous on  $\mathscr{C}x[0,T]$ ;
- (A.2) there exists a bounded measure  $\Gamma$  on  $[-\tau, 0]$  and a positive constant K such that

$$|f(\phi,t)-f(\psi,t)| \leq K \int_{-\tau}^{0} |\phi(s)-\psi(s)| d\Gamma(s)$$

(A.3) on the interval  $[-\tau, 0]$ , x(t) is continuous w.p.1 with  $E|x(\theta)|^4 < \infty, -\tau \le \theta \le 0.$ 

Proposition 5.2.1 Assume (A.1) to (A.3) are satisfied. Then there exists a unique solution to (5.1) which is continuous w.p.1 and has bounded second moment, and

$$E|x(t)|^4 \leq \gamma e^{\gamma t}$$
 for some positive  $\gamma < \infty$ .

Furthermore  $x_t$  is a Markov process.

The proof of Proposition 5.2.1 can be found in Ito and Nisio [52], or Fleming and Nisio [53], and Kushner [54].

Let us now study some of the properties of linear functional stochastic differential equations which are relevant to our later investigations. We consider the equation

$$dx(t) = a(x_{t}, t)dt + dw(t)$$
 (5.4)

there  $a(x_{t},t)$  is given by the Stieltjes integral

$$a(x_{t},t) = \int_{-\tau}^{0} d_{\theta} A(t,\theta) x(t+\theta)$$

Here,  $A(t,\theta)$  is a function on RxR, jointly measurable in  $(t,\theta)$ , continuous in t, of bounded variation in  $\theta$  for each t, with Var  $A(t,\cdot) \leq m(t)$ , a locally integrable function on  $\mathbb{R}^n$ , where Var  $[-\tau,0]$   $[-\tau,0]$   $[-\tau,0]$ means the total variation in  $[-\tau,0]$ . Furthermore,  $A(t,\theta) = 0$  for  $\theta \geq 0$ ,  $A(t,\theta) = A(t,-\tau)$  for  $\theta \leq -\tau$ , and it is continuous from the left in  $\theta$  on  $(-\tau,0)$ . We shall prove a type of variation of constants formula for (5.4). We first prove a lemma which will be used later and may be of some independent interest.

Lemma 5.2.1 Let G(t,s) be a deterministic function on RxR such that  $\int_{t_0}^{T} |G(t,s)|^2 ds < \infty, \text{ for all } t \in [t_0,T], G(t,s) \text{ is continuously differ-}$ entiable in t, and that  $\int_{t_0}^{T} \int_{t_0}^{T} |\dot{G}(t,s)|^2 dt \, ds < \infty(\dot{G}(t,s) = \frac{\partial}{\partial t} G(t,s)).$ Then the Wiener integral  $g(t) = \int_{t_0}^{t} G(t,s) dw(s)$  has a differential on  $[t_0,T]$ 

$$dg(t) = \int_{t_0}^{t} \dot{G}(t,s)dw(s)dt + G(t,t)dw(t)$$
(5.5)

<u>Proof</u>: Proving (5.5) is equivalent to proving that for any  $t_1$ ,  $t_2$  in  $[t_0,T]$ ,

$$g(t_{2}) - g(t_{1}) = \int_{t_{1}}^{t_{2}} \int_{t_{0}}^{t} G(t,s) dw(s) dt + \int_{t_{1}}^{t_{2}} G(t,t) dw(t)$$
(5.6)

By the assumption of square integrability of G(t,s), the Fubini type theorem for Wiener and Lebesgue integrals [50], is valid, and so

$$\int_{t_{1}}^{t_{2}} \int_{t_{0}}^{t} \dot{G}(t,s) dw(s) dt = \int_{t_{0}}^{t_{1}} \int_{t_{1}}^{t_{2}} \dot{G}(t,s) dt dw(s) + \int_{t_{1}}^{t_{2}} \int_{s}^{t_{2}} \dot{G}(t,s) dt dw(s) = \int_{t_{0}}^{t_{1}} G(t_{2},s) dw(s) - \int_{t_{0}}^{t_{1}} G(t_{1},s) dw(s) + \int_{t_{1}}^{t_{2}} G(t_{2},s) dw(s) - \int_{t_{1}}^{t_{2}} G(s,s) dw(s)$$

The right hand side of (5.6) now yields

$$\int_{t_0}^{t_2} G(t_2, s) dw(s) - \int_{t_0}^{t_1} G(t_1, s) dw(s)$$

which is precisely the left hand side of (5.6).

Using Lemma 5.2.1, we can easily obtain a variation of constants formula for (5.4). Let  $\Phi(t,s)$  be the fundamental matrix associated

with (5.4), i.e., that unique matrix which is absolutely continuous in t, essentially bounded in s, and such that

$$\frac{\partial \Phi(t,s)}{\partial t} = \int_{-\tau}^{0} d_{\theta} A(t,\theta) \Phi(t+\theta,s) \qquad t \ge s$$
$$\Phi(t,s) = \begin{cases} 0 & \text{for } s-\tau \le t \le s \\ 1 & \text{for } s-\tau \le t \le s \end{cases}$$

A discussion of the properties of  $\Phi(t,s)$  may be found in Hale [2].

<u>Theorem 5.2.1</u> The solution x(t) of (5.4) can be written as

$$x(t) = \Phi(t,0)x_{0}(0) + \int_{-\tau}^{0} d_{\beta} \left\{ \int_{0}^{\tau} \Phi(t,s)A(s,\beta-s)ds \right\} x_{0}(\beta) + \int_{0}^{t} \Phi(t,s)dw(s)$$
(5.7)

<u>Proof</u>: The fundamental matrix  $\Phi(t,s)$  satisfies the conditions required in Lemma 5.2.1. We may therefore apply Lemma 5.2.1 to the right hand side of (5.7) to obtain

$$dx(t) = \int_{-\tau}^{0} d_{\theta}A(t,\theta)\Phi(t+\theta,0)x_{0}(0)dt$$

$$+ \int_{-\tau}^{0} d_{\beta}\left\{\int_{0}^{\tau}\int_{-\tau}^{0} d_{\theta}A(t,\theta)\Phi(t+\theta,s)A(s,\beta-s)ds\right\}x_{0}(\beta)$$

$$+ \int_{0}^{t} \int_{-\tau}^{0} d_{\theta}A(t,\theta)\Phi(t+\theta,s)dw(s)dt + dw(t)$$

Applying the unsymmetric Fubini theorem of Cameron and Martin [55], we obtain

$$dx(t) = \int_{-\tau}^{0} d_{\theta} A(t, \theta) x(t+\theta) dt + dw(t)$$

so that (5.7) is indeed a solution of (5.4).

<u>Remark 5.2.1</u> Similar variation of constants formula for linear stochastic functional differential equations have been obtained by Lindquist [56]. All are related to the representation of solutions of linear functional differential equations given by Banks [57].

The mean of x(t),  $\dot{x}(t)$ , can easily be seen to satisfy

$$\frac{d}{dt} \overline{x}(t) = \int_{-\tau}^{0} d_{\theta} A(t, \theta) \overline{x}(t+\theta)$$

$$\overline{x}(\theta) = \overline{x}_{0}(\theta) \qquad \theta \varepsilon[-\tau, 0] \qquad (5.8)$$

whose solution is given by

$$\overline{\mathbf{x}}(t) = \Phi(t,0)\overline{\mathbf{x}}_{0}(0) + \int_{-\tau}^{0} d_{\beta} \left\{ \int_{0}^{\tau} \Phi(t,s)A(s,\beta-s)ds \right\} \overline{\mathbf{x}}_{0}(\beta) \quad (5.9)$$

As for the covariance associated with the solution of (5.6), we shall be interested not only in the covariance of x(t), i.e.,  $E\{[x(t)-\overline{x}(t)][x(t)-\overline{x}(t)]'\}$ , but also in the covariance "operator"  $\Sigma(t,\theta,\xi) = E\{[x(t+\theta)-\overline{x}(t+\theta)][x(t+\xi)-\overline{x}(t+\xi)]'\}, -\tau \leq \theta, \xi \leq 0.$ Heuristically, this corresponds to the covariance of  $x_t$ . We first compute it and then give it a more precise interpretation in terms of the characteristic functional of the  $\mathscr{C}$ -valued random variable  $x_t$ .

Let  $e(t) = x(t) - \overline{x}(t)$  then from (5.7) and (5.9), and using the fact that  $\Phi(t,s) = 0$ , t < s, we get that

$$e(t+\theta) = \Phi(t,0)e(0) + \int_{-\tau}^{0} d_{\beta} \left\{ \int_{0}^{\tau} \Phi(t+\theta,s)A(s,\beta-s)ds \right\} e(\beta) + \int_{0}^{t} \Phi(t+\theta,s)dw(s)$$

Using the independence of  $x_0$  and w(t), we get

$$\Sigma(t,\theta,\xi) = E\{e(t+\theta)e'(t+\xi)\}$$

$$= \Phi(t,0)\Sigma(0,0,0)\Phi'(t,0) + \int_{-\tau}^{0} d_{\beta} \left\{ \int_{0}^{\tau} \Phi(t+\theta,s)A(s,\beta-s)d\alpha \right\}$$

$$\cdot \Sigma(0,\beta,0)\Phi'(t,0)$$

$$+ \int_{-\tau}^{0} \Phi(t,0)d_{\beta} \int_{0}^{\tau} \Sigma(0,0,\beta)A'(s,\beta-s)\Phi'(t+\xi,s)ds$$

$$+ \int_{-\tau}^{0} d\alpha \int_{-\tau}^{0} d_{\beta} \left\{ \int_{0}^{\tau} \Phi(t+\theta,s)A(s,\beta-s)ds \right\} \Sigma(0,\beta,\alpha)$$

$$\cdot \left\{ \int_{0}^{\tau} A'(s,\eta-s)\Phi'(t+\xi,s)ds \right\}$$

$$+ \int_{0}^{t} \Phi(t+\theta,s)Q(s)\Phi'(t+\xi,s)ds \qquad (5.10)$$

If we view  $x_t$  as a  $\mathscr{C}$ -valued random variable, then its characteristic functional  $\phi(y, x_t)$  is defined by

$$\phi(y, x_t) = E \exp i \langle y, x_t \rangle$$

where y is an element of  $\mathscr{C}^*$ , the dual space of  $\mathscr{C}$ , which is the space of functions of bounded variations on  $[-\tau, 0]$  with y(0) = 0, and <,> is the pairing between  $\mathscr{C}^*$  and  $\mathscr{C}$  defined by

$$\langle y, x \rangle = \int_{-\tau}^{0} dy(\theta) x(\theta)$$

Recognizing that for each t, <y,  $x_t > = \int_{-\tau}^{0} dy(\theta) x(t+\theta)$  is a Gaussian

random variable, we can easily evaluate

$$\phi(y, x_t) = e^{i < y, \overline{x_t} > -1/2 Q_t(y)}$$
(5.11)

where  $\overline{x}_{t}$  is given by (5.9) and  $Q_{t}(y)$  is a quadratic form in y given by

$$Q_{t}(y) = \int_{-\tau}^{0} \int_{-\tau}^{0} dy(\theta) \Sigma(t,\theta,\xi) dy'(\xi)$$
(5.12)

The form of  $\phi(y, x_t)$  implies that we may interpret  $x_t$  as a  $\mathscr{C}$ -valued Gaussian random variable for eact t, with  $\Sigma(t, \theta, \xi)$  as its covariance operator. These considerations will be useful later in section 5.5 when we study the linear filtering problem for delay systems.

We now state conditions concerning eq. (5.2).

(A.4)  $h(\phi,t)$  is measurable and continuous on  $\mathscr{C}x[0,T]$ .

(A.5) 
$$\int_0^T E[h(x_t,t)'h(x_t,t)]dt < \infty$$

We note that (A.1) - (A.5) are not the weakest assumptions for which our results are valid. However, we have not striven for more generality because this will only introduce technical complications without adding insight into the optimal filter structure for delay systems. From now on, (A.1) to (A.5) will be assumed to hold.

# 5.3 <u>A Representation Theorem for Conditional Moment</u> <u>Functionals</u>

In this section, we derive a representation theorem for the conditional expectation of functionals of the form  $\phi(x_t)$  given the

-103-

observations z(s),  $0 \le s \le t$ . As before, we shall denote  $\sigma\{z(s), 0 \le s \le t\}$ by  $z^t$ , and we shall call objects of the form  $E\{\phi(x_t) | z^t\}$  <u>conditional</u> <u>moment functionals</u>. Our approach makes use of the Girsanov measure transformation technique as given in Wong [58]. Using the representation theorem so derived, stochastic differential equations can be obtained for suitably smooth functionals  $\phi$ , f, and h.

We first note that we can rewrite (5.2) as

$$N^{-1}(t)dz(t) = N^{-1}(t)h(x_{t},t)dt + dw_{2}(t)$$
(5.13)

Clearly z(s),  $0 \le s \le t$  and  $\int_0^s N^{-1}(\sigma) dz(\sigma) \stackrel{\Delta}{=} z_1(s), 0 \le s \le t$ 

generate the same  $\sigma$ -algebra. Define  $h_1(x_t,t) = N^{-1}(t)h(x_t,t)$ . In deriving the representation theorem, we shall use (5.13), i.e.,  $z_1(t)$  and  $h_1(x_t,t)$ , rather than (5.2). The reason for introducing  $z_1(t)$  and  $h_1(x_t,t)$  is that the absolute continuity results given in Lemma 5.3.1 are phrased in terms of standard Wiener processes. Define a new measure  $P_0$  on  $(\Omega, \mathscr{G})$  by the formula

$$\frac{dP_{o}}{dP} = \exp \left[-\int_{0}^{T} h_{1}(x_{t},t)dw_{2}(t) - \frac{1}{2}\int_{0}^{T} h_{1}(x_{t},t)'h_{1}(x_{t},t)dt\right]$$
(5.14)

<u>Lemma 5.3.1</u>  $P_0$  is a probability measure with the following properties:

- (a) Under  $P_0$ ,  $z_1(s)$ ,  $0 \le s \le T$  has components that are independent standard Wiener processes.
- (b) Under  $P_0$ , the processes  $z_1(t) \quad 0 \le t \le T$  and  $x(t), \quad 0 \le t \le T$ are independent.
- (c) The restriction of P to  $\sigma\{x(s), -\tau \leq s \leq T \text{ is the same as }$

the corresponding restriction of P.

(d) 
$$P \ll P_{o}$$
 and  

$$\frac{dP}{dP_{o}} = \exp\left[\int_{0}^{T} h_{1}(x_{t},t)'dz_{1}(t) - \frac{1}{2}\int_{0}^{T} h_{1}'(x_{t},t)h_{1}(x_{t},t)dt\right]$$
(5.15)

<u>Proof</u>: The conditions imposed on the processes involved give rise to a virtually identical situation to that of Proposition 5.1, Chapter 6 of Wong [58]. The proof there applies without change to our case.

Let us denote integration with respect to P and  $P_0$  by the expectation operators E and E<sub>0</sub>, respectively. The reader should note that the true measure relevant to our estimation problem is P. The measure P<sub>0</sub> is strictly a device for obtaining the representation theorem.

Let us write  $\mathscr{F}_t$  for the least  $\sigma$ -algebra containing  $\sigma\{z(s), 0 \le s \le t\}$ and  $\sigma\{x(s), -\tau \le s \le t\}$ . It is well-known [59] that the function  $L_t$ defined by

$$L_{t} = \exp\left[\int_{0}^{t} h_{1}'(x_{s},s)dz_{1}(s) - \frac{1}{2} \int_{0}^{t} h_{1}'(x_{s},s)h_{1}(x_{s},s)ds\right]$$
  
=  $\exp\left[\int_{0}^{t} h'(x_{s},s)R^{-1}(s)dz(s) - \frac{1}{2} \int_{0}^{t} h'(x_{s},s)R^{-1}(s)h(x_{s},s)ds\right]$   
(5.16)

is a  $(\mathscr{F}_t, \mathsf{P}_o)$  martingale. Let  $\phi$  be a real-valued measurable function on  $\mathscr{C}$ , with the property that  $E|\phi(\mathbf{x}_t)| < \infty$ . Then we have the following (compare [60])

<u>Lemma 5.3.2</u> The conditional expectation of  $\phi(x_t)$  given  $z^t$  can be written as

$$E[\phi(x_{t})|z^{t}] = \frac{E_{o}[\phi(x_{t})L_{t}|z^{t}]}{E_{o}[L_{t}|z^{t}]} \quad a.s. P$$
 (5.17)

<u>Proof</u>: Since  $\frac{dP}{dP_o} = L_T$ , it is known [61] that  $E[\phi(x_t)|z^t] = \frac{E_o[\phi(x_t)L_T|z^t]}{E_o[L_T|z^t]}$ 

Using the properties of conditional expectations and the fact that  $L_t$  is a ( $\mathcal{F}_t$ ,  $P_o$ ) martingale, we can write

$$E_{o}[\phi(\mathbf{x}_{t})\mathbf{L}_{T}|\mathbf{z}^{t}] = E_{o}\{E_{o}[\phi(\mathbf{x}_{t})\mathbf{L}_{T}|\mathcal{F}_{t}]|\mathbf{z}^{t}\}$$
$$= E_{o}\{\phi(\mathbf{x}_{t})E_{o}[\mathbf{L}_{T}|\mathcal{F}_{t}]|\mathbf{z}^{t}\}$$
$$= E_{o}\{\phi(\mathbf{x}_{t})\mathbf{L}_{t}|\mathbf{z}^{t}\}$$

Similarly,  $E_o[L_T|z^t] = E_o[L_t|z^t]$ , and the lemma follows.

In what follows, we shall omit the qualification of almost sure equality for conditional expectations. Such qualifications will always be considered understood when conditional expectations are involved.

An application of the exponential formula [59] to the functional  $\rm L_{_{\rm f}}$  yields

$$L_{t} = 1 + \int_{0}^{t} L_{s} h'(x_{s}, s) R^{-1}(s) dz(s)$$
 (5.18)

We shall recast the numerator of (5.17) into a more convenient form for later calculations. To that end, we make one more assumption: (A.6)  $\int_0^T E |\phi(x_t)h(x_t,t)|^2 dt < \infty.$  From now on, (A.6) will be assumed to hold. We then have

Lemma 5.3.3

$$E_{o}\left[\int_{0}^{t} \phi(x_{t})L_{s}h'(x_{s},s)R^{-1}(s)dz(s)|z^{t}\right]$$
  
= 
$$\int_{0}^{t} E_{o}[\phi(x_{t})L_{s}h'(x_{s},s)|z^{t}]R^{-1}(s)dz(s)$$
(5.19)

<u>Proof</u>: An almost identical lemma is proved in Zakai [62]. His proof can be adapted to our case without difficulty.

Lemma 5.3.4 We can then write the numerator of (5.17) as

$$E_{o}[\phi(x_{t})L_{t}|z^{t}] = E_{o}[\phi(x_{t})] + \int_{0}^{t} E_{o}\{E_{o}[\phi(x_{t})|x_{s}]L_{s}h'(x_{s},s)|z^{s}\}R^{-1}(s)dz(s)$$
(5.20)

Proof: Applying (5.18), we can write

$$E_{o}[\phi(x_{t})L_{t}|z^{t}] = E_{o}[\phi(x_{t})(1 + \int_{0}^{t} L_{s}h'(x_{s},s)R^{-1}(s)dz(s))|z^{t}]$$
  
=  $E_{o}[\phi(x_{t})|z^{t}] + E_{o}[\int_{0}^{t}\phi(x_{t})L_{s}h'(x_{s},s)R^{-1}(s)dz(s)|z^{t}]$   
(5.21)

Since under  $P_o$ , x(t) and z(t) are completely independent,  $E_o[\phi(x_t)|z^t] = E_o[\phi(x_t)]$ . Combined with (5.19), we get

$$E_{o}[\phi(x_{t})L_{t}|z^{t}] = E_{o}[\phi(x_{t})] + \int_{0}^{t} E_{o}[\phi(x_{t})L_{s}h'(x_{s},s)|z^{t}]R^{-1}(s)dz(s)$$
Now 
$$\int_{0}^{t} E_{o}[\phi(x_{t})L_{s}h'(x_{s},s)|z^{t}]R^{-1}(s)dz(s)$$

$$= \int_{0}^{t} E_{o}\{E_{o}[\phi(x_{t})L_{s}h'(x_{s},s)|z^{t}Vx^{s}]|z^{t}R^{-1}(s)dz(s)$$

-107-

$$= \int_{0}^{t} E_{o} \{ E_{o} [\phi(x_{t}) | z^{t} v x^{s}] L_{s} h'(x_{s}, s) | z^{t} \} R^{-1}(s) dz(s)$$
(5.22)

since  $L_{s}h'(x_{s},s)$  is measurable with respect to the  $\sigma$ -algebra  $z^{t}vx^{s}$ , the smallest  $\sigma$ -algebra containing  $z^{t}$  and  $\sigma\{x(t), -\tau \leq t \leq s\}$ . Using the Markov property of  $x_{t}$  and the independence of x(t) and z(t) under  $P_{o}$ , we get

$$\int_{0}^{t} E_{o} \{ E_{o} [\phi(x_{t}) | z^{t} v x^{s}] L_{s} h'(x_{s}, s) | z^{t} \} R^{-1}(s) dz(s)$$

$$= \int_{0}^{t} E_{o} \{ E_{o} [\phi(x_{t}) | x_{s}] L_{s} h'(x_{s}, s) | z^{t} \} R^{-1}(s) dz(s)$$

$$= \int_{0}^{t} E_{o} \{ E_{o} [\phi(x_{t}) | x_{s}] L_{s} h'(x_{s}, s) | z^{s} \} R^{-1}(s) dz(s)$$
(5.23)

The lemma follows from (5.21), (5.22) and (5.23).

In order to get the representation for the conditional expectation  $E[\phi(x_t)|z^t]$ , we need to evaluate the denominator of (5.17). This is done in the following lemma.

Lemma 5.3.5 
$$E_o(L_t|z^t)^{-1} = 1 - \int_0^t E_o(L_s|z^s)^{-1}E[h'(x_s,s)|z^s]R^{-1}(s)dv(s)$$
(5.24)

where  $v(t) = z(t) - \int_0^t E[h(x_s,s)|z^s]ds$  is the innovations.

Proof: Since 
$$L_t = 1 + \int_0^t L_s h'(x_s, s) R^{-1}(s) dz(s)$$
 Lemma 5.3.3 yields  
 $E_o(L_t | z^t) = 1 + \int_0^t E_o[L_s h'(x_s, s) | z^s] R^{-1}(s) dz(s)$
Applying the Ito differential rule to the function  $E_0(L_t|z^t)^{-1}$  gives

$$dE_{o}(L_{t}|z^{t})^{-1} = -E_{o}(L_{t}|z^{t})^{-2}E_{o}(L_{t}h'(x_{t},t)|z^{t})R^{-1}(t)dz(t)$$

$$+ E_{o}(L_{t}|z^{t})^{-3}E_{o}[L_{t}h'(x_{t},t)|z^{t}]R^{-1}(t)E_{o}[L_{t}h'(x_{t},t)|z^{t}]dt$$

$$= - E_{o}(L_{t}|z^{t})^{-1}E[h(x_{t},t)|z^{t}]dz(t)$$

$$+ E_{o}(L_{t}|z^{t})^{-1}E[h'(x_{t},t)|z^{t}]R^{-1}(t)E[h(x_{t},t)|z^{t}]dt$$

$$= - E_{o}(L_{t}|z^{t})^{-1}E[h(x_{t},t)|z^{t}]R^{-1}(t)E[h(x_{t},t)|z^{t}]dt$$

Since (5.25) is equivalent to (5.24), the lemma is proved.

From Lemma 5.3.2, we know that  $E[\phi(x_t)|z^t]$  is given by the product of the right hand sides of (5.20) and (5.24). In order not to complicate the computations too much, we evaluate one of the terms separately and state it as a lemma.

$$\frac{\text{Lemma 5.3.6}}{\text{Lemma 5.3.6}} = \int_{0}^{t} E_{0} \{E_{0}[\phi(x_{t})|x_{s}]L_{s}h'(x_{s},s)|z^{s}\}R^{-1}(s)dz(s)$$

$$= \int_{0}^{t} E^{s} \{E_{0}[\phi(x_{t})|x_{s}][h'(x_{s},s)-\hat{h}'(x_{s},s)]\}R^{-1}(s)dv(s)$$

$$+ \int_{0}^{t} E_{0}(L_{s}|z^{s})^{-1}E_{0}[\phi(x_{t})]\hat{h}'(x_{s},s)R^{-1}(s)dv(s)$$
(5.26)

Proof: For convenience, define the functions

$$p(t,s) = E_{o} \{ E_{o}[\phi(x_{t}) | x_{s}] L_{s}h'(x_{s},s) | z^{s} \} R^{-1}(s)$$

$$q(s) = E_{o}(L_{s} | z^{s})^{-1} \hat{h}'(x_{s},s) R^{-1}(s)$$

Using (5.24), we can write the left hand side of (5.26) as

$$\int_{0}^{t} p(t,s)dz(s) - \int_{0}^{t} p(t,s)dz(s) \int_{0}^{t} q(s)dv(s)$$
(5.27)

By properties of Ito integrals [63], (5.27) can be rewritten as

$$\int_0^t p(t,s)dz(s) - \int_0^t \int_0^s p(t,u)dz(u)q(s)dv(s)$$

$$-\int_{0}^{t} p(t,s) \int_{0}^{s} q(u) dv(u) dz(s) - \int_{0}^{t} p(t,s)q'(s) ds$$
$$= \int_{0}^{t} E_{0}(L_{s}|z^{s})^{-1}p(t,s) dz(s) - \int_{0}^{t} p(t,s)q'(s) ds$$
$$-\int_{0}^{t} \int_{0}^{s} p(t,u) dz(u)q(s) dv(s)$$

(where we have used Lemma 5.3.5)

$$= \int_{0}^{t} E_{o}(L_{s}|z^{s})^{-1}p(t,s)dv(s) - \int_{0}^{t} \int_{0}^{s} p(t,u)dz(u)q(s)dv(s)$$
(5.28)

$$Now - \int_{0}^{t} \int_{0}^{s} p(t,u)dz(u)q(s)dv(s)$$

$$= -\int_{0}^{t} \{E_{o}[\phi(x_{t})] + \int_{0}^{s} p(t,u)dz(u)\}q(s)dv(s)$$

$$+ \int_{0}^{t} E_{o}[\phi(x_{t})]q(s)dv(s)$$

$$= -\int_{0}^{t} \{E_{o}[E_{o}(\phi(x_{t})|x_{s})] + \int_{0}^{s} E_{o}[E_{o}(\phi(x_{t})|x_{s}vx_{u})|x_{u}] \cdot \cdot L_{u}h'(x_{u},u)|z^{u}]R^{-1}(u)dz(u)\}q(s)dv(s)$$

$$+ \int_{0}^{t} E_{o}[\phi(x_{t})]q(s)dv(s)$$

(where we have used the smoothing property of conditional expectations)

$$= - \int_{0}^{t} \{ E_{o}[E_{o}(\phi(x_{t})|x_{s})] + \int_{0}^{s} E_{o}[E_{o}(\phi(x_{t})|x_{s})|x_{u}]L_{u}h'(s_{u},u)|z^{u}] \cdot R^{-1}(u)dz(u) \}q(s)dv(s) + \int_{0}^{t} E_{o}[\phi(x_{t})]q(s)dv(s)$$

(using the Markov property of  $x_t$ )

$$= - \int_{0}^{t} E^{s}[E_{o}(\phi(x_{t})|x_{s})]E^{s}[h'(x_{s},s)]R^{-1}(s)dv(s) + \int_{0}^{t} E_{o}(L_{s}|z^{s})^{-1}E_{o}[\phi(x_{t})]\hat{h}'(x_{s},s)R^{-1}(s)dv(s)$$
(5.29)

(where we have successively used Lemma 5.3.4, the definition of q(s), and Lemma 5.3.2). Combining (5.28) and (5.29) gives the conclusion of the lemma.

We are now ready to state the representation theorem for the conditional moment functionals.

<u>Theorem 5.3.1</u> Suppose  $\phi: \mathscr{C} \to \mathbb{R}$  is such that  $\mathbb{E}|\phi(\mathbf{x}_t)| < \infty$ , and (A.1) - (A.6) hold. Then we have the following representation for the conditional expectation of  $\phi$  given  $\mathbf{z}^t$ :

$$\hat{\phi}(\mathbf{x}_{t}) = E_{o}[\phi(\mathbf{x}_{t})] + \int_{0}^{t} E^{s} \{E_{o}[\phi(\mathbf{x}_{t})|\mathbf{x}_{s}][h'(\mathbf{x}_{s},s)-\hat{h}'(\mathbf{x}_{s},s)]\} \cdot \mathbf{x}_{R}^{-1}(s) d\nu(s)$$
(5.30)

Proof: From Lemmas 5.3.2 and 5.3.4, we know that

$$\hat{\phi}(\mathbf{x}_{t}) = E_{o}[\phi(\mathbf{x}_{t})]E_{o}(\mathbf{L}_{t}|z^{t})^{-1} + \int_{0}^{t} E_{o}\{E_{o}[\phi(\mathbf{x}_{t})|\mathbf{x}_{s}]\mathbf{L}_{s}h'(\mathbf{x}_{s},s)|z^{s}\} \cdot \mathbf{R}^{-1}(s)dz(s)E_{o}(\mathbf{L}_{t}|z^{t})^{-1}$$
(5.31)

From Lemma 5.3.5, we conclude that

$$E_{o}[\phi(x_{t})]E_{o}(L_{t}|z^{t})^{-1}$$
  
=  $E_{o}[\phi(x_{t})] - \int_{0}^{t} E_{o}[\phi(x_{t})]E_{o}(L_{s}|z^{s})^{-1}\hat{h}'(x_{s},s)R^{-1}(s)dv(s)$   
(5.32)

Substituting (5.32) and (5.26) into (5.31) gives (5.30). The proof is completed.

<u>Corollary 5.3.1</u> The smoothed estimate  $E^{t}[x(t+\theta)]$ ,  $-\tau \leq \theta < 0$ , is given by

$$E^{t}[x(t+\theta)] = E^{t+\theta}[x(t+\theta)] + \int_{t+\theta}^{t} E^{s}\{E_{0}[x(t+\theta)|x_{s}]\} \cdot [h'(x_{s},s)-\hat{h}'(x_{s},s)] R^{-1}(s) dv(s)$$

$$[h'(x_{s},s)-\hat{h}'(x_{s},s)] R^{-1}(s) dv(s)$$
(5.33)

Proof: Using Theorem 5.3.1,

$$E^{t}[x(t+\theta)] = E_{0}[x(t+\theta)] + \int_{0}^{t} E^{s}\{E_{0}[x(t+\theta)|x_{s}][h'(x_{s},s)-\hat{h}'(x_{s},s)]\}R^{-1}(s)dv(s)$$

$$= E_{o}[x(t+\theta)] + \int_{0}^{t+\theta} E^{S} \{E_{o}[x(t+\theta)|x_{s}][h'(x_{s},s)-\hat{h}'(x_{s},s)]\} \cdot R^{-1}(s) dv(s)$$
$$+ \int_{t+\theta}^{t} E^{S} \{E_{o}[x(t+\theta)|x_{s}][h'(x_{s},s)-\hat{h}'(x_{s},s)]\} R^{-1}(s) dv(s)$$
$$= E^{t+\theta}[x(t+\theta)] + \int_{t+\theta}^{t} E^{S} \{E_{o}[x(t+\theta)|x_{s}][h'(x_{s},s)-\hat{h}'(x_{s},s)]\} \cdot R^{-1}(s) dv(s)$$

 $\cdot R^{-1}(s) dv(s)$ 

which is precisely (5.33).

<u>Remark 5.3.1</u> Theorem 5.3.1 is a generalization of the corresponding representation results for nonlinear filtering of stochastic systems without time delays. While the measure transformation techniques employed here has been used before in connection with systems without delays [58], [62], the present form of the representation appears to be novel even when specialized to systems without delays. The reason for this is that in [58] and [62], the authors were interested in deriving equations for the unnormalized conditional density. They did not, therefore, give explicit formulas for the true conditional moments. In our case, conditional densities for the  $x_t$  process do not even make sense. This motivated us to derive the Representation Theorem 5.3.1 for conditional moment functionals. We feel that the present form of the representation is much more convenient to use when we derive stochastic differential equations for conditional moment functionals. Furthermore, it shows clearly the role played by the infinitesimal generator of the Markov process  $x_t$ . In the next two sections, we shall apply Theorem 5.3.1 to derive stochastic differential equations for the nonlinear and linear filtering problems.

# 5.4 <u>Stochastic Differential Equations for Nonlinear</u> Filtering of Delay Systems

While Theorem 5.3.1 can be thought of as solving our nonlinear filtering problem abstractly, it does not give a recursive solution. That is, for any fixed t, formula (5.30) is valid. However, knowledge of  $E[\phi(x_t)|z^t]$  and the observations z(s),  $t \leq s \leq t + \Delta$ , is not sufficient to determine  $E[\phi(x_{t+\Delta})|z^{t+\Delta}]$ . In fact, we must completely re-process our past observations. For implementation purposes, one would like to obtain a stochastic differential equation for the evolution of  $E[\phi(x_t)|z^t]$ . As is expected, this will require certain smoothness conditions on the functional  $\phi$ . In this section, we shall investigate the following question: under what conditions on  $\phi$  can we obtain a stochastic differential equation for  $E[\phi(x_t)|z^t]$ ?

-114-

Let us first make a few preliminary calculations. Let  $\varepsilon > 0$  be fixed and let  $\hat{h}(x_s,s) = E[h(x_s,s)|z^s]$ . Then

$$\begin{split} & \mathbb{E}[\phi(\mathbf{x}_{t+\epsilon})|\mathbf{z}^{t+\epsilon}] - \mathbb{E}[\phi(\mathbf{x}_{t})|\mathbf{z}^{t}] \\ &= \mathbb{E}_{o}[\phi(\mathbf{x}_{t+\epsilon})-\phi(\mathbf{x}_{t})] + \int_{0}^{t+\epsilon} \mathbb{E}\{\mathbb{E}_{o}[\phi(\mathbf{x}_{t+\epsilon})|\mathbf{x}_{s}][h(\mathbf{x}_{s},s)-\hat{h}(\mathbf{x}_{s},s)]'|\mathbf{z}^{s}\} \cdot \\ & \cdot \mathbb{R}^{-1}(s)d\upsilon(s) \\ &- \int_{0}^{t} \mathbb{E}\{\mathbb{E}_{o}[\phi(\mathbf{x}_{t})|\mathbf{x}_{s}][h(\mathbf{x}_{s},s)-\hat{h}(\mathbf{x}_{s},s)]'|\mathbf{z}^{s}\}\mathbb{R}^{-1}(s)d\upsilon(s) \\ &= \mathbb{E}_{o}[\phi(\mathbf{x}_{t+\epsilon})-\phi(\mathbf{x}_{t})] + \int_{0}^{t} \mathbb{E}\{\mathbb{E}_{o}[\phi(\mathbf{x}_{t+\epsilon})-\phi(\mathbf{x}_{t})|\mathbf{x}_{s}] \cdot \\ & [h(\mathbf{x}_{s},s)-\hat{h}(\mathbf{x}_{s},s)]'|\mathbf{z}^{s}\}\mathbb{R}^{-1}(s)d\upsilon(s) \\ &+ \int_{t}^{t+\epsilon} \mathbb{E}\{\mathbb{E}_{o}[\phi(\mathbf{x}_{t+\epsilon})|\mathbf{x}_{s}][h(\mathbf{x}_{s},s)-\hat{h}(\mathbf{x}_{s},s)]'|\mathbf{z}^{s}\}\mathbb{R}^{-1}(s)d\upsilon(s) \\ &\quad (5.34) \end{split}$$

We will be primarily interested in the limit as  $\varepsilon \rightarrow 0$ . The last term in (5.34) is straightforward to evaluate in that case. The difficult calculation is that of evaluating terms of the form

$$E_{o}[\phi(\mathbf{x}_{t+\varepsilon}) - \phi(\mathbf{x}_{t}) | \mathbf{x}_{s}]$$
  
=  $E_{o}\{E_{o}[\phi(\mathbf{x}_{t+\varepsilon}) - \phi(\mathbf{x}_{t}) | \mathbf{x}_{t} \mathbf{v} \mathbf{x}_{s}] | \mathbf{x}_{s}\}$   
=  $E_{o}\{E_{o}[\phi(\mathbf{x}_{t+\varepsilon}) - \phi(\mathbf{x}_{t}) | \mathbf{x}_{t}] | \mathbf{x}_{s}\}$ 

where we have used the Markov property of  $x_t$ . The evaluation of  $E_0[\phi(x_{t+\varepsilon})-\phi(x_t)|x_t]$  corresponds to the determination of the infinitesimal generator of the Markov process  $x_t$ .

Let us assume for the time being that the functional  $\phi$  is in the domain of the generator,  $\mathscr{A}_t$ , of the Markov process  $x_t$ . Then it is true (see Dynkin [64]) that

$$E_{o}[\phi(\mathbf{x}_{t})|\mathbf{x}_{s}] - \phi(\mathbf{x}_{s}) = \int_{s}^{t} E_{o}[\mathscr{A}_{u}\phi(\mathbf{x}_{u})|\mathbf{x}_{s}]du \qquad (5.35)$$

In this case, we have the following

<u>Theorem 5.4.1</u> Let  $\phi$  satisfy the conditions of Theorem 5.3.1. In addition, let  $\phi(\mathbf{x}_t)$  be in the domain of the infinitesimal generator  $\mathscr{A}_t$  of  $\mathbf{x}_t$  and suppose that  $\int_0^T \mathbf{E} | [\mathscr{A}_t \phi(\mathbf{x}_t)] h(\mathbf{x}_t, t) |^2 dt$ 

Then the functional  $\boldsymbol{\varphi}(\boldsymbol{x}_t)$  satisfies the stochastic differential equation

$$\hat{d\phi}(x_{t}) = E[\mathscr{A}_{t}\phi(x_{t})|z^{t}]dt + E[\phi(x_{t})(h(x_{t},t)-\hat{h}(x_{t},t))'|z^{t}]R^{-1}(t)dv(t)$$
(5.36)

<u>Proof</u>: We start with (5.34). For simplicity, let  $e_h(t) = h(x_t, t) - \hat{h}(x_t, t)$ . Then

$$\begin{split} \int_{0}^{t} & \mathbb{E}\{\mathbb{E}_{o}[\phi(\mathbf{x}_{t+\varepsilon}) - \phi(\mathbf{x}_{t}) | \mathbf{x}_{s}] \mathbf{e}_{h}^{t}(s) | z^{S}\} \mathbb{R}^{-1}(s) d\nu(s) \\ &= \int_{0}^{t} & \mathbb{E}\{\mathbb{E}_{o}\{\mathbb{E}_{o}[\phi(\mathbf{x}_{t+\varepsilon}) - \phi(\mathbf{x}_{t}) | \mathbf{x}_{t}] | \mathbf{x}_{s}\} \mathbf{e}_{h}^{t}(s) | z^{S}\} \mathbb{R}^{-1}(s) d\nu(s) \\ &= \int_{0}^{t} & \mathbb{E}\{\mathbb{E}_{o}\{\int_{t}^{t+\varepsilon} \mathbb{E}_{o}[\mathscr{A}_{u}\phi(\mathbf{x}_{u}) | \mathbf{x}_{t}] du | \mathbf{x}_{s}\} \mathbf{e}_{h}^{t}(s) | z^{S}\} \mathbb{R}^{-1}(s) d\nu(s) \\ &= \int_{0}^{t} & \mathbb{E}\{\int_{t}^{t+\varepsilon} \mathbb{E}_{o}[\mathscr{A}_{u}\phi(\mathbf{x}_{u}) | \mathbf{x}_{s}] \mathbf{e}_{h}^{t}(s) | z^{S}\} du \mathbb{R}^{-1}(s) d\nu(s) \end{split}$$

$$= \int_{t}^{t+\varepsilon} \int_{0}^{t} \mathbb{E} \{ \mathbb{E}_{o} [\mathscr{A}_{u} \phi(\mathbf{x}_{u}) | \mathbf{x}_{s}] \mathbf{e}_{h}^{\dagger}(s) | \mathbf{z}^{s} \} \mathbb{R}^{-1}(s) d\nu(s) du$$
(5.37)

using Lemma 5.3.3.

Next,

Finally

$$E_{o}[\phi(\mathbf{x}_{t+\varepsilon})-\phi(\mathbf{x}_{t})] = E_{o}\{E_{o}[\phi(\mathbf{x}_{t+\varepsilon}) - \phi(\mathbf{x}_{t})|\mathbf{x}_{t}]$$

$$= E_{o}\{\int_{t}^{t+\varepsilon}E_{o}[\mathscr{A}_{u}\phi(\mathbf{x}_{u})|\mathbf{x}_{t}]du\}$$

$$= \int_{t}^{t+\varepsilon}E_{o}[\mathscr{A}_{u}\phi(\mathbf{x}_{u})]du$$

$$= \int_{t}^{t+\varepsilon}E[\mathscr{A}_{u}\phi(\mathbf{x}_{u})|\mathbf{z}^{u}]du-$$

$$- \int_{t}^{t+\varepsilon}\int_{0}^{u}E\{E_{o}[\mathscr{A}_{u}\phi(\mathbf{x}_{u})|\mathbf{x}_{s}]e_{h}^{\prime}(s)|\mathbf{z}^{s}]R^{-1}(s)dv(s)du$$
(5.39)

using the representation in Theorem 5.3.1. Adding (5.37) - (5.39), we get

$$\begin{split} \mathbb{E}[\phi(\mathbf{x}_{t+\varepsilon})|\mathbf{z}^{t+\varepsilon}] &= \mathbb{E}[\phi(\mathbf{x}_{t})|\mathbf{z}^{t}] \\ &= \int_{t}^{t+\varepsilon} \mathbb{E}[\mathcal{A}_{u}\phi(\mathbf{x}_{u})|\mathbf{z}^{u}]du + \int_{t}^{t+\varepsilon} \mathbb{E}[\phi(\mathbf{x}_{u})\mathbf{e}_{h}^{t}(\mathbf{u})|\mathbf{z}^{u}]\mathbf{R}^{-1}(\mathbf{u})dv(\mathbf{u}) \\ &+ \int_{t}^{t+\varepsilon} \int_{s}^{t+\varepsilon} \mathbb{E}\{\mathbb{E}_{o}(\mathcal{A}_{u}\phi(\mathbf{x}_{u})|\mathbf{x}_{s}]\mathbf{e}_{h}^{t}(s)|\mathbf{z}^{s}\}du\mathbf{R}^{-1}(s)dv(s) \\ &- \int_{t}^{t+\varepsilon} \int_{t}^{u} \mathbb{E}\{\mathbb{E}_{o}[\mathcal{A}_{u}\phi(\mathbf{x}_{u})|\mathbf{x}_{s}]\mathbf{e}_{h}^{t}(s)|\mathbf{z}^{s}\}\mathbf{R}^{-1}(s)dv(s)du \end{split}$$
(5.40)

Another application of Lemma 5.3.3 shows that the last two terms of (5.40) add to zero. The proof is completed.

Theorem 5.4.1 is a generalization to systems with delays of the usual formula for conditional moments of ordinary diffusion processes. While the form of the stochastic differential equation is exactly the same as that for diffusion processes, there is a subtle difference. In the diffusion process case, x(t) itself is a Markov process. In the delay case, while x(t) still satisfies a stochastic differential equation, it is no longer a Markov process. This special structure of stochastic delay systems is clearly shown in (5.36), where the infinitesimal generator  $\mathscr{A}_t$  of  $x_t$  plays a crucial role. Thus, in order to apply Theorem 5.4.1, we need to characterize the domain of the infinitesimal generator of the Markov process  $x_t$ . These are functionals for which the limit

-118-

$$\lim_{h \to 0} \frac{1}{h} \{ E[\phi(x_{t+h}) | x_t] - \phi(x_t) \}$$
(5.41)

necessarily exists. This immediately rules out functionals of the form  $\phi(x_t) = \phi[x(t+\theta)]$  for some  $\theta \in (-\tau, 0)$ . For in this case,

$$E[\phi(\mathbf{x}_{t+h}) | \mathbf{x}_{t}] = E\{\phi[\mathbf{x}(t+\theta+h)] | \mathbf{x}_{t}\}$$

$$= \phi[x(t+\theta+h)]$$

whenever  $-\tau \leq \theta + h \leq 0$ . Since the sample path x(t) is not differentiable, the limit in (5.41) does not exist. It is not possible therefore, to derive a stochastic differential equation for a functional of the form  $\phi[x(t+\theta)]$  using Theorem 5.4.1.

There are certain special classes of functionals  $\phi$  which are in the domain of the generator  $\mathscr{A}_t$ . These are those considered by Kushner [54]. We shall simply state these results.

<u>Case 1</u>: Suppose the functional  $\phi(x_t) = \phi[x(t)]$ , and is twice continuously differentiable in its argument, then

$$\mathscr{A}_{t}\phi[x(t)] = f(x_{t},t)'\phi_{x}[x(t)] + \frac{1}{2} tr Q(t)\phi_{xx}(x(t))$$

where  $\phi_x$  is the n-vector whose i<sup>th</sup> component is  $\frac{\partial \phi}{\partial x_i}[x(t)]$ . Substituting into (5.36) gives

$$d\hat{\phi}[x(t)] = E^{t} \{f(x_{t},t)'\phi_{x}[x(t)] + \frac{1}{2} tr Q(t)\phi_{xx}[x(t)]\}dt + E^{t} \{\phi[x(t)][h'(x_{t},t)-\hat{h}'(x_{t},t)]\}R^{-1}(t)dv(t)$$
(5.42)

This is precisely the formula derived in Fujisaki et al., [51] using a different technique. We shall denote the operator  $\mathscr{A}_t$ in this special case by  $\mathscr{L}_t$ . In particular, the conditional mean  $\hat{x}(t)$  satisfies

$$\hat{dx}(t) = E^{t}[f(x_{t},t)]dt + E^{t}\{x(t)[h'(x_{t},t)-\hat{h}'(x_{t},t)]\}R^{-1}(t)dv(t)$$
(5.43)

Equation (5.43) and the formula for the smoothed estimate, (5.33), will be useful in discussing the optimal linear filter.

Case 2: Let 
$$\phi(x_t) = \int_{-\tau}^{0} \psi(\theta) g[x(t+\theta), x(t)] d\theta$$
 where  $\psi$  is continuously differentiable on  $[-\tau, 0]$  and g is twice continuously differentiable in its second argument. Then

$$\mathscr{A}_{t}\phi(\mathbf{x}_{t}) = \dot{\psi}(0)g[\mathbf{x}(t),\mathbf{x}(t)] - \psi(-\tau)g[\mathbf{x}(t-\tau),\mathbf{x}(t)]$$
$$- \int_{-\tau}^{0} \dot{\psi}(\theta)g[\mathbf{x}(t+\theta),\mathbf{x}(t)]d\theta + \int_{-\tau}^{0} \psi(\theta)\mathscr{L}_{t}g[\mathbf{x}(t+\theta),\mathbf{x}(t)]d\theta$$
(5.44)

where  $\mathscr{L}_t$  is the operator defined in Case 1 and acts on g as a function of x(t) only.

<u>Case 3</u>: Let  $\phi(x_t) = D[F(x_t)]$  where D is a twice continuously differentiable real-valued function, and  $F(x_t) = \int_{-\tau}^{0} \psi(\theta)g[x(t+\theta), x(t)]d\theta$ 

is the type of functional described in Case 2. Then

$$\mathscr{A}_{t}\phi(\mathbf{x}_{t}) = \mathbf{D}_{\alpha}(\alpha) \Big|_{\alpha = F(\mathbf{x}_{t})} \mathscr{A}_{t}F(\mathbf{x}_{t}) + \frac{1}{2} \mathbf{D}_{\alpha\alpha}(\alpha) \Big|_{\alpha = F(\mathbf{x}_{t})} \cdot \mathbf{G}$$

where

$$G = \int_{-\tau}^{0} \int_{-\tau}^{0} \psi(\theta)\psi(\eta) \sum_{i,j} g_{\beta_{i}}[x(t+\theta), x(t)]g_{\beta_{j}}[x(t+\theta), x(t)]Q_{ij}(t)d\theta d\eta$$

and  $g_{\beta_i}$  denotes partial differentiation of g with respect to the i<sup>th</sup> component of the second argument.

From the above special cases, we can see that basically we need twice continuous differentiability of  $\phi$  with respect to the dependence on x(t), and Fréchet differentiability with respect to the dependence on the piece of the trajectory  $x_t$ . As discussed before, this rules out functionals of the form  $\phi[x(t+\theta)], \theta \in [-\tau, 0)$ . Hence for nonlinear systems with point delays, any attempt in deriving stochastic differential equations for conditional moment functionals will have to face the difficulty of functionals not being in the domain of the generator of the Markov process  $x_t$ . For example, as in the multipath communication problem mentioned in Chapter 1, if the observation process is of the form

$$dz(t) = \{h_1[x(t)] + h_2[x(t-\tau)]\}dt + dv(t)$$

then for a twice continuously differentiable  $\phi[x(t)]$ , we get

$$d\hat{\phi}[x(t)] = \widehat{\mathscr{Q}_{t}\phi[x(t)]} + [\widehat{\phi}(x(t))h_{1}(x(t)) + \widehat{\phi}(x(t))h_{2}(x(t-\tau)) - \widehat{\phi}(x(t))\hat{h}_{1}(x(t)) - \widehat{\phi}(x(t))\hat{h}_{2}(x(t-\tau))]'R^{-1}(t)dv(t)$$
(5.45)

If we try to write stochastic differential equations for the unknown quantities on the right hand side of (5.45), we see that this cannot

be done for all the unknowns, since  $\phi(\mathbf{x}(t))h_2(\mathbf{x}(t-\tau))$  does not lie in the domain of  $\mathscr{A}_t$ . Of course, there are many physical problems (for example, radar problems with spread targets [11]) where the observations are of the form  $h(\mathbf{x}_t) = \int_{-\tau}^0 \psi(\theta) H[\mathbf{x}(t+\theta), \mathbf{x}(t)] d\theta$ .

Moreover, one can approximate point delays by distributed delays of the above form. This will allow us to write a stochastic differential equation for  $\widehat{\phi[\mathbf{x}(t)]\mathbf{h}(\mathbf{x}_t)}$ . However, we will then get the unknown  $\widehat{\mathscr{A}_t\phi[\mathbf{x}(t)]\mathbf{h}(\mathbf{x}_t)}$  in our equation for  $\widehat{\phi[\mathbf{x}(t)]\mathbf{h}(\mathbf{x}_t)}$ . If  $\psi(-\tau) \neq 0$ ,  $\mathscr{A}_t\phi[\mathbf{x}(t)]\mathbf{h}(\mathbf{x}_t)$  will contain a term with point delay (see Case 2 above), and we are faced with the same problem as before. In general, if the functionals involved are in the domain of  $\mathscr{A}_t^i$ ,  $i = 1, \ldots n$ , we can write n coupled stochastic differential equations involving the moment functionals, just as in the diffusion process case. It should be clear from the above discussion that this puts rather severe restrictions on the functionals involved. Thus, the study of approximations to the optimal nonlinear filter for delay system remains an important and open problem.

There is, however, one special case where the optimal filter can be completely specified even when there are point delays in the system. This is the linear case with Gaussian distributions and will be treated next. · ....

5.5 <u>Optimal Filtering of Linear Stochastic Delay Systems</u> We consider the system defined by

-122-

$$dx(t) = a(x_t, t)dt + dw(t)$$
$$x(\theta) = x_0(\theta) \qquad \theta \in [-\tau, 0]$$

(5.46)

(5.47)

$$dz(t) = c(x_t, t)dt + dv(t)$$

Here

$$\mathscr{C}_{X}[0,T] \rightarrow \mathbb{R}$$

$$c: \mathscr{C}x[0,T] \rightarrow \mathbb{R}$$

are continuous linear functionals given by

$$a(x_{t},t) = \int_{-\tau}^{0} d_{\theta} A(t,\theta) x(t+\theta)$$
$$c(x_{t},t) = \int_{-\tau}^{0} d_{\theta} C(t,\theta) x(t+\theta)$$

where  $A(t,\theta)$  and  $C(t,\theta)$  satisfy the same conditions as those imposed on such functions in section 5.2. Also, we take  $x_0$  to be a Gaussian process on  $[-\tau,0]$  with  $\sup_{\theta} E|x_0(\theta)|^2 < \infty$ . By a similar argument to the case without delays [56], it is readily seen that the conditional distributions of  $x(t+\theta)$ , for any  $\theta \in [-\tau,0]$ , given z(s),  $0 \le s \le t$ , is Gaussian. For greater clarity in the subsequent exposition, we shall write  $\hat{x}(t|t) = E\{x(t)|z^t\}$  and  $\hat{x}(t+\theta|t) = E\{x(t+\theta|z^t\}, \theta \in [-\tau,0]$ . Using (5.43), we immediately obtain the following stochastic differential equation for the conditional mean

$$d\hat{\mathbf{x}}(t|t) = \int_{-\tau}^{0} d_{\theta} A(t,\theta) \hat{\mathbf{x}}(t+\theta|t) dt$$

$$+ \left[\int_{-\tau}^{0} E^{t} (\mathbf{x}(t)\mathbf{x}'(t+\theta|t)) d_{\theta} C'(t,\theta) - \int_{-\tau}^{0} \hat{\mathbf{x}}(t|t)\mathbf{x}'(t+\theta|t) d_{\theta} C'(t,\theta) \right] R^{-1}(t) d\nu(t)$$
(5.48)

-123-

Here the innovations process v(t) is given by  $z(t) - \int_0^t \int_{-\tau}^0 d_{\theta} C(s,\theta) \hat{x}(s+\theta|s) ds$ 

and hence depends, in general, on the smoothed estimates  $\hat{x}(s+\theta|s)$ , - $\tau \leq \theta \leq 0$ ,  $0 \leq s \leq t$ . Define the "smoothed" conditional error covariance as

$$P(t,\theta,\xi) = E^{t} \{ [x(t+\theta) - \hat{x}(t+\theta|t)] [x(t+\xi) - \hat{x}(t+\xi|t)]' \}$$

Then (5.48) can be rewritten as

$$d\hat{\mathbf{x}}(t|t) = \int_{-\tau}^{0} d_{\theta} A(t,\theta) \hat{\mathbf{x}}(t+\theta|t) dt + \int_{-\tau}^{0} P(t,0,\theta) d_{\theta} C'(t,\theta) R^{-1}(t) dv(t)$$
(5.49)

To evaluate the unknown terms on the right side of (5.49), we use (5.44) to write the smoothed estimate as

$$\hat{\mathbf{x}}(t+\theta|t) = \hat{\mathbf{x}}(t+\theta|t+\theta) + \int_{t+\theta}^{t} \mathbf{E}^{\mathbf{s}} \{\mathbf{x}(t+\theta) [\mathbf{c'}(\mathbf{x}_{s},s) - \hat{\mathbf{c'}}(\mathbf{x}_{s},s)] \} \mathbf{R}^{-1}(s) dv(s)$$

$$= \hat{\mathbf{x}}(t+\theta|t+\theta) + \int_{t+\theta}^{t} \int_{-\tau}^{0} P(s,t+\theta-s,\xi) d_{\xi} C(s,\xi)' R^{-1}(s) dv(s)$$
(5.50)

for  $\theta \in [-\tau, 0]$ .

An inspection of (5.49) and (5.50) shows that the optimal linear filter is completely characterized by  $\hat{x}(t+\theta|t)$ ,  $-\tau \leq \theta \leq 0$ , and the "smoothed" error covariance function P(t, $\theta$ , $\xi$ ). It remains only to derive appropriate equations for P(t, $\theta$ , $\xi$ ). Since the derivations are quite lengthy, we will present them in Appendix B, and state the final result here as Theorem 5.5.1 The optimal filter for the system (5.46), (5.47) is characterized by the following sets of equations:

(i) Generation of the condition mean  $\hat{x}(t \mid t)$ :

$$d\hat{\mathbf{x}}(t|t) = \int_{-\tau}^{0} d_{\theta} A(t,\theta) \hat{\mathbf{x}}(t+\theta|t) dt + \int_{-\tau}^{0} P(t,0,\theta) d_{\theta} C'(t,\theta) R^{-1}(t) dv(t)$$
(5.51)

(ii) Generation of the smoothed estimate  $\hat{x}(t+\theta|t)$ ,  $-\tau \leq \theta \leq 0$ :

$$\hat{\mathbf{x}}(\mathbf{t}+\boldsymbol{\theta}|\mathbf{t}) = \hat{\mathbf{x}}(\mathbf{t}+\boldsymbol{\theta}|\mathbf{t}+\boldsymbol{\theta}) + \int_{\mathbf{t}+\boldsymbol{\theta}}^{\mathbf{t}} \int_{-\tau}^{0} P(\mathbf{s},\mathbf{t}+\boldsymbol{\theta}-\mathbf{s},\boldsymbol{\xi}) d_{\boldsymbol{\xi}} C(\mathbf{s},\boldsymbol{\xi})' \mathbf{R}^{-1}(\mathbf{s}) d\boldsymbol{\nu}(\mathbf{s})$$
(5.52)

(iii) Generation of the smoothed error covariance  $P(t,\theta,\xi)$ :

$$\frac{\mathrm{d}}{\mathrm{d}t} P(t,0,0) = \int_{-\tau}^{0} P(t,0,\theta) \mathrm{d}_{\theta} A'(t,\theta) + \int_{-\tau}^{0} \mathrm{d}_{\theta} A(t,\theta) P(t,\theta,0)$$
$$- \int_{-\tau}^{0} \int_{-\tau}^{0} P(t,0,\theta) \mathrm{d}_{\theta} C'(t,\theta) R^{-1}(t) \mathrm{d}_{\xi} C(t,\xi) P(t,\xi,0) + Q(t)$$
(5.53)

$$\sqrt{2} P_{\eta}(t,\theta,0) = \int_{-\tau}^{0} P(t,\theta,\xi) d_{\xi} A'(t,\xi)$$

$$- \int_{-\tau}^{0} \int_{-\tau}^{0} P(t,\theta,\xi) d_{\xi} C'(t,\xi) R^{-1}(t) d_{\alpha} C(t,\alpha) P(t,\alpha,0)$$
(5.54)

$$\sqrt{3} P_{\sigma}(t,\theta,\xi) = - \int_{-\tau}^{0} \int_{-\tau}^{0} P(t,\theta,\beta) d_{\beta}C'(t,\beta)R^{-1}(t) d_{\alpha}C(t,\alpha)P(t,\alpha,\xi)$$
(5.55)

where  $\eta$  is the unit vector in the (1,-1) direction,  $\sigma$  is the unit vector in the (1,-1,-1) direction, and  $P_{\eta}(t,\theta,0)$  and  $P_{\sigma}(t,\theta,\xi)$  are

the directional derivatives of  $P(t,\theta,0)$  and  $P(t,\theta,\xi)$  in the directions  $\eta$  and  $\sigma$  respectively. The initial conditions are given by

$$\hat{\mathbf{x}}(0|0) = \overline{\mathbf{x}}_{0}(0), \ \hat{\mathbf{x}}(\theta|0) = \overline{\mathbf{x}}_{0}(\theta) \quad \theta \in [-\tau, 0]$$

$$P(0, \theta, \xi) = \Sigma_{0}(\theta, \xi) = E\{[\mathbf{x}_{0}(\theta) - \overline{\mathbf{x}}_{0}(\theta)][\mathbf{x}_{0}(\xi) - \overline{\mathbf{x}}_{0}(\xi)]'\}$$
(5.56)

A few comments about Theorem 5.5.1 is in order. Notice the great similarity as well as the striking differences in the form of the solution to the filtering problem for linear delay systems with Gaussian distributions and the corresponding solution for linear systems without delays. In our case, we need to characterize not only the conditional mean  $\hat{x}(t|t)$  (as in the non-delay case), but also the smoothed estimate  $\hat{x}(t+\theta|t)$ , not only the estimation error covariance P(t,0,0), but also the smoothed error covariance function  $P(t,\theta,\xi)$ . The complexity of the solution is clearly considerably increased when there are delays in the system. It is hoped, however, that the results presented in this and the next chapter will provide a theoretical foundation for the study of implementable filters for delay systems.

Note that our development only shows that  $P(t,\theta,\xi)$  is continuous and has directional derivatives. Of course, if  $P(t,\theta,\xi)$  were actually differentiable in  $(t,\theta,\xi)$ , we could rewrite

$$\sqrt{2} P_{\eta}(t,\theta,0) = \left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta}\right) P(t,\theta,0)$$
$$\sqrt{3} P_{\sigma}(t,\theta,0) = \left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta} - \frac{\partial}{\partial \xi}\right) P(t,\theta,\xi)$$

and

We would then obtain partial differential equations for  $P(t, \theta, \xi)$ .

-126-

Theorem 5.5.1 completely solves the filtering problem for general linear stochastic delay systems. Notice that although functionals of the form  $\phi(\mathbf{x}_t) = \mathbf{x}(t+\theta)$ ,  $-\tau < \theta < 0$ , do not lie in the domain of the generator of  $\mathbf{x}_t$ , we are still able to calculate the directional derivatives of the covariances. This is a consequence of the linearity and Gaussian assumptions.

Let us make a few remarks on the relationship of our work to past investigations. The linear filtering problem for systems with only point delays was first considered by Kwakernaak [65], whose results are similar to those presented in this section. He restricted his attention to linear filters and followed the approach of Kalman and Bucy [66], and his derivations were formal. In particular, he used partial derivatives liberally when these were not defined. His "proofs", therefore, were far from satisfactory. Lindquist also considered the linear filtering problem in [56]. However, he only gave a complicated integral equation for the filter gain, and did not derive equations for the covariances. His results are thus incomplete and do not display the structure of the filter equations. In particular, they are not suitable for the filter stability investigations that we pursue in Chapter 6. Recently, Mitter and Vinter [67] have studied the linear filtering problem from the viewpoint of stochastic evolution equations. While they did take in account some of the special features of delay systems, they had to exclude point delays from their observation model, an unsatisfactory restriction from various points of view. Furthermore, their derivation of the optimal filtering equations proceeds via a

-127-

dual control problem. As such, their approach cannot be extended to the nonlinear situation. By contrast, our approach enables us to allow a very general model for the stochastic delay system under consideration, including distributed as well as point delays. Covariance equations have been derived rigorously and the structure of the optimal filter is clearly displayed. We feel, therefore, that our results on linear filtering are the most complete and satisfactory to date.

In the special case where  $x_0 \equiv 0$ ,  $a(x_t,t) = Ax(t) + Bx(t-t)$ ,  $c(x_t,t) = Cx(t), Q$ , R constant matrices, it can be shown by exploiting the connection between linear optimal filtering and optimal control with quadratic criterion that  $P(t,\theta,\xi)$  are in fact continuously differentiable in  $(t,\theta,\xi)$ . When we compare the solutions to the linear optimal control and optimal linear filtering problems in Chapter 6, it will be helpful to use the notation

$$P_{o}(t) = P(t,0,0)$$
 (5.57)

$$P_{1}(t,\theta) = P(t,\theta,0)$$
(5.58)

$$P_{2}(t,\theta,\xi) = P(t,\theta,\xi)$$
(5.59)

In this case the optimal filter is given by the equations

$$d\hat{\mathbf{x}}(t|t) = A\hat{\mathbf{x}}(t|t) + B\hat{\mathbf{x}}(t-\tau|t) + P_{o}(t)C'R^{-1}[dz(t) - C\hat{\mathbf{x}}(t|t)dt]$$
(5.60)

$$\hat{\mathbf{x}}(t-\tau|t) = \hat{\mathbf{x}}(t-\tau|t-\tau) + \int_{t-\tau}^{t} P_2(s,t-\tau-s,0)C'R^{-1}[dz(s)-C\hat{\mathbf{x}}(s|s)ds]$$
(5.61)

$$\hat{\mathbf{x}}(\boldsymbol{\theta} \mid \mathbf{0}) = \mathbf{0} \qquad -\tau \leq \boldsymbol{\theta} \leq \mathbf{0}$$

$$\frac{d}{dt} P_{o}(t) = AP_{o}(t) + P_{o}(t)A' - P_{o}(t)C'R^{-1}CP_{o}(t) + Q + BP_{1}(t, -\tau) + P_{1}'(t, -\tau)B'$$
(5.62)

$$\left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta}\right) P_{1}(t,\theta) = P_{1}(t,\theta) [A' - C'R^{-1}CP_{0}(t)] + P_{2}(t,\theta,-\tau)B' \qquad (5.63)$$

$$\left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta} - \frac{\partial}{\partial \xi}\right) P_2(t,\theta,\xi) = -P_1(t,\theta)C'R^{-1}CP'_1(t,\xi)$$
(5.64)

with 
$$P_{0}(0) = P_{1}(0,\theta) = P_{2}(0,\theta,\xi) = 0$$
  $-\tau \le \theta, \ \xi \le 0$   
 $P_{1}(t,0) = P_{0}(t)$   
 $P_{2}(t,\theta,0) = P_{1}(t,\theta)$   
 $P_{0}(t) = P_{0}'(t), \ P_{2}(t,\theta,\xi) = P_{2}'(t,\xi,\theta)$  (5.65)

We shall be using these equations extensively in the next chapter.

Notice that in this special case,  $\hat{x}(t|t)$  depends only on  $\hat{x}(s|s)$ ,  $t-\tau \leq s \leq t$ , and from (5.60) and (5.61), we can obtain an explicit delay equation for  $\hat{x}(t|t)$ . This is because we have no delays in the observations. If  $c(x_t,t)$  were of the form  $C_{o}x(t) + C_{1}x(t-\tau)$ , then we would have the following equations

$$d\hat{x}(t|t) = A\hat{x}(t|t) + B\hat{x}(t-\tau|t) + [P_{o}(t)C_{o}' + P_{1}'(t,-\tau)C_{1}']R^{-1} \cdot [dz(t) - C_{o}\hat{x}(t|t)dt - C_{1}\hat{x}(t-\tau|t)dt]$$
(5.66)

$$\hat{\mathbf{x}}(t-\tau|t) = \hat{\mathbf{x}}(t-\tau|t-\tau) + \int_{t-\tau}^{t} [P_2(s,t-\tau-s,0)C_0' + P_2(s,t-\tau-s,-\tau)C_1']R^{-1} \cdot [dz(s) - C_0\hat{\mathbf{x}}(s|s)ds - C_1\hat{\mathbf{x}}(s-\tau|s)ds]$$

$$[dz(s) - C_0\hat{\mathbf{x}}(s|s)ds - C_1\hat{\mathbf{x}}(s-\tau|s)ds]$$
(5.67)

If we substitute (5.67) into (5.66), we see that  $\hat{x}(t|t)$  depends not only on  $\hat{x}(s|s)$ ,  $t-\tau \leq s \leq t$ , but also on delayed smoothed estimates  $\hat{x}(s-\tau|s)$ ,  $t-\tau \leq s \leq t$ . The presence of the delayed smoothed estimates prevents us from obtaining a delay equation for  $\hat{x}(t|t)$ . The additional complications which delayed observations introduce will be discussed further in section 6.7.

It is worth pointing out that in the case where  $x_0 \equiv 0$ ,  $\hat{x}(t+\theta|t)$ ,  $-\tau \leq \theta \leq 0$ , can be expressed as a Wiener integral with respect to the observations z(s),  $0 \leq s \leq t$ , i.e.,

$$\hat{\mathbf{x}}(t+\theta|t) = \int_{0}^{t} \mathbf{K}(t,\theta,s) dz(s)$$
(5.66)

for some kernel K(t, $\theta$ ,s). To see this, note that x(t) is a Gaussian process. Hence the orthogonal projection of x(t+ $\theta$ ) onto the Hilbert space spanned by the observations z(s),  $0 \le s \le t$  is the same as the conditional mean. Since the orthogonal projections can be expressed in the form of the right hand side of (5.66), so can  $\hat{x}(t+\theta|t)$ . This fact will be used in Chapter 6.

So far, we have studied optimal filtering for stochastic delay systems on a finite time interval. To make the filtering theory of delay systems more complete, we shall study the asymptotic behavior and stability properties of the optimal linear filter in the next chapter.

#### CHAPTER 6

#### STABILITY OF LINEAR OPTIMAL FILTERS AND CONTROL SYSTEMS

### 6.1 Introduction

In the previous chapter, we derived optimal filter equations for linear and nonlinear stochastic delay differential systems. In this chapter, we study the stability of the linear optimal filter for delay systems. We shall concern ourselves primarily with stochastic systems with delays in the system dynamics, but no delays in the observations. Extensions to the general case, including delays in the observations, will also be discussed. In proving the stability results, we make essential use of the duality between optimal filtering of linear stochastic delay systems and optimal control of linear delay systems with quadratic cost. We shall therefore begin the development with a summary of the known results for optimal control of linear delay systems with quadratic cost, and give some new results on the asymptotic stability of the optimal closed-loop control system. This brings into focus the role played by the concepts of stabilizability and observability. Next, we give a duality theorem which establishes the connection between optimal linear filtering and optimal control of linear delay systems. This enables us to identify the gains of the optimal linear filter with the gains of a "dual" control system in an appropriate way. We can therefore exploit the known convergence of the control gains to infer the convergence of the filter gains. The stability of the optimal filter is then established by constructing a Lyapunov functional. Once the stability of the optimal linear filter has been

-131-

established, it is relatively straightforward to prove the asymptotic stability of the optimal linear stochastic control system, drawing upon known results on the separation theorem for stochastic delay systems [56], [68].

### 6.2 <u>Optimal Control of Linear Delay Systems with</u> <u>Quadratic Cost</u>

The problem of optimal control of linear delay systems with quadratic cost has received considerable attention in recent years. Various methods [69] - [73] have been devised for solving this problem. The simplest version of the problem can be formulated as follows:

We are given a linear constant delay system of the form

$$\frac{d\mathbf{x}(t)}{dt} = A\mathbf{x}(t) + B\mathbf{x}(t-\tau) + C\mathbf{u}(t)$$
$$\mathbf{x}(\theta) = \phi(\theta) \qquad \theta \in [-\tau, 0]$$
(6.1)

Our choice of function space will be  $\mathscr{C}$ , although basically the same results hold for other choices, such as  $M^2$  [29], [73]. The admissible control set U is the set of  $R^m$ -valued  $L_2$  functions on [0, T]. The cost functional is given by

$$J_{T}(u,\phi) = \int_{0}^{T} [x'(t)Qx(t) + u'(t)Ru(t)]dt$$

where Q and R are symmetric matrices of appropriate dimensions,  $Q \ge 0$ , R > 0. The objective is to find a control u  $\varepsilon$  U such that  $J_T(u, \phi)$  is minimized. The cases when T is finite or infinite have been studied. We shall briefly survey the known results and then give some extensions. We start with the case when  $T < \infty$ . Then it is well-known [69] -[73] that the optimal control can be expressed in feedback form by

$$u^{*}(t) = -R^{-1}C'K_{0}(t)x(t) - R^{-1}C'\int_{-\tau}^{0}K_{1}(t,\theta)x(t+\theta)d\theta \quad (6.2)$$

The feedback gains satisfy the following coupled set of partial differential equations

$$\frac{d}{dt} K_{o}(t) = -A'K_{o}(t) - K_{o}(t)A + K_{o}(t)CR^{-1}C'K_{o}(t) - Q - K_{1}(t,0) - K_{1}'(t,0)$$
(6.3)

$$\left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta}\right) K_{1}(t,\theta) = -[A' - K_{0}(t)CR^{-1}C']K_{1}(t,\theta) - K_{2}(t,0,\theta)$$
(6.4)

$$\left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta} - \frac{\partial}{\partial \xi}\right) K_2(t, \theta, \xi) = K_1'(t, \theta) CR^{-1} C' K_1(t, \xi)$$
(6.5)

The boundary conditions are

$$K_{0}(T) = K_{1}(T,\theta) = K_{2}(T,\theta,\xi) = 0 \qquad -\tau \leq \theta, \ \xi \leq 0$$

$$K_{1}(t,-\tau) = K_{0}(t)B$$

$$K_{2}(t,-\tau,\theta) = B'K_{1}(t,\theta)$$
(6.6)

Furthermore,

$$K_{o}(t) = K'_{o}(t), \quad K_{2}(t,\theta,\xi) = K'_{2}(t,\xi,\theta)$$
 (6.7)

The optimal cost can be expressed as

$$J_{T}^{\star}(\phi) = \phi'(0)K_{0}(0)\phi(0) + \int_{-\tau}^{0} \phi'(0)K_{1}(0,\theta)\phi(\theta)d\theta$$
$$+ \int_{-\tau}^{0} \phi'(\theta)K_{1}'(0,\theta)\phi(0)d\theta$$
$$+ \int_{-\tau}^{0} \int_{-\tau}^{0} \phi'(\theta)K_{2}(0,\theta,\xi)\phi(\xi)d\theta d\xi \qquad (6.8)$$

The integral term in Eq. (6.2) represents the effects of the delay on the optimal control. Thus the optimal control is seen to be given by a linear map K operating on the complete state  $x_t$ . Similarly, the optimal cost can be thought of as a generalized quadratic form.

We now discuss the infinite-time control problem, i.e.,  $T = \infty$ . For the control problem to be meaningful, we need some condition to ensure that the optimal cost will be finite. It turns out that the system-theoretic concepts of stabilizability and observability introduced in Chapter 2 play a crucial role. The following is the basic result on the convergence of the optimal control law and control gains for the infinite-time problem [32], [72].

<u>Proposition 6.2.1</u> Assume that system (6.1) is stabilizable. Then the gains  $K_0(t)$ ,  $K_1(t,\theta)$  and  $K_2(t,\theta,\xi)$ , for t < T, converge uniformly in t to  $K_0$ ,  $K_1(\theta)$ , and  $K_2(\theta,\xi)$  respectively as  $T \rightarrow \infty$ . The optimal control law for the infinite-time control problem with quadratic cost is given by

-134-

$$u^{*}(t) = -R^{-1}C^{*}K_{o}x(t) - \int_{-\tau}^{0} R^{-1}C^{*}K_{1}(\theta)x(t+\theta)d\theta \qquad (6.10)$$

where  ${\rm K}_{_{\rm O}},~{\rm K}_{1}^{}(\theta)$  and  ${\rm K}_{2}^{}(\theta,\xi)$  satisfy the following set of equations

$$A'K_{o} + K_{o}A - K_{o}CR^{-1}C'K_{o} + Q + K'_{1}(0) + K'_{1}(0) = 0$$
(6.11)

$$\frac{d}{d\theta} K_1(\theta) = [A' - K_0 CR^{-1}C']K_1(\theta) + K_2(0,\theta)$$
(6.12)

$$\left(\frac{\partial}{\partial\theta} + \frac{\partial}{\partial\xi}\right) K_2(\theta,\xi) = -K_1'(\theta) CR^{-1}C'K_1(\xi)$$
(6.13)

with boundary conditions

$$K_{1}(-\tau) = K_{0}B$$

$$K_{2}(\theta, -\tau) = K'_{1}(\theta)B$$
(6.14)

and symmetry conditions

$$K_{0} = K'_{0}, \qquad K_{2}(\theta,\xi) = K'_{2}(\xi,\theta)$$
 (6.15)

The optimal cost is given by

$$J_{\infty}^{*}(\phi) = \phi'(0)K_{0}\phi(0) + \int_{-\tau}^{0} \phi'(0)K_{1}(\theta)\phi(\theta)d\theta + \int_{-\tau}^{0} \int_{-\tau}^{0} \phi'(\theta)K_{2}(\theta,\xi)\phi(\xi)d\theta d\xi + \int_{-\tau}^{0} \int_{-\tau}^{0} \phi'(\theta)K_{2}(\theta,\xi)\phi(\xi)d\theta d\xi$$
(6.16)

Asymptotic stability of the closed-loop system have also been studied previously by Datko [72], and Delfour et al., [32]. In both papers, the assumption that  $Q \ge 0$  was made. Here we relax the assumption to observability. Let Q = H'H and let x(t) denote the optimal trajectory corresponding to the optimal control  $u^{*}(t)$ , i.e., x(t) satisfies

$$\dot{\mathbf{x}}(t) = (\mathbf{A} - \mathbf{CR}^{-1}\mathbf{C'K}_{0})\mathbf{x}(t) + \mathbf{B}\mathbf{x}(t-\tau) - \int_{-\tau}^{0} \mathbf{CR}^{-1}\mathbf{C'K}_{1}(\theta)\mathbf{x}(t+\theta)d\theta$$
(6.17)

We then have

<u>Theorem 6.2.1</u> Assume (A,B,C) is stabilizable and (A,B,H) is observable. Then the optimal control for the infinite time problem is given by (6.10)and the optimal closed-loop system (6.17) is asymptotically stable.

<u>Proof</u>: The first part of the theorem is just Proposition 6.2.1. To prove asymptotic stability of the closed-loop system, consider the Lyapunov functional

$$V(\mathbf{x}_{t}) = \mathbf{x}'(t)K_{0}\mathbf{x}(t) + \int_{-\tau}^{0} \mathbf{x}'(t)K_{1}(\theta)\mathbf{x}(t+\theta)d\theta$$
  
+ 
$$\int_{-\tau}^{0} \mathbf{x}'(t+\theta)K_{1}'(\theta)\mathbf{x}(t)d\theta + \int_{-\tau}^{0} \int_{-\tau}^{0} \mathbf{x}'(t+\theta)K_{2}(\theta,\xi)\mathbf{x}(t+\xi)d\theta d\xi$$
  
(6.18)

This, however, precisely corresponds to the optimal cost for the infinitetime problem starting at time t with initial function  $x_t$ . Thus we can write

$$V(x_t) = \int_t^\infty [x'(s)Qx(s) + u*'(s)Ru*(s)]ds$$

 $V(x_t)$  is clearly nonnegative. We claim that  $V(x_t) = 0$  implies x(t) = 0:

 $V(x_t) = 0$  if and only if

u(s) = 0  $s \ge t$ 

and x'(s)Qx(s) = x'(s)H'Hx(s) = 0  $s \ge t$ 

By observability of (A,B,H), this implies

$$x(s) = 0$$
  $s \ge t$ 

Let  $U_{\ell} = \{\phi: V(\phi) < \ell\}$ . Then for some  $\ell_1 > 0$ ,  $x_t \in U_{\ell_1}$  implies  $|x(t)| \leq K$  for some nonnegative constant K. Since  $V(x_t) \geq 0$  and  $\dot{V}(x_t) = -[x'(t)Qx(t) + u*'(t)Ru*(t)] \leq 0$  we can apply the invariance principle for functional differential equations [74] to conclude that the solutions of (6.17)  $x_t$  tends to M, the largest invariant set contained in the set

 $S = \{\phi: \dot{V}(\phi) = 0\}$ 

Since  $\dot{V}(x_{t}) = -[x'(t)Qx(t) + u*'(t)Ru*(t)]$ 

$$\dot{V}(x_t) = 0 \implies Hx(t) = 0$$

By observability, the largest invariant set M in S are those solutions of (6.17) for which  $|\mathbf{x}(t)| = 0$  for  $-\infty < t < \infty$ . Hence M = {0}, and the asymptotic stability of (6.17) is proved.

<u>Remark 6.2.1</u> It is known that asymptotic stability of autonomous linear functional differential equations is equivalent to exponential stability [2]. Hence the optimal closed-loop system for the infinitetime problem is globally asymptotically stable under the conditions of stabilizability and observability. <u>Corollary 6.2.1</u> Assume (A,B,C) is stabilizable and Q > 0. Then the optimal closed-loop system (6.17) is asymptotically stable. Since Q > 0 implies  $Q^{1/2} = H > 0$ , we immediately have that (A,B,H) is observable and we may invoke Theorem 6.2.1.

<u>Remark 6.2.2</u> Corollary 6.2.1 was proved earlier by Datko [72] and Delfour et al., [32] using somewhat different techniques.

<u>Corollary 6.2.2</u> If (A,B,C) is stabilizable and (A,B,H) is strongly observable, then (6.17) is asymptotically stable.

<u>Proof</u>: Follows from Theorem 6.2.1 since strong observability implies observability.

In proving asymptotic stability of the optimal filter, the same basic technique will be used: construction of a suitable Lyapunov functional and an appeal to the invariance principle. The systemtheoretic concepts of detectability and controllability are exploited there in a natural way. Before proceeding on that route, we shall first study the duality between optimal estimation and control for linear delay systems.

## 6.3 <u>Duality Between Estimation and Control for Linear</u> Delay Systems

In this section, we extend the duality principle between estimation and control for linear ordinary differential systems to linear delay differential systems. We shall exploit this duality in proving the convergence of the optimal filter gains in the next section. Similar results on duality are given by Lindquist [75], and Mitter and Vinter [67].

-138-

Consider the optimal filtering problem (in the minimum mean square error sense) for the stochastic delay system

$$dx(t) = [Ax(t) + Bx(t-\tau)]dt + dw(t)$$
  

$$x(\theta) = 0 \qquad \theta \le 0 \qquad (6.19)$$

$$dz(t) = [C_{0}x(t) + C_{1}x(t-\tau)]dt + dv(t)$$
(6.20)

where w(t) and v(t) are independent Wiener processes with  $cov[w(t); w(s)] = Qmin(t,s), cov[v(t); v(s)] = Rmin(t,s), Q \ge 0, and R > 0. We know from$ Chapter 5 that for any vector b, the optimal estimate for b'x(t) can be $expressed as <math>\int_{0}^{t} k'_{t}(s)dz(s)$  for some function  $k_{t}(s)$ . We define the "dual" control problem to the optimal filtering problem to be:

minimize 
$$J_{T}(b,u) = \int_{0}^{T} [y'(t)Qy(t)+u'(t)Ru(t)]dt$$
 (6.21)

subject to the constraint

$$\dot{y}(t) = -A'y(t) - B'y(t+\tau) - C_0 u(t) - C_1 u(t+\tau)$$
 (6.22)

with the boundary conditions

$$y(T) = b$$
  
 $y(s) = 0$   $s > T$   
 $u(s) = 0$   $s > T$ 

As in section 6.2, the admissible controls are  $L_2$  functions on [0, T].

The following duality theorem holds.

<u>Theorem 6.3.1</u> Consider the optimal estimation of the quantity b'x(T)in the system (6.19) - (6.20). Let the optimal estimate for b'x(T) be  $b'\hat{x}(T|T)$  and let the optimal control for the dual problem (6.21) - (6.22) be  $\mathbf{u}_{T}^{T}$ . Then  $b'\hat{\mathbf{x}}(T|T)$  is related to  $\mathbf{u}_{T}^{T}$  by

$$b'\hat{x}(T|T) = -\int_{0}^{T} u'_{T}(s)dz(s)$$
 (6.23)

<u>Proof</u>: Consider the function

$$\mathscr{H}(y,x,t) \stackrel{\Delta}{=} y'(t)x(t) + \int_{t-\tau}^{t} y'(s+\tau)Bx(s)ds$$

where y(t) and x(t) are solutions of (6.22) and (6.19) respectively. Applying Ito's differential rule, we see that

$$d\mathscr{H}(y,x,t) = -[y'(t)A+y'(t+\tau)B+u'(t)C_0+u'(t+\tau)C_1]x(t)dt + y'(t) {[Ax(t)+Bx(t-\tau)]dt+dw(t)} + y'(t+\tau)Bx(t)dt-y'(t)Bx(t-\tau)dt = -[u'(t)C_0x(t)+u'(t+\tau)C_1x(t)]dt+y'(t)dw(t) = -u'(t)[dz(t)-C_1x(t-\tau)dt-dv(t)]-u'(t+\tau)C_1x(t)dt+y'(t)dw(t)$$

In view of the boundary conditions for (6.19) and (6.22), we obtain

$$\begin{aligned} \mathscr{H}(y,x,T) &-\mathscr{H}(y,x,0) \\ &= y'(T)x(T) = b'x(T) \\ &= -\int_0^T u'(t)dz(t) + \int_0^T u'(t)dv(t) \\ &+ \int_0^T y'(t)dw(t) + \int_0^T u'(t)C_1x(t-\tau)dt - \int_0^T u'(t+\tau)C_1x(t)dt \\ &= -\int_0^T u'(t)dz(t) + \int_0^T u'(t)dv(t) + \int_0^T y'(t)dw(t) \end{aligned}$$

Hence

$$E\{b'x(T) + \int_{0}^{T} u'(t)dz(t)\}^{2} = \int_{0}^{T} [y'(t)Qy(t) + u'(t)Ru(t)]dt$$

Minimizing the mean square estimation error is thus the same problem as the dual optimal control problem, and the theorem follows.

<u>Remark 6.3.1</u> The duality theorem can be extended to more general delay systems such as those with multiple or distributed delays. However, as we shall not need such generality in subsequent developments, we shall not pursue them here. The restriction of  $x(\theta) = 0$ , for  $\theta \leq 0$ , is imposed because we want the dual control problem to be one whose solution is known explicitly. If we allow a zero-mean random initial function  $x_0$ , with  $E[x_0(\theta)x'_0(\xi)] = \sum_0(\theta,\xi), -\tau \leq \theta, \xi \leq 0$ , similar arguments show that the dual control problem is to minimize

$$J_{T}^{\prime}(b,u) = y^{\prime}(0)\Sigma_{o}(0,0)y(0) + \int_{-\tau}^{0} y^{\prime}(\theta+\tau)B\Sigma_{o}(\theta,0)d\theta y(0)$$
$$+ \int_{-\tau}^{0} y^{\prime}(0)\Sigma_{o}(0,\theta)B^{\prime}y(\theta+\tau)d\theta + \int_{-\tau}^{0} \int_{-\tau}^{0} y^{\prime}(\theta+\tau)B\Sigma_{o}(\theta,\xi)B^{\prime}y(\xi+\tau)d\theta d\xi$$
$$+ \int_{0}^{T} y^{\prime}(t)Qy(t)dt + \int_{0}^{T} u^{\prime}(t)Ru(t)dt$$

In order to use the duality theorem, we would have to first solve the control problem for the cost  $J_T'(b,u)$ . These are additional complications which do not require any new concepts. To avoid obscuring the main points of our development, we have opted to concentrate on the case  $x(\theta) = 0, \ \theta \leq 0.$ 

We will now restrict our attention to the case where  $C_1 = 0$ . Denoting  $C_0$  by C, the system now under consideration is

$$dx(t) = [Ax(t) + Bx(t-\tau)]dt + dw(t)$$

$$x(\theta) = 0 \qquad -\tau \le \theta \le 0$$

$$dz(t) = Cx(t)dt + dv(t) \qquad (6.25)$$

The case where  $C_1 \neq 0$  will be discussed at the end of this chapter since additional difficulties arise. We first recall from Chapter 5 the optimal filter equations for the system defined by (6.24) - (6.25).

$$d\hat{\mathbf{x}}(t|t) = [\mathbf{A} - \mathbf{P}_{0}(t)\mathbf{C'R}^{-1}\mathbf{C}]\hat{\mathbf{x}}(t|t) + B\hat{\mathbf{x}}(t-\tau|t-\tau)$$

$$-B \int_{-\tau}^{0} \mathbf{P}_{2}(t,-\tau,\theta)\mathbf{C'R}^{-1}\mathbf{C}\hat{\mathbf{x}}(t+\theta|t+\theta)d\theta + \mathbf{P}_{0}(t)\mathbf{C'R}^{-1}d\mathbf{z}(t)$$

$$+B \int_{t-\tau}^{t} \mathbf{P}_{2}(s,t-\tau-s,0)\mathbf{C'R}^{-1}d\mathbf{z}(s)dt \qquad (6.26)$$

$$\hat{\mathbf{x}}(\theta|0) = 0 \qquad -\tau \leq \theta \leq 0$$

By the variation of constants formula from Chapter 5, we can write the solution to (6.26) as

$$\hat{x}(T|T) = \int_{0}^{T} \Phi(T,s)P_{o}(s)C'R^{-1}dz(s) + \int_{0}^{T} \Phi(T,s)B \int_{s-T}^{s} P_{2}(u,s-\tau-u,0)C'R^{-1}dz(u)ds$$
(6.27)

where  $\Phi(t,s)$  is the fundamental matrix associated with (6.26), and satisfies the matrix differential equation

$$\frac{\partial \Phi(t,s)}{\partial t} = [A-P_{o}(t)C'R^{-1}C]\Phi(t,s) + B\Phi(t-\tau,s)$$

$$-B \int_{-\tau}^{0} P_{2}(t,-\tau,\theta)C'R^{-1}C\Phi(t+\theta,s)d\theta$$

$$\Phi(s,s) = I, \quad \Phi(t,s) = 0 \quad t < s$$
(6.28)

Applying a Fubini-type theorem for Lebesgue and Wiener integrals to (6.27) [50], we can write

$$\int_{0}^{T} \Phi(T,s) B \int_{s-\tau}^{s} P_{2}(u,s-\tau-u,0) C' R^{-1} dz(u) ds$$
$$= \int_{-\tau}^{T} \int_{u}^{u+\tau} \Phi(T,s) BP_{2}(u,s-\tau-u,0) C' R^{-1} ds dz(u)$$
$$= \int_{0}^{T} \int_{0}^{\tau} (T,u+\theta) BP_{1}(u,\theta-\tau) C' R^{-1} d\theta dz(u)$$

using the definitions of  $P_2(t,\theta,\xi)$  and  $P_1(t,\theta)$ , and the fact that z(u) = 0u < 0. Hence

$$\hat{\mathbf{x}}(\mathbf{T}|\mathbf{T}) = \int_{0}^{T} \{\Phi(\mathbf{T}, \mathbf{s}) P_{o}(\mathbf{s}) C' R^{-1} + \int_{0}^{T} \Phi(\mathbf{T}, \mathbf{s}+\theta) B P_{1}(\mathbf{s}, \theta-\tau) C' R^{-1} d\theta \} dz(\mathbf{s})$$
(6.29)

This gives an explicit representation of  $\hat{x}(T \, \big| \, T)$  in the form

$$\hat{\mathbf{x}}(\mathbf{T}|\mathbf{T}) = \int_{0}^{\mathbf{T}} K(\mathbf{T}, \mathbf{s}) d\mathbf{z}(\mathbf{s})$$

On the other hand, Theorem 6.3.1 gives a representation of  $b\,' \hat{x}\,(T\,|\,T)$  in the form

$$b'\hat{x}(T|T) = -\int_{0}^{T} u_{T}'(s)dz(s)$$
 (6.30)

We can therefore compare the representations in (6.29) and (6.30) and identify the control and filter gains appropriately. This will enable us to exploit the known convergence properties of the optimal control gains. We shall do this in the next section and also prove the asymptotic stability of the optimal linear filter.

## 6.4 <u>Stability of the Optimal Filter for Linear Systems with</u> Delays in the Dynamics

In this section, we shall put together the results developed in the previous sections and prove the asymptotic stability of the optimal filter. It is worth noting here that our proof of the convergence of the filter gains is rather indirect, being based on the convergence of the optimal control gains for the dual control problem. However, it is not obvious how a direct argument can be used. Indeed, variational arguments for proving convergence of the control gains do not as yet have a probabilistic interpretation in the filtering context for delay systems. At the present state of our knowledge, it appears that the indirect approach has to be adopted.

We begin now our investigations on the stability of the optimal filter. We first adapt the solution of the optimal control problem for linear delay systems with quadratic cost to solve the dual control problem posed in Eq. (6.21) - (6.22) in the previous section. Throughout this section, we will only consider the case where  $C_1 = 0$ ,  $C_0 = C$ .
We make the change of variables s = T-t (s will be defined this way until further notice). Then Eq. (6.22) becomes

$$-\frac{d}{ds} y(T-s) = -A'y(T-s)-B'y(T-s+\tau)-C'u(T-s)$$
(6.31)

Define

$$y(s) = y(T-s) = y(t)$$

$$u(s) = u(T-s) = u(t)$$

(6.31) becomes

$$\frac{d}{ds} \overline{y}(s) = A'\overline{y}(s) + B'\overline{y}(s-\tau) + C'\overline{u}(s)$$

$$\overline{y}(0) = b, \quad \overline{y}(\theta) = 0, \quad -\tau \le \theta < 0$$
(6.32)

(Although the initial function here is not continuous, the results in section 6.2 can be readily extended to cover this case as well [35].) The cost functional  $J_{T}(b,u)$  can be written as

$$J_{T}(b,u) = \int_{0}^{T} [\overline{y}(s)'Q\overline{y}(s) + \overline{u}'(s)R\overline{u}(s)]ds \qquad (6.33)$$

The dual control problem has now been cast into the standard form, and the results of section 6.2 can be applied. We can therefore state that the optimal control  $\overline{u^*}(s)$  is given by

$$\overline{u}^{*}(s) = -R^{-1}C\overline{K}_{o}(s)\overline{y}(s) - \int_{-\tau}^{0} R^{-1}C\overline{K}_{o}(s,\theta)\overline{y}(s+\theta)d\theta \qquad (6.34)$$

where  $\overline{\overline{K}}_{0}(s) = -\overline{\overline{K}}_{0}(s)A' - A\overline{\overline{K}}_{0}(s) + \overline{\overline{K}}_{0}(s)C'R^{-1}C\overline{\overline{K}}_{0}(s) - Q-\overline{\overline{K}}_{1}(s,0) - \overline{\overline{K}}_{1}(s,0)$  (6.35)

$$\left(\frac{\partial}{\partial s} - \frac{\partial}{\partial \theta}\right) \overline{K}_{1}(s,\theta) = -[A - \overline{K}_{0}(s)C'R^{-1}C]\overline{K}_{1}(s,\theta) - \overline{K}_{2}(s,0,\theta)$$
(6.36)

$$\left(\frac{\partial}{\partial s} - \frac{\partial}{\partial \theta} - \frac{\partial}{\partial \xi}\right) \overline{K}_{2}(s, \theta, \xi) = \overline{K}_{1}'(s, \theta) C' R^{-1} C \overline{K}_{1}(s, \xi)$$
(6.37)

with boundary conditions

$$\overline{K}_{0}(T) = \overline{K}_{1}(T,\theta) = K_{2}(T,\theta,\xi) = 0 , \quad -\tau \leq \theta, \xi \leq 0$$

$$\overline{K}_{1}(s,-\tau) = \overline{K}_{0}(s)B' \qquad (6.38)$$

$$\overline{K}_{2}(s,-\tau,\theta) = B\overline{K}_{1}(s,\theta)$$

and symmetry conditions

$$\overline{K}_{o}(s) = \overline{K}'_{o}(s), \ \overline{K}_{2}(s,\theta,\xi) = \overline{K}'_{2}(s,\xi,\theta)$$
(6.39)

The closed-loop optimal system is given by

$$\frac{d\overline{y}(s)}{ds} = [A' - C'R^{-1}C\overline{K}_{0}(s)]\overline{y}(s) + B'\overline{y}(s-\tau) - \int_{-\tau}^{0} C'R^{-1}C\overline{K}_{1}(s,\theta)\overline{y}(s+\theta)d\theta$$
(6.40)

Using the variation of constants formula and the fact that  $\overline{y}(s) = 0$ - $\tau \leq s < 0$ , we get that

$$\overline{\mathbf{y}}(\mathbf{s}) = \overline{\Phi}(\mathbf{s}, \mathbf{0})\mathbf{b} \tag{6.41}$$

where  $\overline{\Phi}(s,r)$  is the fundamental matrix associated with (6.40). Now define

$$K_{o}(t) = \overline{K}_{o}(s) = \overline{K}_{o}(T-t)$$
$$K_{1}(t,\theta) = \overline{K}_{1}(s,-\theta) = \overline{K}_{1}(T-t,-\theta)$$

$$K_2(t,\theta,\xi) = \overline{K}_2(s,-\theta,-\xi) = \overline{K}_2(T-t,-\theta,-\xi)$$

The following lemma is now immediate and summarizes the solution of the dual control problem.

Lemma 6.4.1 The optimal control for the problem (6.21) - (6.22) is given by

$$u^{*}(t) = -R^{-1}CK_{0}(t)y(t) - \int_{-\tau}^{0} R^{-1}CK_{1}(t, -\theta)y(t-\theta)d\theta \qquad (6.42)$$

where

$$\dot{K}_{o}(t) = AK_{o}(t) + K_{o}(t)A' - K_{o}(t)C'R^{-1}CK_{o}(t) + Q + K_{1}(t,0) + K_{1}'(t,0)$$
(6.43)

$$\left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta}\right) K_{1}(t,\theta) = [A - K_{0}(t)C'R^{-1}C]K_{1}(t,\theta) + K_{2}(t,0,\theta) \quad (6.44)$$

$$\left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta} - \frac{\partial}{\partial \xi}\right) K_2(t,\theta,\xi) = -K_1'(t,\theta)C'R^{-1}CK_1(t,\xi)$$
(6.45)

with

$$K_{0}(0) = K_{1}(0,\theta) = K_{2}(0,\theta,\xi) = 0$$
  $0 \le \theta, \xi \le \tau$ 

$$K_{1}(t,\tau) = K_{0}(t)B'$$

$$K_{2}(t,\tau,\xi) = BK_{1}(t,\xi)$$

$$K_{0}(t) = K_{0}'(t), K_{2}(t,\theta,\xi) = K_{2}'(t,\xi,\theta)$$
(6.46)

and

Furthermore, the optimal closed-loop system satisfies

$$\frac{d}{dt} y(t) = -[A'-C'R^{-1}CK_{0}(t)]y(t) - B'y(t+\tau)$$

$$+ \int_{0}^{\tau} C'R^{-1}CK_{1}(t,\theta)y(t+\theta)d\theta$$

$$y(T) = b$$
  
 $y(t) = 0$   $t > T$  (6.47)

Let the fundamental matrix associated with (6.47) be Y(t,r). We then have the following:

Lemma 6.4.2 Y(t,r) is related to the matrix  $\overline{\Phi}(s,r)$  introduced in (6.41) by

$$Y(t,T-r) = \overline{\Phi}(s,r) \tag{6.48}$$

<u>Proof</u>: By straightforward verification, the above two matrices satisfy the same delay equation as well as boundary conditions. Uniqueness then yields (6.48).

Combining Lemmas 6.4.1 and 6.4.2 and using (6.41), we obtain an alternative formula for the optimal control

$$u^{\star}(t) = -[R^{-1}CK_{0}(t)Y(t,T) + \int_{-\tau}^{0} R^{-1}CK_{1}(t,-\theta)Y(t-\theta,T)d\theta]b$$
$$= -[R^{-1}CK_{0}(t)Y(t,T) + \int_{0}^{\tau} R^{-1}CK_{1}(t,\theta)Y(t+\theta,T)d\theta]b$$
(6.49)

Substituting (6.49) into (6.30) gives

.

$$b'\hat{x}(T|T) = b'\left\{\int_{0}^{T} Y'(t,T)K_{0}(t)C'R^{-1}dz(t) + \int_{0}^{T} \int_{0}^{T} Y(t+\theta,T)'K_{1}'(t,\theta)C'R^{-1}d\theta dz(t)\right\}$$

$$-148-$$

Since b is arbitrary, we obtain

Lemma 6.4.3 The optimal estimate  $\hat{\mathbf{x}}(T|T)$  for the stochastic delay system (6.24) - (6.25) can be written as

$$\hat{x}(T|T) = \int_{0}^{T} \{Y'(t,T)K_{0}(t)C'R^{-1} + \int_{0}^{T} Y(t+\theta,T)'K_{1}(t,\theta)C'R^{-1}d\theta\}dz(t)$$
(6.50)

We are now in a position to compare (6.50) and (6.29). First note that they are of exactly the same form, with  $\Phi$ ,  $P_0$ ,  $BP_1$  occurring at the same positions as Y, K<sub>0</sub>, and K<sub>1</sub>. One suspects therefore, that

$$\Phi(T,t) = Y(t,T)'$$

$$P_{o}(t) = K_{o}(t)$$
and
$$BP_{1}(t,\theta-\tau) = K_{1}'(t,\theta)$$

We proceed to confirm this. Define

$$\tilde{P}_{2}(t,\theta,\xi) = BP_{2}(t,\theta-\tau,\xi-\tau)B$$

$$\tilde{P}_{1}(t,\theta) = P_{1}'(t,\theta-\tau)B'$$

$$\tilde{P}_{0}(t) = P_{0}(t)$$

and recall that  $P_0(t)$ ,  $P_1(t,\theta)$ , and  $P_2(t,\theta,\xi)$  satisfy (5.62) - (5.65). Then it is straightforward to verify that  $\tilde{P}_0(t)$ ,  $\tilde{P}_1(t,\theta)$ , and  $\tilde{P}_2(t,\theta,\xi)$  satisfy exactly the same equations and boundary conditions as  $K_0(t)$ ,  $K_1(t,\theta)$ , and  $K_2(t,\theta,\xi)$ . By uniqueness of the optimal control and optimal filter, we may identify

۲

$$P_{o}(t) = K_{o}(t)$$

$$P'_{1}(t, \theta - \tau)B' = K_{1}(t, \theta)$$

$$BP_{2}(t, \theta - \tau, \xi - \tau)B' = K_{2}(t, \theta, \xi)$$
(6.54)

$$0 \le t \le T$$
,  $0 \le \theta$ ,  $\xi \le \tau$ 

It remains to prove that  $Y(t,T)' = \Phi(T,t)$ . Using (6.54), we see that

$$K_{1}(t,-\theta) = P_{1}'(t,-\theta-\tau)B'$$
  
=  $P_{2}(t-\theta,-\tau,\theta)'B'$  (6.55)

(6.54), (6.55) and (6.47) imply that Y(t,r) is the fundamental matrix of

$$\frac{d}{dt}y(t) = -[A'-C'R^{-1}CP_{0}(t)]y(t)-B'y(t+\tau) + \int_{-\tau}^{0} C'R^{-1}CP_{2}(t-\theta,-\tau,\theta)'B'y(t-\theta)d\theta \qquad (6.56)$$

But (6.56) is precisely the adjoint equation to

$$\frac{d}{dt} \mathbf{x}(t) = [A-P_{0}(t)C'R^{-1}C]\mathbf{x}(t) + B\mathbf{x}(t-\tau) - \int_{-\tau}^{0} BP_{2}(t,-\tau,\theta)C'R^{-1}C\mathbf{x}(t+\theta)d\theta$$

and it is known [2], [57] that

$$Y(t,r) = \Phi(r,t)'$$

We summarize this development in

<u>Lemma 6.4.4</u> The optimal filter gains for (6.19) - (6.20) are related to the optimal control gains for the dual problem (6.21) - (6.22) by

$$P_{o}(t) = K_{o}(t)$$

$$P'_{1}(t, \theta - \tau)B' = K_{1}(t, \theta)$$

$$BP_{2}(t, \theta - \tau, \xi - \tau)B' = K_{2}(t, \theta, \xi)$$

$$0 \le t \le T , \quad 0 \le \theta, \xi \le \tau$$

The fundamental matrix  $\Phi(t,r)$  associated with the optimal filter (6.26) is related to that of the optimal control system (6.47) by

$$\Phi(t,r) = Y(r,t)'$$

We are now ready to prove asymptotic stability of the optimal filter for (6.19) - (6.20). Let Q = H'H.

<u>Theorem 6.4.1</u> Suppose (A,B,C) is detectable and (A,B,H') is controllable. Then the gains of the optimal filter converge, and the steady state optimal filter is asymptotically stable.

<u>Proof</u>: From the results of Chapter 2, we know that our hypothesis implies that (A',B',C') is stabilizable and (A',B',H) is observable. By stabilizability, Proposition 6.2.1 shows that the optimal gains for the dual control problem  $\overline{K}_{0}(s)$ ,  $\overline{K}_{1}(s,\theta)$  and  $\overline{K}_{2}(s,\theta,\xi)$  converge to matrices  $\overline{K}_{0}$ ,  $\overline{K}_{1}(\theta)$  and  $\overline{K}_{2}(\theta,\xi)$  respectively as  $T \rightarrow \infty$ . The definitions of  $K_{0}(t)$ ,  $K_{1}(t,\theta)$  and  $K_{2}(t,\theta,\xi)$  now imply that these converge to  $K_{0}$ ,  $K_{1}(\theta)$ ,  $K_{2}(\theta,\xi)$  respectively as  $t \rightarrow \infty$ . By Lemma 6.4.4, we can conclude that

$$P_{o}(t) \rightarrow P_{o}$$

$$BP_{1}(t,\theta) \rightarrow BP_{1}(\theta) \quad \text{as } t \rightarrow \infty$$

$$BP_{2}(t,\theta,\xi)B' \rightarrow BP_{2}(\theta,\xi)B'$$

where (see also section 6.2)

-151-

$$AP_{o} + P_{o}A' - P_{o}C'R^{-1}CP_{o} + Q + BP_{1}(-\tau) + P_{1}'(-\tau)B' = 0$$
(6.57)

$$\frac{d}{d\theta} BP_{1}(\theta) = -BP_{1}(\theta) [A' - C'R^{-1}CP_{0}] - BP_{2}(\theta, -\tau)B'$$
(6.58)

$$\left(\frac{\partial}{\partial\theta} + \frac{\partial}{\partial\xi}\right) BP_2(\theta,\xi)B' = BP_1(\theta)C'R^{-1}CP_1'(\xi)B'$$
(6.59)

with

$$P_{1}(0) = P_{0}$$

$$P_{2}(0,0) = P_{1}(0)$$

$$P_{0} = P'_{0}, P_{2}(0,\xi) = P'_{2}(\xi,0)$$
(6.60)

Stability of the steady state filter is then governed by the stability of the equation

$$\frac{d}{dt} x(t) = [A - P_o C' R^{-1} C) x(t) + B x(t - \tau)$$

$$-B \int_{-\tau}^{0} P_2(-\tau, \theta) C' R^{-1} C x(t + \theta) d\theta \qquad (6.61)$$

Using Lemma 6.4.4 and (6.47), we see that the adjoint to the undriven steady-state filter equation (6.61) is given by

$$\frac{d}{dt} y(t) = -[A'-C'R^{-1}CP_{0}]y(t) - B'y(t+\tau)$$

$$+ \int_{0}^{T} C'R^{-1}CP'_{1}(\theta-\tau)B'y(t+\theta)d\theta \qquad (6.62)$$

Let s = -t,  $\overline{y}(s) = y(-s) = y(t)$ . Then

$$\frac{d}{ds} \overline{y}(s) = (A-C'R^{-1}CP_{0})\overline{y}(s) + B'\overline{y}(s-\tau) - \int_{0}^{\tau} C'R^{-1}CP_{1}'(\theta-\tau)B'\overline{y}(s-\theta)d\theta$$
(6.63)

-152-

We prove that (6.63) is asymptotically stable. Introduce the Lyapunov functional

$$\nabla(\overline{y}_{s}) = \overline{y}'(s)P_{0}\overline{y}(s) + \overline{y}'(s) \int_{0}^{t} P_{1}'(\theta - \tau)B'\overline{y}(s - \theta)d\theta$$
$$+ \int_{0}^{\tau} \overline{y}'(s - \theta)BP_{1}(\theta - \tau)\overline{y}(s)d\theta + \int_{0}^{\tau} \int_{0}^{\tau} \overline{y}'(s - \theta)BP_{2}(\theta - \tau, \xi - \tau)B'$$
$$\cdot \overline{y}(s - \xi)d\theta d \qquad (6.64)$$

-

Some standard computations show that

$$\dot{\mathbf{V}}(\mathbf{\bar{y}}_{s}) = -\left\{ \overline{\mathbf{y}'}(s) Q \overline{\mathbf{y}}(s) + [R^{-1} C P_{o} \overline{\mathbf{y}}(s) + \int_{0}^{T} R^{-1} C P_{1}'(\theta - \tau) B' \overline{\mathbf{y}}(s - \theta) d\theta ]' R \right.$$
$$\left. \cdot [R^{-1} C P_{o} \overline{\mathbf{y}}(s) + \int_{0}^{T} R^{-1} C P_{1}'(\theta - \tau) B' \overline{\mathbf{y}}(s - \theta) d\theta ]\right\}$$
(6.65)

Comparing (6.64) and (6.65) with the situation in Theorem 6.2.1, we see that they are completely analogous. By the same arguments and the observability of (A',B',H), we conclude that (6.63) is asymptotically stable. Let the fundamental matrices associated with (6.61), (6.62) and (6.63) be  $\Phi_0(t,r)$ ,  $Y_0(t,r)$  and  $\overline{Y}_0(t,r)$  respectively. Then it is easy to see that  $Y_0(t,r) = \overline{Y}_0(-t,-r)$ . It is known [2] that asymptotic stability of (6.63) is equivalent to the existence of constants  $\alpha > 0$ ,  $M \ge 1$  such that

$$\left|\left|\overline{Y}_{o}(s,r)\right|\right| \leq Me^{-\alpha(s-r)}$$

Thus  $||Y_{o}(t,r)|| = ||\overline{Y}_{o}(-t,-r)|| \leq Me^{-\alpha(r-t)}$ 

Since  $\Phi_{o}(t,r) = Y_{o}'(r,t), ||\Phi_{o}(t,r)|| \leq Me^{-\alpha(t-r)}$ 

The proof of the theorem is now complete.

<u>Remark 6.4.1</u> It should be pointed out that the convergence of the filter gains  $P_0(t)$ ,  $BP_1(t,\theta)$  and  $BP_2(t,\theta,\xi)B'$  do not provide as yet the convergence of the "smoothed" error covariance  $P_2(t,\theta,\xi)$ . Of course, if B is nonsingular,  $BP_2(t,\theta,\xi)B'$  converging will imply that  $P_2(t,\theta,\xi)$ . It is interesting to note that the nonsingularity of B seems to be related to many questions in the theory of delay equations. For example, it is related to the solution semigroup T(t) being one-to-one [2], the completeness of exponential solutions [76], and the convergence of projection series [77].

Instead of identifying  $P_o(t)$ ,  $BP_1(t,\theta)$ , and  $BP_2(t,\theta,\xi)B'$  with the corresponding control gains as in Lemma 6.4.4, one can directly identify  $P_o(t)$ ,  $P_1(t,\theta)$ , and  $P_2(t,\theta,\xi)$  with the Fredholm kernel  $P_c(t,\theta,\xi)$  introduced by Manitius in the optimal control with quadratic cost [71]. The details are straightforward and are left as an amusement for the reader.

### 6.5 Convergence of Estimation Error Covariance Operator

The results of the last section show that under suitable detectability and controllability conditions, the estimation error covariance  $P_0(t)$  converges. Our objective in this section is to prove the convergence of the error covariance operator for the steady state optimal filter. If we express the estimation error  $e(t|t) = x(t) - \hat{x}(t|t)$  in terms of a stochastic delay equation, its solutions define a trajectory in the space of continuous functions. By the error covariance operator

-154-

we mean the function

$$\Sigma(t,\theta,\xi) = E\{e(t+\theta | t+\theta)e'(t+\xi | t+\xi)\}, \quad -\tau \leq \theta, \xi \leq 0.$$

Of course  $\Sigma(t,0,0) = P_0(t)$ . However,  $\Sigma(t,\theta,0) \neq P_1(t,\theta)$ , and  $\Sigma(t,\theta,\xi) \neq P_2(t,\theta,\xi)$ . This is because  $\Sigma(t,\theta,0)$  and  $\Sigma(t,\theta,\xi)$  are covariances of the filtering estimation error, whereas  $P_1(t,\theta)$  and  $P_2(t,\theta,\xi)$  are covariances of the smoothing estimation error. While we can show that  $\Sigma(t,\theta,\xi)$  converges to a steady-state operator  $\Sigma_{\alpha}(\theta,\xi)$ , we have not been able to show the convergence of  $P(t,\theta,\xi)$ , unless B is invertible (see Remark 6.4.1).

It is easy to see that e(t|t) satisfies the stochastic differential equation

$$de(t|t) = [(A-P_{o}C'R^{-1}C)e(t|t) + Be(t-\tau|t-\tau)]dt$$

$$+ \int_{-\tau}^{0} BP_{2}(-\tau,\theta)C'R^{-1}Ce(t+\theta|t+\theta)d\theta dt + dw(t)$$

$$- P_{o}C'R^{-1}dv(t) - B \int_{-\tau}^{0} P_{2}(-\tau,\theta)C'R^{-1}d_{\theta}v(t+\theta)dt$$
(6.66)

Note that for any fixed t,  $v(t+\theta) \ \theta \in [-\tau,0]$  is a Brownian motion, and hence the last stochastic integral is well-defined. Again, let  $\Phi_0(t,s)$ be the fundamental matrix associated with (6.66). Since (6.66) is an autonomous equation  $\Phi_0(t,s) = \Phi_0(t-s,0) \stackrel{\triangle}{=} \Phi_1(t-s)$ . Then the variation of constants formula gives

$$e(t|t) = \int_{0}^{t} \Phi_{1}(t-s) dw(s) - \int_{0}^{t} \Phi_{1}(t-s) P_{0}C'R^{-1}dv(s)$$
$$- \int_{0}^{t} \Phi_{1}(t-s)B \int_{-\tau}^{0} P_{2}(-\tau,\theta)C'R^{-1}d_{\theta}v(s+\theta)ds$$
(6.67)

Using a Fubini-type theorem in Doob [50], the last term in (6.67) can be written as

$$\int_{0}^{t} \Phi_{1}(t-s) \int_{s-\tau}^{s} P_{2}(-\tau, \sigma-s) dv(\sigma) ds$$
$$= \int_{0}^{t} \int_{\sigma}^{\min(\sigma+\tau, t)} \Phi_{1}(t-s) P_{2}(-\tau, \sigma-s) ds dv(\sigma)$$

$$= \int_{0}^{t} \int_{0}^{\min(t-\sigma,\tau)} \Phi_{1}(t-\sigma-\beta)P_{2}(-\tau,-\beta)d\beta dv(\sigma) \qquad (6.68)$$

where we have used v(t) = 0,  $t \leq 0$ . For convenience, we define

$$L(t-s) = \Phi_1(t-s)P_0C'R^{-1} + \int_0^{\min(t-s,\tau)} \Phi_1(t-s-\beta)P_2(-\tau,-\beta)d\beta$$

If we now compute the characteristic functional of the  $\mathscr{C}$ -valued random variable  $e_{t|t}(e_{t|t}(\theta) = e(t+\theta|t+\theta))$ , then by exactly the same arguments as in Chapter 5, we get, for  $y \in \mathscr{C}^*$ ,

φ

$$(y,e_{t}|t) = E \exp \{ i \int_{-\tau}^{0} dy'(r) \left[ \int_{0}^{t} \Phi_{1}(t+r-s) dw(s) - \int_{0}^{t} L(t+r-s) dv(s) \right] \}$$

using  $\Phi_1(t-s) = 0$ , t < s.

Since for each fixed t, the exponent is a sum of Gaussian random variables, and w(t) and v(t) are completely independent, we can evaluate the expectation as

$$\exp - \frac{1}{2} \left\{ \int_{-\tau}^{0} \int_{-\tau}^{0} dy'(r) \int_{0}^{t} \Phi_{1}(t+r-s)Q\Phi_{1}'(t+\sigma-s)dsdy(\sigma) + \int_{-\tau}^{0} \int_{-\tau}^{0} dy'(r) \int_{0}^{t} L(t+r-s)RL'(t+\sigma-s)dsdy(\sigma) \right\}$$
(6.69)

Therefore, the covariance operator is given by

$$\Sigma(t,\theta,\xi) = \int_0^t \Phi_1(t+\theta-s)Q\Phi_1'(t+\xi-s)ds$$
  
+ 
$$\int_0^t L(t+\theta-s)RL'(t+\xi-s)ds$$
  
= 
$$\int_0^t [\Phi_1(\theta+s)Q\Phi_1'(\xi+s) + L(\theta+s)RL'(\xi+s)]ds \qquad (6.70)$$

The convergence of the covariance operator  $\Sigma(t,\theta,\xi)$  is given by

<u>Theorem 6.5.1</u> The covariance operator  $\Sigma(t,\theta,\xi)$  for the optimal estimation error converges pointwise in  $\theta$  and  $\xi$  to a stationary covariance operator  $\Sigma_{\infty}(\theta,\xi)$ , with

$$\begin{split} \Sigma_{\infty}(\theta,\xi) &= \int_{0}^{\infty} \Phi_{1}(\theta+s) Q \Phi_{1}^{\prime}(\xi+s) ds \\ &+ \int_{0}^{\infty} [\Phi_{1}(\theta+s) P_{0}C^{\prime}R^{-1} + \int_{0}^{\min(\theta+s,\tau)} \Phi_{1}(\theta+s-\beta) P_{2}(-\tau,-\beta) d\beta]^{\prime} \\ &\cdot R[\Phi_{1}(\xi+s) P_{0}C^{\prime}R^{-1} \\ &+ \int_{0}^{\min(\xi+s,\tau)} \Phi_{1}(\xi+s-\beta) P_{2}(-\tau,-\beta) d\beta] ds \quad (6.71) \end{split}$$

<u>Proof</u>: From (6.70), it is clear that  $\Sigma(t,\theta,\xi)$  is a symmetric operator, monotonically increasing in t. (See Remark 6.5.1 following the proof.) Since the optimal filter is asymptotically stable,  $||\phi_1(t-s)|| \leq Me^{-\alpha(t-s)}$ for some  $\alpha > 0$ ,  $M \geq 1$ . We can therefore make the estimate

$$\begin{aligned} &|| \int_{0}^{t} \Phi_{1}(\theta+s)Q\Phi_{1}(\xi+s)ds || \\ &\leq \int_{0}^{t} ||\Phi_{1}(\theta+s)Q\Phi_{1}(\xi+s)||ds \\ &\leq ||Q|| \int_{0}^{t} M^{2}e^{-\alpha(\theta+s)} e^{-\alpha(\xi+s)}ds \\ &= ||Q||M^{2}e^{-\alpha(\theta+\xi)} \left(1 - \frac{e^{-2\alpha t}}{2\alpha}\right) \end{aligned}$$

which converges as  $t \rightarrow \infty$ . By a similar argument, the second term in (6.70) is also uniformly bounded in t for each fixed  $\theta$  and  $\xi$ . We can now apply a theorem on the convergence of monotone symmetric operators

[78] to conclude that

$$\lim_{t \to \infty} \Sigma(t, \theta, \xi) = \Sigma_{\infty}(\theta, \xi)$$

exists pointwise in  $\theta$  and  $\xi$ , and is given by (6.71).

<u>Remark 6.5.1</u> The filtering error covariance operator  $\Sigma(t,\theta,\xi)$  is monotonically increasing in t because we have zero initial estimation error (i.e., the initial condition for the system is deterministic and known, taken to be zero in our formulation). If we have a random initial condition, we may still be able to prove (although we have not done so) that the resulting  $\Sigma(t,\theta,\xi)$  converges to  $\Sigma_{\infty}(\theta,\xi)$ , since initial conditions are usually "forgotten" as  $t \to \infty$ . We may note that analogous situations also arise in the filtering problem for systems without delays.

# 6.6 <u>Stochastic Control of Linear Delay Systems</u> and Stability

The preceding development in the stability of the optimal closedloop control system and the stability of the optimal linear filter suggests that by putting the two together, asymptotic stability of the closed-loop stochastic control system can be obtained. This is of course the case in linear stochastic system without delays. In this section, we examine the corresponding stochastic control system for delay systems and prove a similar result.

We shall first discuss the separation theorem for delay systems as proved by Lindquist [56]. Again we consider the following linear

-159-

stochastic delay system

$$dx(t) = [Ax(t) + Bx(t-\tau) + Gu(t)]dt + M'dw_{1}(t)$$

$$x(\theta) = \phi(\theta) \qquad \theta \varepsilon [-\tau, 0] \qquad (6.72)$$

$$dz(t) = Cx(t)dt + Ndw_{2}(t)$$
 (6.73)

 $w_1$ ,  $w_2$  are standard Wiener processes, completely independent of each other, and independent of the initial function  $\phi$ , which is a Gaussian process. The matrix N is assumed to be positive definite. The objective is to choose a control law u in a suitable set of admissible controls such that the cost functional

$$J_{T}(u,\phi) = E \int_{0}^{T} [x'(t)Qx(t) + u'(t)Ru(t)]dt \qquad (6.74)$$

is minimized, where  $T < \infty$ .

Before we define the set of admissible controls, we establish some notation. By the variation of constants formula, we obtain

$$z(t) = \int_{0}^{t} C\Phi(s,0)\phi(0)ds + \int_{0}^{t} \int_{-\tau}^{0} C\Phi(s,\sigma+\tau)B\phi(\sigma)d\sigma ds$$
  
+ 
$$\int_{0}^{t} \int_{0}^{s} C\Phi(s,\sigma)Gu(\sigma)d\sigma ds + \int_{0}^{t} \int_{0}^{s} C\Phi(s,\sigma)M'dw(\sigma)ds + Nw_{2}(t)$$
  
$$\triangleq z_{0}(t) + \int_{0}^{t} \int_{0}^{s} \Phi(s,\sigma)Gu(\sigma)d\sigma ds \qquad (6.75)$$

Define the set U consisting of the class of processes u(t) satisfying the following conditions:

- (i) u(t) is measurable with respect to  $\sigma\{z(s), 0 \le s \le t\}$ for each t;
- (ii) For each  $u \in U_0$ , there exist unique solutions to (6.72) and (6.73); (iii)  $\int_0^T E|u(t)|^2 dt < \infty$
- (iv) For each  $u \in U_0$ ,  $\sigma\{z(s), 0 \le s \le t\} = \sigma\{z_0(s), 0 \le s \le t\}$ . In words, (iv) means that by using a control law  $u \in U_0$ , we cannot gain, from the resulting outputs, more information about the system than that obtained by using no control at all. We shall take  $U_0$  to be the set of admissible controls. The following theorem has been proved by Lindquist [56].

<u>Proposition 6.6.1</u> The problem of determining  $u \in U_0$  so as to minimize (6.74) has the following solution

$$u^{*}(t) = [-R^{-1}G'K_{o}(t)\hat{x}(t|t) - R^{-1}G'\int_{-\tau}^{0}K_{1}(t,\theta)\hat{x}(t+\theta|t)d\theta \qquad (6.76)$$

where  $K_0(t)$  and  $K_1(t,\theta)$  are the optimal gains for the deterministic optimal control problem (section 6.2), and  $\hat{x}(s|t)$ ,  $t-\tau \leq s \leq t$  is the conditional expectation of x(s) given  $z(\sigma)$ ,  $0 \leq \sigma \leq t$  (section 5.5).

<u>Remark 6.6.1</u> A few comments on the choice of  $U_0$  as the set of admissible controls are in order. Let us define the set U to be the class of processes u(t) that are measurable with respect to  $\sigma\{z(s), 0 \le s \le t\}$  for each t, such that for each u  $\in$  U, there exist unique solutions to (6.72) and (6.73), and for which  $\int_0^T E|u(t)|^2 dt < \infty$ .

Clearly  $U_0 \subset U$ . The basic difference between the sets U and  $U_0$  is that the requirement  $\sigma\{z(s), 0 \le s \le t\} = \sigma\{z_0(s), 0 \le s \le t\}$  is not imposed on U. However, if we examine the proof of Proposition 6.6.1 in [56], we see that the requirement of  $\sigma\{z(s), 0 \le s \le t\} = \sigma\{z_0(s), 0 \le s \le t\}$ , for any u in the set of admissible laws, is crucial. One would really like to prove that the separation theorem holds also for the set U (which is equivalent to showing  $U_0 = U$ ) but the truth or falsehood of this is not known at present. We may note that in the separation theorem without delays [79], the conditions (i) to (iv) satisfied by all u  $\varepsilon$  U<sub>0</sub> are obtained by requiring the control laws to be Lipschitz functionals of the past observations. Even in the case without delays, it has not been established that  $U_0 = U$ .

The expression for the optimal cost is given in

Lemma 6.6.1 Corresponding to the optimal control (6.76), the optimal cost J\* associated with the stochastic control problem (6.72) - (6.74) is given by

$$J^{*} = EV(x_{o}) + \int_{0}^{T} tr M'MK_{o}(t) dt$$

$$+ \int_{0}^{T} tr \{K_{o}(t) GR^{-1}G'K_{o}(t)P_{o}(t) + \int_{-\tau}^{0} K_{1}'(t,\theta) GR^{-1}G'K_{o}(t)P_{1}'(t,\theta) d\theta$$

$$+ \int_{-\tau}^{0} K_{o}(t) GR^{-1}G'K_{1}(t,\theta)P_{1}(t,\theta) d\theta + \int_{-\tau}^{0} \int_{-\tau}^{0} K_{1}'(t,\theta) GR^{-1}G'K_{1}(t,\xi) \cdot P_{2}(t,\xi,\theta) d\theta d\xi \} dt \qquad (6.77)$$

where 
$$V(x_t) = x'(t)K_0(t)x(t) + \int_{-\tau}^0 x'(t)K_1(t,\theta)x(t+\theta)d\theta$$
  
+  $\int_{-\tau}^0 x'(t+\theta)K_1'(t,\theta)x(t)d\theta + \int_{-\tau}^0 \int_{-\tau}^0 x'(t+\theta)K_2(t,\theta,\xi)x(t+\xi)d\theta d\xi$   
(6.78)

Proof: See Appendix C.

<u>Remark 6.6.2</u> Lemma 6.6.1 shows that the optimal cost in the delay case has the same structure as the case without delays: there is a term due to initial conditions, a term due to the noise in the system dynamics, and a term due to the estimation error.

We turn our attention now to the stochastic control system defined by using the steady state version of (6.76):

$$u(t) = -R^{-1}G'K_{0}\hat{x}(t|t) - R^{-1}G'\int_{-\tau}^{0}K_{1}(\theta)\hat{x}(t+\theta|t)d\theta$$
 (6.79)

where  $\hat{x}(t|t)$  is generated by the steady state optimal filter. Heuristically, this law is the optimal one for the functional

$$E \int_{0}^{\infty} [x'(t)Qx(t) + u'(t)Ru(t)]dt \qquad (6.80)$$

with observations started back at  $-\infty$ , except that it would give rise to an infinite J\*. One must therefore modify the control problem to be one of minimizing

$$\lim_{T \to \infty} \frac{1}{T} \left\{ E \int_0^T \left[ x'(t) Q x(t) + u'(t) R u(t) \right] dt \right\}$$

or 
$$\lim_{T \to \infty} \frac{1}{T} \int_0^T [x'(t)Qx(t) + u'(t)Ru(t)]dt$$

This introduces notions of ergodicity and invariant measures into the discussion. Even in the case without delays, the infinite problem has not really been completely resolved. We therefore content ourselves with simply the result on asymptotic stability of the closed-loop system under the law (6.79).

<u>Theorem 6.6.1</u> Let Q = H'H. Assume (A,B,G) is stabilizable, (A,B,M') is controllable, (A,B,C) is detectable, and (A,B,H) is observable. Then the control law

$$u(t) = -R^{-1}G'K_{o}\hat{x}(t|t) - R^{-1}G' \int_{-\tau}^{0} K_{1}(\theta)x(t+\theta|t)d\theta,$$

where  $\hat{x}(t|t)$  is generated by the steady state filter of Theorem 6.4.1, and  $K_0$ ,  $K_1(\theta)$  are generated by the deterministic stationary control law of Theorem 6.2.1, gives rise to an asymptotically stable closedloop system.

<u>Proof</u>: Stabilizability of (A,B,G) ensures that  $K_0$ ,  $K_1(\theta)$ , and  $K_2(\theta,\xi)$  are well-defined. Observability of (A,B,H) then guarantees that solutions of the system

$$x(t) = (A - GR^{-1}G'K_{0})x(t) + Bx(t-\tau) - \int_{-\tau}^{0} GR^{-1}G'K_{1}(\theta)x(t+\theta)d\theta$$
(6.81)

are asymptotically stable (see Theorem 6.2.1). Detectability of (A,B,C) ensures that the steady state filter is well-defined, and controllability of (A,B,M') guarantees that the steady state filter is asymptotically stable. The closed-loop system is defined by the coupled set of equations

$$dx(t) = Ax(t)dt + Bx(t-\tau)dt - GR^{-1}G'K_{0}\hat{x}(t|t)$$
$$- \int_{-\tau}^{0} GR^{-1}G'K_{1}(\theta)\hat{x}(t+\theta|t)d\theta + dw(t) \qquad (6.82)$$

$$d\hat{\mathbf{x}}(t|t) = A\hat{\mathbf{x}}(t|t)dt + B\hat{\mathbf{x}}(t-\tau|t-\tau)dt$$

+B 
$$\int_{t-\tau}^{t} P_2(-\tau,s-t)C'(NN')^{-1}[dz(s)-C\hat{x}(s|s)ds]dt$$

$$+P_{O}C'(NN')^{-1}[dz(t) - C\hat{x}(t|t)dt]$$
(6.83)

Expressing  $\hat{x}(t|t) = x(t) - e(t|t)$ , we get

$$dx(t) = (A - GR^{-1}G'K_{0})x(t)dt + Bx(t-\tau)dt$$

$$- \int_{-\tau}^{0} GR^{-1}G'K_{1}(\theta)x(t+\theta)d\theta + dw(t)$$

$$+ GR^{-1}G'K_{0}e(t|t) + \int_{-\tau}^{0} GR^{-1}G'K_{1}(\theta)e(t+\theta|t)d\theta \quad (6.84)$$

and 
$$de(t|t) = [A-P_{O}C'(NN')^{-1}C]e(t|t)dt + Be(t-\tau|t-\tau)dt$$

+ 
$$\int_{-\tau}^{0} BP_{2}(-\tau,\theta)C'(NN')^{-1}Ce(t+\theta|t+\theta)d\theta dt$$
  
+ 
$$dw(t) - P_{0}C'R^{-1}dv(t) - B \int_{-\tau}^{0} P_{2}(-\tau,\theta)C'R^{-1}d_{\theta}v(t+\theta)dt$$

(6.85)

Since (6.85) is decoupled from (6.84), the stability properties of the closed-loop system are precisely those of (6.81) and the steady state optimal filter. Since both of these are asymptotically stable as a consequence of our assumptions, the closed-loop stochastic control system is asymptotically stable as well.

<u>Remark 6.6.3</u> When the control law (6.79) is applied to the problem of minimizing

$$J_{r} = \lim_{T \to \infty} \frac{1}{T} E \int_{0}^{T} [x'(t)Qx(t) + u'(t)Ru(t)]dt$$

Lemma 6.6.1 shows

$$J_{r} = trM'MK_{o}$$

$$+ tr\{K_{o}GR^{-1}G'K_{o}P_{o} + \int_{-\tau}^{0} K_{1}'(\theta)GR^{-1}G'K_{o}P_{1}'(\theta)d\theta$$

$$+ \int_{-\tau}^{0} K_{o}GR^{-1}G'K_{1}(\theta)P_{1}(\theta)d\theta + \int_{-\tau}^{0} \int_{-\tau}^{0} K_{1}'(\theta)GR^{-1}G'K_{1}(\xi)P_{2}(\xi,\theta)d\xi d\theta\}$$

This can be thought of as the cost rate associated with the control law (6.79).

# 6.7 <u>Filter Stability for Systems with Delays in</u> The Observations

In the previous sections, we have discussed the stability properties of the optimal filter and control system rather thoroughly for linear stochastic systems with a single delay in the system dynamics. It is relatively straightforward to extend the duality theorem of section 6.3

-166-

to cover situations with multiple delays in the state dynamics. Combining this with the results of Delfour, McCalla, and Mitter [32] on optimal control, it should not be difficult to extend our stability results to these systems. What is more interesting is the case in which there are delays in the observations. Since similar difficulties arise for distributed delays as well as point delays, we shall, for convenience, consider systems of the form

$$dx(t) = [Ax(t) + Bx(t-\tau)]dt + dw(t)$$

$$x(\theta) = 0 \quad -\tau \leq \theta \leq 0$$
(6.86)

$$dz(t) = [C_{0}x(t) + C_{1}x(t-\tau)]dt + dv(t)$$
(6.87)

From the results of Chapter 5, we know that the optimal filter is given by the equations

$$d\hat{x}(t|t) = [A\hat{x}(t|t) + B\hat{x}(t-\tau|t)]dt + [P_{o}(t)C_{o}'R^{-1} + P_{2}(t,0,-\tau)C_{1}'R^{-1}] \cdot [dz(t) - C_{o}\hat{x}(t|t)dt - C_{1}\hat{x}(t-\tau|t)dt] (6.88) \hat{x}(t+\theta|t) = \hat{x}(t+\theta|t+\theta) + \int_{t+\theta}^{t} [P_{2}(s,t+\theta-s,0)C_{o}' + P_{2}(s,t+\theta-s,-\tau)C_{1}']R^{-1}$$

$$\cdot [dz(s) - C_{o}\hat{x}(s|s)ds - C_{1}\hat{x}(s-\tau|s)ds]$$
(6.89)

If we apply the same technique as that of section 6.4 to (6.88) - (6.89), we see that

$$d\hat{x}(t|t) = [A\hat{x}(t|t) + B\hat{x}(t-\tau|t-\tau)]dt$$

$$+ B \int_{t-\tau}^{t} [P_{2}(s,t-\tau-s,0)C_{0}' + P_{2}(s,t-\tau-s,-\tau)C_{1}']R^{-1}$$

$$\cdot [dz(s) - C_{0}\hat{x}(s|s)ds - C_{1}\hat{x}(s-\tau|s)ds]dt$$

$$+ [P_{0}(t)C_{0}' + P_{2}(t,0,-\tau)C_{1}']R^{-1}$$

$$\cdot [dz(t) - C_{0}\hat{x}(t|t)dt - C_{1}\hat{x}(t-\tau|t)dt]$$
(6.90)

which is not a stochastic delay equation in  $\hat{x}(t|t)$ . Indeed, it is not possible to express  $\hat{x}(t-\tau|t)$  solely in terms of  $\hat{x}(s|s)$ ,  $t - \tau \leq s \leq t$ . Without this, the technique presented in section 6.4 cannot be applied.

The situation is very similar in the dual control problem, which is to minimize:

$$J_{T}(b,u) = \int_{0}^{T} [y'(t)Qy(t) + u'(t)Ru(t)]dt$$

subject to the constraint

$$\dot{y}(t) = -A'y(t) - B'y(t+\tau) - C_0u(t) - C_1u(t+\tau)$$

$$y(T) = b$$

$$y(s) = 0 \qquad s > T$$

$$u(s) = 0 \qquad s > T$$

Theorem 6.3.1 shows that the optimal estimate of b'x(T) for (6.86) - (6.87) is given by  $\int_0^T u'_T(s)dz(s)$ , where  $u_T(t)$  is the optimal control to the

dual problem.

By the time reversal technique of section 6.4, we see that the above control problem is equivalent to an optimal control problem for systems with delays in the control as well as in the state. It is known [80] that this problem is much more complex than the optimal control problem with delays in the state only. To understand the structure of the optimal filter when there are delays in the observations, it is worthwhile to discuss briefly the control problem with delays in both the state and control.

Consider the system

$$\dot{\mathbf{x}}(t) = A\mathbf{x}(t) + B\mathbf{x}(t-\tau) + C_0 \mathbf{u}(t) + C_1 \mathbf{u}(t-\tau)$$
$$\mathbf{x}(\theta) = \phi(\theta) \qquad \theta \in [-\tau, 0]$$
$$\mathbf{u}(\theta) = 0 \qquad \theta \in [-\tau, 0) \qquad (6.91)$$

The problem is to find a control u such that the cost functional

$$J_{T}(u,\phi) = \int_{0}^{T} [x'(t)Qx(t) + u'(t)Ru(t)]dt$$

is minimized. The solution to this problem is given in [80]:

$$u(t) = -R^{-1}C_{0}'[P_{c}(t,t,t)x(t) + \int_{t-\tau}^{t} P_{c}(t,s+\tau,t)Bx(s)ds + \int_{t-\tau}^{t} P_{c}(t,s+\tau,t)C_{1}u(s)ds -\gamma(t)R^{-1}C_{1}'[P_{c}(t+\tau,t,t)x(t) + \int_{t-\tau}^{t} P_{c}(t+\tau,s+\tau,t)Bx(s)ds + \int_{t-\tau}^{t} P_{c}(t+\tau,s+\tau,t)C_{1}u(s)ds] + (6.92)$$

where 
$$\gamma(t) = \begin{cases} 0 & \text{if } t \in [T-\tau,T] \\ 1 & \text{otherwise} \end{cases}$$

and  $P_{c}(t,s,r)$  is a type of Fredholm kernel introduced by Manitius [71] and satisfies appropriate partial differential equations.

The optimal control u(t) is not expressed solely in terms of  $x_t$ , but also in terms of u(s),  $t-\tau \leq s \leq t$ . In order to express u(t) solely in terms of x, we can regard (6.92) as an integral equation in u(t). In fact, if we write

$$q(t) = -R^{-1}C_{0}'[P_{c}(t,t,t)x(t) + \int_{t-\tau}^{t} P_{c}(t,s+\tau,t)Bx(s)ds]$$
  
- $\gamma(t)R^{-1}C_{1}'[P_{c}(t+\tau,t,t)x(t) + \int_{t-\tau}^{t} P_{c}(t+\tau,s+\tau,t)Bx(s)ds]$   
N(t,s) = -[R^{-1}C\_{0}'P\_{c}(t,s+\tau,t)C\_{1} + \gamma(t)R^{-1}C\_{1}'P\_{c}(t+\tau,s+\tau,t)C\_{1}]

and

we get

$$u(t) = q(t) + \int_{t-T}^{t} N(t,s)u(s)ds$$
 (6.94)

The properties of  $P_c(t,s,r)$  [80] clearly implies that N(t,s) is a Volterra  $L_2$ -kernel. Substituting (6.94) into itself, we obtain

$$u(t) = q(t) + \int_{t-\tau}^{t} N(t,s)q(s)ds$$
  
+ 
$$\int_{t-\tau}^{t} N(t,s) \int_{s-\tau}^{s} N(s,r)u(r)drds$$
  
= 
$$q(t) + \int_{t-\tau}^{t} N(t,s)q(s)ds +$$
  
+ 
$$\int_{t-\tau}^{t} \int_{t-\tau}^{t} N(t,s)q(s)ds +$$

$$\int_{t-2\tau} \int_{max(t-\tau,r)} h(t,s)h(s,r) ds d(r) dr$$

By induction, it is easily seen that in fact u(t) satisfies a Volterra integral equation of the form

$$u(t) = q(t) + \int_0^t M(t,s)u(s)ds$$

with M(t,s) an  $L_2$ -kernel. The solution for the optimal control u in state feedback form can therefore be expressed as

$$u(t) = q(t) + \int_{0}^{t} R(t,s)q(s)ds$$
  
=  $\int_{-\tau}^{t} d_{s}F(t,s)x(s)$  (6.95)

where F(t,s) is an  $L_2$ -kernel of bounded variation in s. Notice that in state feedback form, the entire past of the state trajectory must be used. There are two ways to study the stability of the closed-loop system. We can either study the system defined by (6.91) and (6.94) as system of coupled differential and integral equations, or we can substitute (6.95) into (6.91) and study the resulting integrodifferential equation. In any case, the situation is considerably more complex than that without delays in the control. Stability properties of the optimal system are left as a subject for future research.

One can see that a similar situation happens in the optimal filter for systems with delays in the observations. Defining

$$W(t,s) = [P_2(s,t-\tau-s,0)C'_0 + P_2(s,t-\tau-s,-\tau)C'_1]R^{-1},$$
  
$$\hat{x}(t-\tau|t) = \hat{x}(t-\tau|t-\tau) + \int_{t-\tau}^{t} W(t,s)[dz(s)-C_0\hat{x}(s|s)ds]$$

$$-\int_{t-\tau}^{t} W(t,s)C_{1}\hat{x}(s-\tau|s)ds \qquad (6.96)$$

One can view (6.96) as an integral equation in  $\hat{x}(t-\tau|t)$ . Again the kernel W(t,s)C<sub>1</sub> is a deterministic L<sub>2</sub>-kernel. As before, we can solve  $\hat{x}(t-\tau|t)$  in terms of  $\hat{x}(s|s)$ ,  $0 \leq s \leq t$ . The stability of the optimal filter can thus be studied from either an integrodifferential equation or from a coupled set of differential and integral equations. In view of the duality theorem between estimation and control, the stability of the dual control system should be intimately related to the stability of the optimal filter, in the same sense as the connection discussed in section 6.4.

### 6.8 Perspective

In this chapter, we have studied the stability of the optimal linear control system and the optimal linear filter for systems with delays only in the state dynamics. Since the main results are interconnected in a rather complicated way, it seems worthwhile to recapitulate the various steps involved and put our results in perspective.

We first showed that under the conditions of stabilizability and observability, the infinite-time optimal linear control system with quadratic cost is asymptotically stable. This result not only extends previous work by other investigators, but also shows the important role played by system-theoretic concepts.

In section 6.3, we established a duality theorem which relates the solution of the optimal linear filtering problem to the solution of a dual optimal control problem. This result was motivated by the work of Lindquist [75], and Mitter and Vinter [67], and paved the way for exploiting the results of section 6.2 in studying the stability of the optimal linear filter.

Theorem 6.4.1 in section 6.4 contains the central results of this chapter, and combines the results of Chapter 2 with those of sections 6.2 and 6.3. First, we used the duality theorem to give the precise formulas relating the optimal gains and the fundamental matrix of the optimal filter to those of the corresponding dual optimal control system. This is the content of Lemmas 6.4.1 through 6.4.4. Next, we used the duality results of Chapter 2 to assert that

-173-

detectability and controllability of the original system corresponds to stabilizability and observability of the dual system. Thus, under the assumptions of detectability and controllability of the original system, we were able to invoke the stability result of section 6.2 to conclude asymptotic stability of the dual optimal control system. Finally, the formulas relating the optimal filter and the dual optimal control system allowed us to conclude asymptotic stability of the optimal filter from that of the dual control system. Once these results were established, it was straightforward to prove, in section 6.5, that the closed-loop stochastic control system is also asymptotically stable. While the analogs of these results for ordinary differential systems are well-known, they are new for linear delay systems, and constitute the first complete extension of the finite dimensional linear-quadratic-Gaussian theory. Furthermore, they bring into focus the importance of the structural properties and duality relations in control and estimation problems for linear delay systems. While our results deal only with a special class of delay systems, we feel that the ideas and techniques will be fundamental to the development of a complete linear-quadratic-Gaussian theory for more general delay systems. We hope that the work reported in this thesis will serve as a stimulus to the development of such a theory.

-174-

## CHAPTER 7

## CONCLUSIONS AND SUGGESTIONS FOR FUTURE RESEARCH

In this thesis, we have studied two areas in the theory of delay systems: structural properties and their applications to control in linear time-invariant delay systems, and optimal linear and nonlinear filtering for stochastic delay systems. We summarize our findings here and indicate what we feel are the main contributions of the thesis.

In Chapter 2, we discussed the various notions of controllability, stabilizability, observability, and detectability in connection with linear delay systems. New definitions of controllability and observability were given and some duality results established. While we have not conducted an in-depth study of these concepts, the ideas presented have proved to be very useful in connection with linear optimal control with quadratic cost and optimal linear filtering.

In Chapter 3, we studied the concept of pointwise degeneracy. We completely characterized 3 dimensional systems which can be made pointwise degenerate by delay feedback, and gave sufficient conditions for this to be possible in higher dimensional systems. The conditions obtained are felt to be much simpler and more intuitive than previous results obtained by other investigators.

In Chapter 4, we applied the pointwise degeneracy property to the control of linear systems. A complete solution to the minimum

-175-

time output zeroing problem by delay feedback was given for 3 dimensional linear systems, while sufficient conditions were given for the solvability of the problem in higher dimensions. The solvability conditions are again very simple, although the actual construction of the feedback controllers is rather complicated. Sensitivity of the control system under perturbations of its parameters was also studied and it was shown that "small" perturbations give rise to "small" degradation in the system performance. Some examples were then given to illustrate the theory. All these results are new and show that the concept of delay feedback has potential applications in linear system theory which are worth exploring further.

We began the study of the optimal filtering problem in Chapter 5. The general nonlinear filtering problem was first studied, and a representation theorem for conditional moment functionals derived. The form of the representation is new even when specialized to ordinary differential systems, and can be thought of as solving the nonlinear filtering problem abstractly. We then showed that if the moment functionals were suitably "smooth," stochastic differential equations could be derived. When these results were specialized to the linear case, we obtained a complete solution to the optimal linear filtering problem, generalizing previously known results.

In Chapter 6, we studied the stability properties of the optimal linear filter. The special case where the observations do not involve any delay terms was considered. We showed that there is a duality relation between optimal control with quadratic cost and

-176-

optimal linear filtering. Under the hypothesis of stabilizability and observability, we showed that the optimal closed-loop system for the infinite time control problem is asymptotically stable. Under the hypothesis of detectability and controllability, we showed that the optimal linear filter is asymptotically stable. Finally, the closed-loop stochastic control system using the deterministic optimal control law and the optimal filter estimate was also shown to be asymptotically stable. This we feel is the first complete extension of the familiar linear-quadratic-Gaussian theory for ordinary differential systems. On the one hand, almost complete analogs of the finite dimensional results have been obtained: the linear feedback nature of the solutions for the optimal control and filtering problems, the Riccati-type partial differential equations, the duality between control and filtering, and the stability results based on system-theoretic concepts. On the other hand, the additional complications due to the presence of delays are also clearly demonstrated by the form of the solutions. These similarities and differences between the finite dimensional theory and the theory developed in this work give a lot of insight into the behavior of linear delay systems.

The results obtained in this thesis suggest that further investigations in the following topics would be fruitful.

 Study in greater depth the relations between the various notions of controllability, stabilizability, observability, and detectability for delay systems. In particular, algebraic

-177-

conditions for checking these properties would be very useful. In addition to the notions discussed in Chapter 2, there are other algebraic formulations of the notions of controllability and observability [42], [43]. It would be very worthwhile to relate these various approaches and obtain a unified theory. This is very much an open area of research.

- 2. Extend the applications of the pointwise degeneracy property to the construction of delay observers. We have given an example of a "deadbeat" observer. The general theory, however, is lacking, and it would be interesting to see what precise conditions are required. It is suspected that this problem will be related to the properties of the adjoint system.
- 3. In studying the nonlinear filtering problem, we have shown that in order to derive stochastic differential equations, moment functionals must be in the domain of the infinitesimal generator of the Markov process  $x_t$ . The characterization of this operator is important in its own right as well as in applying the results of Chapter 5. It is hoped that a detailed characterization of this operator will pave the way for useful approximations to the optimal nonlinear filter.
- 4. Extend the stability results of Chapter 6 to cover more general models, for example, those with delays in the observations. As discussed in Chapter 6, this will be

-178-

intimately related to the control problem with delays in the control. It is not clear what techniques are suitable for this problem. In fact, it is not clear whether the controllability and observability concepts which have proved fruitful need to be reformulated or not. Since many practical systems have delays in the observations, this is an important practical as well as theoretical problem.

### APPENDIX A

### PROOF OF LEMMAS 4.1 AND 4.2

We refer the reader to Chapter 4 for the statement of these lemmas.

<u>Proof of Lemma 4.1</u>: The extrema of the function  $f(\lambda) \stackrel{\Delta}{=} (\alpha_1 + \alpha_2 \lambda + \alpha_3 \lambda^2) e^{\lambda \tau}$ occur at

$$\lambda = \frac{-(\alpha_2 \tau + 2\alpha_3) \pm \sqrt{\alpha_2^2 \tau^2 + 4\alpha_3^2 - 4\alpha_1 \alpha_3 \tau^2}}{2\alpha_3 \tau}$$

Since  $\alpha_2^2 - 4\alpha_1 \alpha_3 > 0$ , there are two real extrema for  $f(\lambda)$ . Also, the zeros of  $f(\lambda)$  occur at the zeros of  $\alpha_1 + \alpha_2 \lambda + \alpha_3 \lambda^2$ , and these are at

$$\frac{-\alpha_2 \pm \sqrt{\alpha_2^2 - 4\alpha_1 \alpha_3}}{2\alpha_3}$$

Again, since  $\alpha_2^2 - 4\alpha_1 \alpha_3 > 0$ , these give two real zeros of  $f(\lambda)$ . Let

$$s_{1} = \frac{-\alpha_{2} + \sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}}}{2\alpha_{3}}$$
$$s_{2} = \frac{-\alpha_{2} - \sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}}}{2\alpha_{3}}$$

Let us perturb  $\boldsymbol{\lambda}_{i}$  by some number  $\boldsymbol{\epsilon}$  and let

$$\eta_{\varepsilon}(s_{i}) = \alpha_{3}(s_{i}+\varepsilon)^{2} + \alpha_{2}(s_{i}+\varepsilon) + \alpha_{1}$$
$$= \varepsilon(\alpha_{3}\varepsilon + 2s_{i}\alpha_{3} + \alpha_{2})$$

•
Hence 
$$\eta_{\varepsilon}(s_1) = \varepsilon(\alpha_3 \varepsilon + \sqrt{\alpha_2^2 - 4\alpha_1 \alpha_3})$$
  
 $\eta_{\varepsilon}(s_2) = \varepsilon(\alpha_3 \varepsilon - \sqrt{\alpha_2^2 - 4\alpha_1 \alpha_3})$ 

This yields

$$\begin{split} n_{\varepsilon}(\mathbf{s}_{1}) &\geq 0 & \text{ if } \varepsilon \geq 0, \text{ and } \varepsilon \alpha_{3} \geq -\sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \\ &< 0 & \text{ if } \varepsilon \geq 0, \text{ and } \varepsilon \alpha_{3} \leq -\sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \\ &\geq 0 & \text{ if } \varepsilon \leq 0, \text{ and } \varepsilon \alpha_{3} \leq -\sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \\ &< 0 & \text{ if } \varepsilon \leq 0, \text{ and } \varepsilon \alpha_{3} \geq -\sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \end{split}$$

Similarly,

$$\begin{split} &\eta_{\varepsilon}(\mathbf{s}_{2}) \geq 0 \qquad \text{if } \varepsilon \geq 0, \text{ and } \varepsilon \alpha_{3} \geq \sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \\ &< 0 \qquad \text{if } \varepsilon \geq 0, \text{ and } \varepsilon \alpha_{3} \leq \sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \\ &\geq 0 \qquad \text{if } \varepsilon \leq 0, \text{ and } \varepsilon \alpha_{3} \leq \sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \\ &< 0 \qquad \text{if } \varepsilon \leq 0, \text{ and } \varepsilon \alpha_{3} \geq \sqrt{\alpha_{2}^{2} - 4\alpha_{1}\alpha_{3}} \end{split}$$

Let us now examine the graph of the function f( $\lambda$ ). There are eight cases, depending on the signs of the  $\alpha_i$ 's.

Case I:  $\alpha_1 > 0$ ,  $\alpha_2 > 0$ ,  $\alpha_3 > 0$ . In this case  $s_1$  and  $s_2$  are both negative. If  $\varepsilon > 0$ ,  $\eta_{\varepsilon}(s_1) > 0$ , and if

$$\frac{-\sqrt{\alpha_2^2-4\alpha_1\alpha_3}}{\alpha_3} < \varepsilon < 0, \ \eta_{\varepsilon}(s_1) < 0.$$

A little thought now shows that the graph of  $f(\lambda)$  is of the form

-181-



It is obvious that for any finite real  $\beta \neq s_1$  or  $s_2$ , there exists a  $\gamma_{\beta}$  such that the line  $\gamma_{\beta}(\lambda-\beta)$  intersects  $f(\lambda)$  at three real distinct values of  $\lambda$ , none of which equals  $s_1$  or  $s_2$ . There also exists a constant  $\omega(\omega > 0$  in this case) such that the line  $\lambda=\omega$  intersects  $f(\lambda)$  at three real distinct points.

For the other seven cases, entirely similar arguments show that  $f(\lambda)$  must have one positive maximum and one negative minimum, and hence the required straight lines exist. We shall omit the details. The lemma is now proved.

<u>Proof of Lemma 4.2</u>: Since C is of full rank, we may assume, without loss of generality, that  $q'c_1 = \alpha_3 \neq 0$ , where  $c_i$  is the i<sup>th</sup> column of C. Since (A,C) is controllable, we have the following possibilities:

(i)  $\{c_1, Ac_1, c_2\}$  are linearly independent or (ii)  $\{c_1, c_2, Ac_2\}$  are linearly independent Case (i):  $\{c_1, Ac_1, c_2\}$  linearly independent.

Construct a matrix  $K_1$  as follows:

$$K_1 = [\underline{0} \ e_2 \ \underline{0}] [c_1 \ Ac_1 \ c_2]^{-1}$$

where  $e_2$  is the second natural Euclidean basis vector in 2 dimensions, and <u>0</u> is the null vector. Then

$$(A + CK_1)c_1 = Ac_1$$
  
 $(A + CK_1)^2 = A^2c_1 + c_2$ 

and  $(A_1, c_1)$  is controllable, where  $A_1 = (A + CK_1)$ .

Let the characteristic polynomial of A<sub>1</sub> be  $p(s) = s^3 + p_2 s^2 + p_1 s + p_0$ . We can then write

$$c_{2} = \beta_{1} (A_{1}^{2} c_{1}^{+p} A_{1}^{A} c_{1}^{+p} c_{1}^{A}) + \beta_{2} (A_{1}^{-1} c_{1}^{+p} c_{1}^{-1}) + \beta_{3}^{-1} c_{1}^{-1}$$

for some  $\beta_1$ ,  $\beta_2$ , and  $\beta_3$ . The independence of  $c_1$ ,  $Ac_1$ , and  $c_2$  implies  $\beta_1 \neq 0$ . The transformation

$$P = [A_1^2 c_1 A_1 c_1 c_1] \begin{bmatrix} 1 & 0 & 0 \\ p_2 & 1 & 0 \\ p_1 & p_2 & 1 \end{bmatrix}$$

gives  $P^{-1}A_1P$  in companion form and

$$P^{-1}C = \begin{bmatrix} 0 & \beta_1 \\ 0 & \beta_2 \\ 1 & \beta_3 \end{bmatrix}$$

Let  $q'P = (\alpha_1 \alpha_2 \alpha_3)$ . Then

$$\alpha_{1} = q'(A^{2}c_{1}+c_{2}) + p_{2}q'Ac_{1}+p_{1}\alpha_{3}$$

$$\alpha_{2} = q'Ac_{1}+p_{2}\alpha_{3}$$

$$\alpha_{3} = q'c_{1}$$

If  $\alpha_2^2 - 4\alpha_1\alpha_3 > 0$ , and  $\frac{\beta_2}{\beta_1}$  is not a zero of  $\eta(\lambda) = \alpha_1 + \alpha_2 \lambda + \alpha_3 \lambda^2$ , we are done. If not, take a matrix F of the form

$$F = \begin{bmatrix} -f_{0} & -f_{1} & -f_{2} \\ 0 & 0 & 0 \end{bmatrix}$$

then  $P^{-1}(A+CK_1)P + P^{-1}CF$ 

$$= P^{-1} (A + CK_{1} + CFP^{-1})P = P^{-1}A_{2}P$$
$$= \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ -P'_{0} & -P'_{1} & -P'_{2} \end{bmatrix}$$

with  $p'_i = p_i + f_i$ . Define a matrix  $P_1$  by

$$P_{1} = P^{-1} [A_{2}^{2}c_{1} A_{2}c_{1} c_{1}] \begin{bmatrix} 1 & 0 & 0 \\ p_{2}^{\prime} & 1 & 0 \\ p_{1}^{\prime} & p_{2}^{\prime} & 1 \end{bmatrix}$$

 $\stackrel{\text{A}}{=} P^{-1}P_2$ 

Then 
$$P_1^{-1}P^{-1}A_2PP_1 = P_2^{-1}A_2P_2 = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ -P'_0 & -P'_1 & -P'_2 \end{bmatrix}$$
  
Furthermore  $P_1^{-1}P^{-1}C = P_2^{-1}C = \begin{bmatrix} 0 & d_{12} \\ 0 & d_{22} \\ 1 & d_{32} \end{bmatrix}$ , for some  $d_{12}$ ,  $d_{22}$ ,  $d_{32}$ 

Now  $A_2c_1 = (A+CK_1)c_1+CFP^{-1}c_1$ 

$$= \operatorname{Ac}_{1} + C \begin{bmatrix} -f_{2} \\ 0 \end{bmatrix} = \operatorname{Ac}_{1} - f_{2}c_{1}$$

Similarly 
$$A_2^2 c_1 = A^2 c_1 + c_2 - f_2 A c_1 + (f_2^2 - p_2 f_2 + f_1) c_1$$

Choose  $f_0$ ,  $f_1$ , and  $f_2$  to satisfy the properties

(a) 
$$\alpha_2'^2 - 4\alpha_1'\alpha_3' > 0$$
, where  
 $\alpha_3' = \alpha_3 = q'c_1$   
 $\alpha_2' = q'A_2c_1 + p'2\alpha_3 = q'Ac_1 - f_2q'c_1 + (p_2 + f_2)\alpha_3 = \alpha_2$ 

and  $\alpha'_{1} = q'A_{2}^{2}c_{1}+p'_{2}q'A_{2}c_{1}+p'_{1}\alpha_{3}$ 

$$= q'(A^{2}c_{1}+c_{2})+p_{2}q'Ac_{1}+(f_{2}^{2}-f_{2}-p_{2}f_{2}+p_{1}+2f_{1})\alpha_{3}$$

and  
(b) 
$$\frac{d_{22}}{d_{12}}$$
 is not a zero of  $\eta'(\lambda) = \alpha_1' + \alpha_2 \lambda + \alpha_3 \lambda^2$ .

A little algebra shows that in fact  $d_{12} = \beta_1$ ,  $d_{22} = \beta_2$ , and  $d_{32} = \beta_1 p_1 + \beta_3 - 2\beta_1 f_1 + 2\beta_1 p_2 f_2$ .

Some reflection now shows that appropriate choices for  $f_0$ ,  $f_1$ , and  $f_2$  exist. This completes the consideration of case (i).

Case (ii):  $\{c_1, c_2, Ac_2\}$  linearly independent.

We construct a matrix  $K_2$  as follows

$$K_2 = [e_2 \underline{0} \underline{0}][c_1 c_2 Ac_2]^{-1}$$

Let  $A_3 = A+CK_2$ . Then  $A_3c_1 = Ac_1+c_2$  and  $A_3^2c_1 = A^2c_1+Ac_2+CK_2Ac_1$ . Let the characteristic polynomial of  $A_3$  be  $p(s) = s^3+p_2s^2+p_1s+p_0$ . We can write

$$c_{2} = \beta_{1} [A_{3}^{2} c_{1}^{+p} 2^{A_{3}} c_{1}^{+p} 1 c_{1}^{]+\beta} 2^{[A_{3}} c_{1}^{+p} 2^{c_{1}^{}]+\beta} 3^{c_{1}}$$

 $\beta_2$  is clearly nonzero. If  $\beta_1 \neq 0$ , we carry out the constructions in exactly the same fashion as in case (i). This shows that we can again find appropriate matrices P and K such that  $P^{-1}(A+CK)P$ , q'P, and  $P^{-1}C$ have the required structure. The proof of the lemma is completed.

## APPENDIX B

## DERIVATION OF EQUATIONS (5.53) - (5.55)

To derive the appropriate equations for  $P(t,\theta,\xi)$ , we have to consider three cases:

- (i)  $0 \leq t \leq T$ ,  $-\tau \leq \theta < 0$ ,  $-\tau \leq \xi < 0$
- (ii)  $0 \leq t \leq T$ ,  $-\tau \leq \theta < 0$ ,  $\xi = 0$

(this also covers the case  $0 \le t \le T$ ,  $\theta = 0$ ,  $-\tau \le \xi < 0$ , since clearly  $P(t, \theta, \xi) = P'(t, \xi, \theta)$ ).

(iii) 
$$0 < t < T$$
,  $\theta = \xi = 0$ .

Since the derivations are somewhat tedious, we shall give details only for the equation satisfied by  $P(t,\theta,\xi)$  for  $0 \le t \le T$ ,  $-\tau \le \theta < 0$ ,  $-\tau \le \xi < 0$ . To avoid notational complications (since  $P(t,\theta,\xi)$  is a matrix, to use Theorem 5.3.1 requires the calculation of  $P_{ij}(t,\theta,\xi)$ , 1  $\le i,j \le n$ ), we will assume everything is scalar. For convenience we shall write  $e_c(x_s,s) = c(x_s,s) - \hat{c}(x_s,s)$ 

We appeal to Theorem 5.3.1 and (5.50) to write

$$P(t,\theta,\xi) = E^{t} \{x(t+\theta)x(t+\xi)\} - \hat{x}(t+\theta|t)\hat{x}(t+\xi|t)$$
$$= E_{0} \{x(t+\theta)x(t+\xi)\} + \int_{0}^{t} E^{s} \{E_{0}[x(t+\theta)x(t+\xi)|x_{s}]e_{0}(x_{s},s)]\}$$
$$\cdot R^{-1}(s)dv(s)$$

$$\begin{split} &-[\hat{x}(t+\theta|t+\theta)+\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)] [\hat{x}(t+\xi|t+\xi) \\ &+\int_{t+\xi}^{t} E^{S}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)] \\ &= E_{o}[x(t+\theta)x(t+\xi)] - \hat{x}(t+\theta|t+\theta)\hat{x}(t+\xi|t+\xi) \\ &+\int_{0}^{t} E^{S}\{E_{o}[x(t+\theta)x(t+\xi)|x_{s}]e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &-\hat{x}(t+\theta|t+\theta)\int_{t+\xi}^{t} E^{S}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &-\hat{x}(t+\xi|t+\xi)\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &-\hat{x}(t+\xi|t+\xi)\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &-\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t+\xi}^{t} E^{S}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &-\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t+\xi}^{t} E^{S}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &=\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t+\xi}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &=\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t+\xi}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &=\int_{t+\theta}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \\ &=\int_{t+\xi}^{t} E^{S}\{x(t+\theta)e_{c}(x_{s},s)\}$$

It is readily seen from (B.1) that  $P(t,\theta,\xi)$  is continuous in  $(t,\theta,\xi)$ . Similarly for  $\varepsilon$  in  $(0,\tau)$  such that  $-\tau \leq \theta + \varepsilon \leq 0$ ,  $-\tau \leq \xi + \varepsilon \leq 0$ ,  $P(t-\varepsilon, \theta+\varepsilon, \xi+\varepsilon)$ 

$$= E_0 \{x(t+\theta)x(t+\xi)\} - \hat{x}(t+\theta|t+\theta)\hat{x}(t+\xi|t+\xi)$$

$$+ \int_0^{t-\varepsilon} E^s \{E_0 \{x(t+\theta)x(t+\xi)|x_s\} e_c(x_s,s)\} R^{-1}(s) dv(s)$$

$$- \hat{x}(t+\theta|t+\theta) \int_{t+\xi}^{t-\varepsilon} E^s \{x(t+\xi)e_c(x_s,s)\} R^{-1}(s) dv(s)$$

$$-\hat{\mathbf{x}}(t+\xi|t+\xi) \int_{t+\theta}^{t-\varepsilon} \mathbf{E}^{\mathbf{s}} \{\mathbf{x}(t+\theta) \mathbf{e}_{\mathbf{c}}(\mathbf{x}_{\mathbf{s}},\mathbf{s})\} \mathbf{R}^{-1}(\mathbf{s}) d\mathbf{v}(\mathbf{s})$$

$$-\int_{t+\theta}^{t-\varepsilon} E^{s}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)d\nu(s)\int_{t+\xi}^{t-\varepsilon} E^{s}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)d\nu(s)$$

Hence 
$$P(t,\theta,\xi) - P(t-\varepsilon, \theta+\varepsilon, \xi+\varepsilon)$$

$$= \int_{t-\varepsilon}^{t} E^{s} \{ E_{o}[x(t+\theta)x(t+\xi) | x_{s}] e_{c}(x_{s},s) \} R^{-1}(s) dv(s)$$

$$-\hat{\mathbf{x}}(t+\theta|t+\theta) \int_{t-\varepsilon}^{t} \mathbf{E}^{s} \{\mathbf{x}(t+\xi)\mathbf{e}_{c}(\mathbf{x},s)\} \mathbf{R}^{-1}(s) d\mathbf{v}(s)$$

$$-\hat{\mathbf{x}}(t+\xi|t+\xi) \int_{t-\varepsilon}^{t} \mathbb{E}^{s} \{\mathbf{x}(t+\theta) \mathbf{e}_{c}(\mathbf{x}_{s},s)\} \mathbb{R}^{-1}(s) d\mathbf{v}(s)$$

$$-\int_{t+\theta}^{t} E^{s}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)d\nu(s)\int_{t+\xi}^{t} E^{s}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)d\nu(s)$$

$$+ \int_{t+\theta}^{t-\varepsilon} E^{s} \{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) \int_{t+\xi}^{t-\varepsilon} E^{s} \{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$
(B.2)

The last two terms of (B.2) can be written as

$$-\int_{t-\varepsilon}^{t} E^{s}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t+\xi}^{t} E^{s}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$

$$-\int_{t+\theta}^{t-\varepsilon} E^{s}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t-\varepsilon}^{t} E^{s}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$

$$= -\int_{t-\varepsilon}^{t} E^{s}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t-\varepsilon}^{t} E^{s}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$
  
$$-\int_{t-\varepsilon}^{t} E^{s}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t+\xi}^{t-\varepsilon} E^{s}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$
  
$$-\int_{t+\theta}^{t-\varepsilon} E^{s}\{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)\int_{t-\varepsilon}^{t} E^{s}\{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$
  
(B.3)

Substituting (B.3) into (B.2) and using properties of Ito integrals, we get

$$P(t,\theta,\xi) - P(t-\varepsilon, \theta+\varepsilon, \xi+\varepsilon)$$

$$= \int_{t-\varepsilon}^{t} E^{s} \{E_{o}[x(t+\theta)x(t+\xi)|x_{s}]e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$

$$-\hat{x}(t+\theta|t-\varepsilon) \int_{t-\varepsilon}^{t} E^{s} \{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$

$$-\hat{x}(t+\xi|t-\varepsilon) \int_{t-\varepsilon}^{t} E^{s} \{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$

$$- \int_{t-\varepsilon}^{t} E^{s} \{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s) \int_{t-\varepsilon}^{u} E^{u} \{x(t+\xi)e_{c}(x_{u},u)\}R^{-1}(u)dv(u)dv(s)$$

$$- \int_{t-\varepsilon}^{t} \int_{t-\varepsilon}^{s} E^{u} \{x(t+\theta)e_{c}(x_{u},u)\}R^{-1}(u)E^{s} \{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(u)dv(s)$$

$$-\int_{t-\varepsilon}^{t} E^{s} \{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)E^{s} \{x(t+\xi)e_{c}(x_{s},s)\}ds$$

$$=\int_{t-\varepsilon}^{t} E^{s} \{E_{o}[x(t+\theta)x(t+\xi)|x_{s}]e_{c}(x_{s},s)]R^{-1}(s)dv(s)$$

$$-\int_{t-\varepsilon}^{t} \hat{x}(t+\xi|s)E^{s} \{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$

$$-\int_{t-\varepsilon}^{t} \hat{x}(t+\theta|s)E^{s} \{x(t+\xi)e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$

$$-\int_{t-\varepsilon}^{t} E^{s} \{x(t+\theta)e_{c}(x_{s},s)\}R^{-1}(s)E^{s} \{x(t+\xi)e_{c}(x_{s},s)\}ds \qquad (B.4)$$

Since for each  $\theta \in [-\tau, 0]$ ,  $x(t+\theta)$  given  $z^t$  is Gaussian, the first three terms of (B.4) add to zero, being the third central moment of a Gaussian process. Hence

$$P(t,\theta,\xi) - P(t-\varepsilon, \theta+\varepsilon, \xi+\varepsilon)$$
  
=  $-\int_{t-\varepsilon}^{t} E^{S} \{x(t+\theta)e_{c}(x,s)\}R^{-1}(s)E^{S} \{x(t+\xi)e_{c}(x,s)\}ds$  (B.5)

Note that there are no random terms in (B.5). Since (B.5) holds for arbitrary  $\varepsilon$  in  $(0,\tau)$  with  $-\tau \leq \theta + \varepsilon \leq 0$ ,  $-\tau \leq \xi + \varepsilon \leq 0$ , we can divide by  $\frac{1}{\varepsilon}$  and let  $\varepsilon$  go to 0. If we let  $\sigma$  be the unit vector in the direction (1, -1, -1), then the directional derivative,  $P_{\sigma}(t,\theta,\xi)$ of P(t, $\theta,\xi$ ) in the direction  $\sigma$  is given by

$$\sqrt{3} P_{\sigma}(t,\theta,\xi) = -E^{t} \{x(t+\theta)e_{c}(x_{t},t)\}R^{-1}(t)E^{t} \{x(t+\xi)e_{c}(x_{t},t)\}$$
 (B.6)

Equation (B.6) is the desired equation for  $P(t, \theta, \xi)$ ,  $0 \le t \le T$ ,  $-\tau \le \theta < 0$ ,  $-\tau \le \xi < 0$ . The same technique can be applied to the derivation of the equation for  $P(t, \theta, 0)$ ,  $0 \le t \le T$ ,  $-\tau \le \theta < 0$ , and that for P(t, 0, 0). We only sketch the steps, omitting the details.

By Theorem 5.3.1,

$$P(t,\theta,0) = E_{0}[x(t+\theta)x(t)] + \int_{0}^{t} E^{s} \{E_{0}[x(t+\theta)x(t)|x_{s}]e_{c}(x_{s},s)\}R^{-1}(s)dv(s)$$
$$-\hat{x}(t+\theta|t)\hat{x}(t|t) \qquad (B.7)$$

(B.7) implies that  $P(t,\theta,0)$  is continuous in  $(t,\theta)$ . Some calculations yield, for  $\varepsilon$  in  $(0,\tau)$  such that  $-\tau \leq \theta + \varepsilon \leq 0$ ,  $P(t,\theta,0) - P(t-\varepsilon, \theta+\varepsilon, 0)$ 

$$= \int_{t-\varepsilon}^{\varepsilon} E^{s} \{x(t+\theta)a(x_{s},s)\} ds + \int_{t-\varepsilon}^{t} E^{s} \{x(t+\theta)x(s)e_{c}(x_{s},s)\}R^{-1}(s)dv(s) - \hat{x}(t+\theta|t)\hat{x}(t|t) + \hat{x}(t+\theta|t-\varepsilon)\hat{x}(t-\varepsilon|t-\varepsilon)$$
(B.8)

The last two terms of (B.8) can be evaluated to give

$$-\hat{\mathbf{x}}(\mathbf{t}+\boldsymbol{\theta}|\mathbf{t})\hat{\mathbf{x}}(\mathbf{t}|\mathbf{t})+\hat{\mathbf{x}}(\mathbf{t}+\boldsymbol{\theta}|\mathbf{t}-\varepsilon)\hat{\mathbf{x}}(\mathbf{t}-\varepsilon|\mathbf{t}-\varepsilon)$$

$$= -\int_{\mathbf{t}-\varepsilon}^{\mathbf{t}}\hat{\mathbf{x}}(\mathbf{t}+\boldsymbol{\theta}|\mathbf{s})\hat{\mathbf{a}}(\mathbf{x}_{s},\mathbf{s})d\mathbf{s} - \int_{\mathbf{t}-\varepsilon}^{\mathbf{t}}\hat{\mathbf{x}}(\mathbf{t}+\boldsymbol{\theta}|\mathbf{s})\mathbf{E}^{\mathbf{s}}\{\mathbf{x}(\mathbf{s})\mathbf{e}_{c}(\mathbf{x}_{s},\mathbf{s})\}\mathbf{R}^{-1}(\mathbf{s})d\mathbf{v}(\mathbf{s})$$

$$-\int_{\mathbf{t}-\varepsilon}^{\mathbf{t}}\mathbf{E}^{\mathbf{s}}\{\mathbf{x}(\mathbf{t}+\boldsymbol{\theta})\hat{\mathbf{x}}(\mathbf{s}|\mathbf{s})\mathbf{e}_{c}(\mathbf{x}_{s},\mathbf{s})\}\mathbf{R}^{-1}(\mathbf{s})d\mathbf{v}(\mathbf{s})$$

$$-\int_{\mathbf{t}-\varepsilon}^{\mathbf{t}}\mathbf{E}^{\mathbf{s}}\{\mathbf{x}(\mathbf{t}+\boldsymbol{\theta})\mathbf{e}_{c}(\mathbf{x}_{s},\mathbf{s})\}\mathbf{R}^{-1}(\mathbf{s})\mathbf{E}^{\mathbf{s}}\{\mathbf{x}(\mathbf{s})\mathbf{e}_{c}(\mathbf{x}_{s},\mathbf{s})\}d\mathbf{s} \qquad (B.9)$$

Substituting (B.9) into (B.8) and using again the Gaussian property, we see that

$$P(t,\theta,0) - P(t-\varepsilon, \theta+\varepsilon,0)$$

$$= \int_{t-\varepsilon}^{t} E^{s} \{x(t+\theta)e_{a}(x_{s},s)\} ds - \int_{t-\varepsilon}^{t} E^{s} \{x(t+\theta)e_{c}(x_{s},s)\} R^{-1}(s) \cdot E^{s} \{x(s)e_{c}(x_{s},s)\} ds \qquad (B.10)$$

Again there are no random terms so that  $P(t,\theta,0)$  has a directional derivative in the (1,-1) direction. Denoting the unit vector in the (1,-1) direction by  $\eta$ , we obtain

$$\sqrt{2} P_{\eta}(t,\theta,0) = E^{t} \{x(t+\theta)e_{a}(x_{t},t)\}$$
$$-E^{t} \{x(t+\theta)e_{c}(x_{t},t)\}R^{-1}(t)E^{t} \{x(t)e_{c}(x_{t},t)\}$$
(B.11)

Finally, similar calculations (or using (5.42)) show that P(t,0,0) has a derivative and satisfies

$$\frac{d}{dt} P(t,0,0) = E^{t} \{x(t)e_{a}(x_{t},t)\} + E^{t} \{e_{a}(x_{t},t)x(t)\} -E^{t} \{x(t)e_{c}(x_{t},t)\}R^{-1}(t)E\{e_{c}(x_{t},t)x(t)\} + Q(t)$$
(B.12)

In the vector case, equations identical to those of (3.6), (B.11), and (B.12) hold for  $P(t,\theta,\xi)$ ,  $P(t,\theta,0)$ , and P(t,0,0). This completes the derivation of equations (5.53) - (5.55). The initial conditions (5.56) follow immediately from the properties of conditional expectations.

# APPENDIX C

# PROOF OF LEMMA 6.6.1

We apply the Ito differential rule to the function  $V(x_t)$  defined in (6.78). We calculate the first and second terms to illustrate the computations involved.

$$dx'(t)K_{0}(t)x(t) = [dx'(t)]K_{0}(t)x(t)dt +x'(t)[dK_{0}(t)]x(t)dt+x'(t)K_{0}(t)[dx(t)]dt+trM'MK_{0}(t)dt = x'(t-\tau)B'K_{0}(t)x(t)dt+u'(t)G'K_{0}(t)x(t)dt +dw'(t)K_{0}(t)x(t)dt+x'(t)K_{0}(t)Bx(t-\tau)dt+x'(t)K_{0}(t)Gu(t)dt +x'(t)K_{0}(t)dw(t)-x'(t)Qx(t)dt+x'(t)K_{0}(t)GR^{-1}G'K_{0}(t)x(t)dt -x'(t)K_{1}'(t,0)x(t)dt-x'(t)K_{1}(t,0)x(t)dt+trM'MK_{0}(t)dt dt \int_{-\tau}^{0} x'(t)K_{1}(t,\theta)x(t+\theta)d\theta = d_{t}[x'(t)\int_{t-\tau}^{t} K_{1}(t,\sigma-t)x(\sigma)d\sigma] = \{[x'(t)A'+x'(t-\tau)B'+u'(t)G']dt+dw'(t)\}\int_{-\tau}^{0} K_{1}(t,\theta)x(t+\theta)d\theta +x'(t)K_{1}(t,0)x(t)dt-x'(t)K_{1}(t,-\tau)x(t-\tau)dt +x'(t)\int_{t-\tau}^{t} d_{t}K_{1}(t,\sigma-t)x(\sigma)d\sigma dt$$

$$= \left[x'(t)A'+x'(t-\tau)B'+u'(t)G'\right] \int_{-\tau}^{0} K_{1}(t,\theta)x(t+\theta)d\theta dt$$
  
+dw'(t)  $\int_{-\tau}^{0} K_{1}(t,\theta)x(t+\theta)d\theta+x'(t)K_{1}(t,0)x(t)dt$   
-x'(t)K\_{1}(t,-\tau)x(t-\tau)dt+x'(t)  $\int_{-\tau}^{0} \left[\left(\frac{\partial}{\partial t} - \frac{\partial}{\partial \theta}\right) K_{1}(t,\theta)\right]x(t+\theta)d\theta dt$ 

Similar calculations on the last two terms on the right hand side of (6.78) yields the following expression

$$dV(x_{t}) = V_{1}(t)dt + dw'(t)K_{0}(t)x(t) + x'(t)K_{0}(t)dw(t) + trM'MK_{0}(t)dt + dw'(t) \int_{-T}^{0} K_{1}(t,\theta)x(t+\theta)d\theta + \int_{-T}^{0} x'(t+\theta)K_{1}'(t,\theta)d\theta dw(t) - x'(t)Qx(t)dt - u'(t)Ru(t)dt$$
(C.1)  
where  $V_{1}(t) = [u(t) + R^{-1}G'K_{0}(t)x(t) + \int_{-T}^{0} R^{-1}G'K_{1}(t,\theta)x(t+\theta)d\theta]' \cdot R[u(t) + R^{-1}G'K_{0}(t)x(t) + \int_{-T}^{0} R^{-1}G'K_{1}(t,\xi)x(t+\xi)d\xi]$ 

Using the boundary conditions at T for  $K_0(t)$ ,  $K_1(t,\theta)$ , and  $K_2(t,\theta,\xi)$ , we see that  $V(x_T) = 0$ . Therefore

$$\int_{0}^{T} [x'(t)Qx(t)+u'(t)Ru(t)]dt$$

$$= V(x_{o}) + \int_{0}^{T} V_{1}(t)dt + \int_{0}^{T} trM'MK_{o}(t)dt$$

$$+ 2 \int_{0}^{T} x'(t)K_{o}(t)dw(t) + 2 \int_{0}^{T} \int_{-\tau}^{0} x'(t+\theta)K_{1}'(t,\theta)d\theta dw(t)$$

Taking expectations, we get

$$E \int_{0}^{T} [x'(t)Qx(t)+u'(t)Ru(t)]dt$$
  
=  $EV(x_{0}) + E \int_{0}^{T} V_{1}(t)dt + \int_{0}^{T} trM'MK_{0}(t)dt$  (C.3)  
Now  $E \int_{0}^{T} V_{1}(t)dt = \int_{0}^{T} EV_{1}(t)dt$   
=  $\int_{0}^{T} E\{E[V_{1}(t)|z^{t}]\}dt$ 

using Fubini's theorem and properties of conditional expectations. Substituting the control law in (6.76) into (C.2) we get that

$$\begin{split} \mathbb{E}[\mathbb{V}_{1}(t)|z^{t}] &= \mathbb{E}\{[\mathbb{R}^{-1}G'\mathbb{K}_{o}(t)e(t|t) + \int_{-\tau}^{0} \mathbb{R}^{-1}G'\mathbb{K}_{1}(t,\theta)e(t+\theta|t)d\theta]' \\ &\cdot \mathbb{R}[\mathbb{R}^{-1}G'\mathbb{K}_{o}(t)e(t|t) + \int_{-\tau}^{0} \mathbb{R}^{-1}G'\mathbb{K}_{1}(t,\theta)e(t+\theta|t)d\theta]|z^{t}\} \end{split}$$

which is precisely

$$tr\{K_{o}(t)GR^{-1}G'K_{o}(t)P_{o}(t)+\int_{-\tau}^{0}K_{1}'(t,\theta)GR^{-1}G'K_{o}(t)P_{1}'(t,\theta)d\theta$$
$$+\int_{-\tau}^{0}K_{o}(t)GR^{-1}G'K_{1}(t,\theta)P_{1}(t,\theta)d\theta$$
$$+\int_{-\tau}^{0}\int_{-\tau}^{0}K_{1}'(t,\theta)GR^{-1}G'K_{1}(t,\xi)P_{2}(t,\xi,\theta)d\thetad\xi\} \qquad (C.4)$$

 $E\{V_1(t)|z^t\}$  is now seen to be a deterministic function and hence equal to  $EV_1(t)$ . Substituting (C.4) into (C.3) yields the conclusion of the lemma.

#### REFERENCES

- R. Bellman and K.L. Cooke, "Differential-Difference Equations," Academic Press, New York, 1963.
- [2] J.K. Hale, "Functional Differential Equations," Springer-Verlag, New York, 1971.
- [3] R.D. Driver, "Existence and stability of solutions of a delaydifferential system," Arch. for Rat. Mech., Vol. 10, 1962, pp. 401-426.
- [4] L. Euler, "Investigatio curvarum quae evalutae sui similis producunt," Commentarii Academiae Scientarum Imperalis Petropolitanae, Vol. 20, 1750, pp. 3-52.
- [5] V. Volterra, "Sur la theorie mathematique des phenomes hereditaires," Journal Mathematique Pures et Appliquees, ser. 9, Vol. 7, 1928, pp. 249-298.
- [6] N.N. Krasovskii, "Stability of Motion," Stanford University Press, Stanford, 1963.
- [7] N.H. Choksy, "Time-lag systems a bibliography," IRE Trans. on Auto. Control, Vol. 5, 1960, pp. 66-70.
- [8] A. Manitius, "Mathematical models of hereditary systems," Res. Rep., Centre de Recherches Mathématiques, Université de Montréal, Nov. 1974.
- [9] H.T. Banks and A. Manitius, "Application of abstract variational theory to hereditary systems - a survey," IEEE Trans. Auto. Control, Vol. AC-19, Oct. 1974, pp. 524-533.
- [10] D.W. Ross, "Controller design for time lag systems via a quadratic criterion," IEEE Trans. Auto. Control, Vol. AC-16, Dec. 1971, pp. 664-672.
- [11] H.L. Van Trees, "Detection, Estimation, and Modulation Theory, Part III," John Wiley, New York, 1971.
- [12] J.A. Yorke, "Selected topics in differential-delay equations," in Lecture Notes in Mathematics, No. 243, Springer-Verlag, Berlin, 1971, pp. 16-28.
- [13] N.K. Gupta and R.E. Rink, "A model for communicable disease control," in Proc. 24th Annual Conf. Engineering in Medicine and Biology, Las Vegas, Nevada, 1971, p. 296.
- [14] M.A. Cruz and J.K. Hale, "Existence, uniqueness and continuous dependence for hereditary systems," Annali di Mat. Pura Appl., Vol. 85, No. 4, 1970, pp. 63-82.

- [15] V.M. Popov, "Pointwise degeneracy of linear time-invariant, delaydifferential equations," J. Diff. Eqs., Vol. 11, 1972, pp. 541-561.
- [16] A.M. Zverkin, "On pointwise completeness of systems with lag," Differentsial'nye Uravneniya, Vol. 9, 1973, pp. 430-436.
- [17] R.B. Zmood and N.H. McClamroch, "On the pointwise completeness of differential-difference equations," J. Diff. Eqs., Vol. 12, 1972, pp. 474-486.
- [18] B. Asner and A. Halanay, "Algebraic theory of pointwise degenerate delay-differential systems," J. Diff. Eqs., Vol. 14, 1973, pp. 293-306.
- [19] A.K. Choudhury, "Necessary and Sufficient conditions of pointwise completeness of linear time-invariant delay-differential systems," Int. J. Control, Vol. 16, 1972, pp. 1083-1100.
- [20] F. Kappel, "Some remarks to the problem of degeneracy for functional differential equations," in "Equation differentielles et fonctionnelles," eds. P. Jansens, J. Mawhin, and N. Rouché, Hermann, Paris, 1974.
- [21] R.W. Brockett, "Finite Dimensional Linear Systems," John Wiley, New York, 1970.
- [22] W.M. Wonham, "On a matrix Riccati equation of stochastic control," SIAM J. Control, Vol. 6, 1968, pp. 681-697.
- [23] F.M. Kirillova and S.V. Churakova, "On the problem of controllability of linear systems with aftereffect," Differentsial'nye Uravneniya, Vol. 3, 1967, pp. 436-445.
- [24] F.M. Kirillova and S.V. Churakova, "Relative controllability of systems with time lag," Dokl. Acad. Nauk, USSR, Vol. 176, 1967, pp. 1260-1263.
- [25] L. Weiss, "On the controllability of delay-differential systems," SIAM J. Control, Vol. 5, 1967, pp. 575-587.
- [26] L. Weiss, "Lectures on Controllability and Observability," CIME Course on Controllability and Observability, Rome, 1969, pp. 205-289.
- [27] A. Manitius, "On controllability conditions for systems with distributed delays in state and control," Archiwum Autom. i Telem. Vol. 17, 1972, pp. 363-377.
- [28] R.B. Zmood, "The Euclidean space controllability of control systems with delay," SIAM J. Control, Vol. 12, Nov. 1974, pp. 609-623.

- [29] M.C. Delfour and S.K. Mitter, "Controllability, observability and optimal feedback control of affine hereditary differential systems," SIAM J. Control, Vol. 10, May 1972, pp. 298-328.
- [30] Yu S. Osipov, "Stabilization of controlled systems with delays," Differentsial'nye Uravneniya, Vol. 1, 1965, pp. 605-618.
- [31] H.F. Vandevenne, "Qualitative properties of a class of infinite dimensional systems," Doctoral dissertation, Electrical Engineering Department, M.I.T., Cambridge, Mass., 1972.
- [32] M.C. Delfour, C. McCalla, and S.K. Mitter, "Stability and the infinite-time quadratic cost problem for linear hereditary differential systems," SIAM J. Control, Vol. 13, Jan. 1975, pp. 48-88.
- [33] H.T. Banks, M.Q. Jacobs, and C.E. Langenhop, "Characterization of the controlled states in W<sub>2</sub> (1) of linear hereditary systems," SIAM J. Control, Vol. 13, May 1975, pp. 611-649.
- [34] A. Halanay, "On the controllability of linear difference-differential systems," in Lecture Notes in Operations Research and Mathematical Economics, 12, Mathematical Systems Theory and Economics II, ed. H.W. Kuhn and G.P. Szego, Springer-Verlag, Berlin, 1970, pp. 329-335.
- [35] A. Manitius, "Optimal control of hereditary systems," Lecture Notes for the course "Control Theory and Topics in Functional Analysis," International Center for Theoretical Physics, Miramare, Trieste, Italy, Sept. 11 - Nov. 29, 1974.
- [36] A. Halanay, "Differential Equations," Academic Press, New York, 1966.
- [37] D.H. Chyung and E.B. Lee, "Optimal systems with time delay," Proc. of the Third Congress of the International Federation of Automatic Control, London, 1966.
- [38] D.H. Chyung and E.B. Lee, "Linear optimal systems with time delays," SIAM J. Control, Vol. 4, 1966, pp. 548-575.
- [39] R. Triggiani, "Controllability and observability in Banach space with bounded operators," SIAM J. Control, Vol. 13, Feb. 1975, pp. 462-491.
- [40] M. Slemrod, "A note on complete controllability and stabilizability for linear control systems in Hilbert space," SIAM J. Control, Vol. 12, Aug. 1974, pp. 500-508.
- [41] R. Triggiani, "On the stabilizability problem in Banach space," to appear in J. Math. Anal. Appl.

- [42] A.S. Morse, "Ring models for delay-differential systems," to appear.
- [43] E.W. Kamen, "Module structure of infinite-dimensional systems with applications to controllability," to appear in SIAM J. Control.
- [44] V.M. Popov, "Delay-feedback, time-optimal, linear time-invariant control systems," in Ordinary Differential Equations, 1971 NRL-MRC Conference, ed. L. Weiss, Academic Press, New York, 1972.
- [45] P.R. Halmos, "Finite-Dimensional Vector Spaces," D. Van Nostrand, Princeton, 1958.
- [46] C.T. Chen, "Introduction to Linear System Theory," Holt, Rinehart and Winston, New York, 1970.
- [47] E.B. Lee and L. Markus, "Foundations of Optimal Control Theory," John Wiley, New York, 1967.
- [48] D. Henry, "Small solutions of linear autonomous functional differential equations," J. Diff. Eqs., Vol. 8, 1970, pp. 494-501.
- [49] P. Lancaster, "Theory of Matrices," Academic Press, New York, 1969.
- [50] J.L. Doob, "Stochastic Processes," John Wiley, New York, 1953.
- [51] M. Fujisaki, G. Kallianpur, and H. Kunita, "Stochastic differential equations for the nonlinear filtering problem," Osaka J. Math., Vol. 9, 1972, pp. 19-40.
- [52] K. Ito and M. Nisio, "On stationary solutions of a stochastic differential equation," Kyoto J. Math., Vol. 4, 1964, pp. 1-75.
- [53] W.H. Fleming and M. Nisio, "On the existence of optimal stochastic controls," J. Math. Mech., Vol. 15, 1966, pp. 777-794.
- [54] H.J. Kushner, "On the stability of processes defined by stochastic differential-difference equations," J. Diff. Eqs., Vol. 4, 1968, pp. 424-443.
- [55] R.H. Cameron and W.T. Martin, "An unsymmetric Fubini theorem," Bull. Amer. Math. Soc., Vol. 47, 1941, pp. 121-125.
- [56] A. Lindquist, "Optimal control of linear stochastic systems with applications to time lag systems," Information Sciences, Vol. 5, 1973, pp. 81-126.
- [57] H.T. Banks, "Representations for solutions of linear functional differential equations," J. Diff. Eqs., Vol. 5, 1969, pp. 399-409.

- [58] E. Wong, "Stochastic Processes in Information and Dynamical Systems," McGraw-Hill, New York, 1971.
- [59] C. Doléans-Dade, "Quelques applications de la formule de changement de variables pour les semimartingales," Z. Wahrscheinlichkeits. verw. Geb., Vol. 16, 1970, pp. 181-194.
- [60] T.E. Duncan, "Probability densities for diffusion processes with applications to nonlinear filtering theory and detection theory," Ph.D. dissertation, Dep. Elec. Eng., Stanford Univ., Stanford, Calif., 1967.
- [61] M. Loève, "Probability Theory," 3rd edition, D. Van Nostrand, Princeton, 1963.
- [62] M. Zakai, "On the optimal filtering of diffusion processes,"
   Z. Wahrscheinlichkeitstheorie verw. Geb., Vol. 11, 1969, pp. 230-243.
- [63] K. Ito, "On a formula concerning stochastic differentials," Nagoya Math. J., Vol. 3, 1951, pp. 55-65.
- [64] E.B. Dynkin, "Markov Processes," Springer-Verlag, Berlin, 1965.
- [65] H. Kwakernaak, "Optimal filtering in linear systems with time delays," IEEE Trans. Auto. Control, Vol. AC-12, April 1967, pp. 169-173.
- [66] R.E. Kalman and R.S. Bucy, "New results in linear filtering and prediction theory," J. Basic Engin., Vol. 83, March 1961, pp. 95-108.
- [67] S.K. Mitter and R.B. Vinter, "Filtering for linear stochastic hereditary differential systems," Intern. Symp. Control Theory, Numerical Methods and Computer Systems Modeling, I.R.I.A., Rocquencourt, France, June 1974.
- [68] A. Lindquist, "On feedback control of linear stochastic systems," SIAM J. Control, Vol. 11, May 1973, pp. 323-343.
- [69] H.J. Kushner and D.I. Barnea, "On the control of a linear functional differential equation with quadratic cost," SIAM J. Control, Vol. 8, 1970, pp. 257-272.
- [70] Y. Alekal, P. Brunovsky, D.H. Chyung and E.B. Lee, "The quadratic problem for systems with time delays, "IEEE Trans. Auto. Control, Vol. AC-16, Dec. 1971, pp. 673-688.
- [71] A. Manitius, "Optimal control of time-lag systems with quadratic performance indexes," IV Congress of IFAC, Warsaw, 1969, paper 13.2.

- [72] R. Datko, "Unconstrained control problems with quadratic cost," SIAM J. Control, Vol. 11, 1973, pp. 32-52.
- [73] M.C. Delfour, "Numerical solution of the optimal control problem for linear hereditary differential systems with a linear-quadratic cost function and approximation of the Riccati differential equation, "Research Report CRM-408, Centre de Recherches Mathématiques, Université de Montréal, 1974. See also ref. [29].
- [74] J.K. Hale, "Sufficient conditions for stability and instability of autonomous functional-differential equations," J. Diff. Eqs., Vol. 1, 1965, pp. 452-482.
- [75] A. Lindquist, "A theorem on duality between estimation and control for linear stochastic systems with time delay," J. Math. Anal. Appl., Vol. 37, 1972, pp. 516-536.
- [76] N. Levinson and C. McCalla, "Completeness and independence of the exponential solutions of some functional differential equations," Studies in Appl. Math., Vol. 53, 1974, pp. 1-16.
- [77] H.T. Banks and A. Manitius, "Projection series for functional differential equations with applications to optimal control problems," to appear in J. Diff. Eqs., 1975.
- [78] F. Riesz and B. Sz.-Nagy, "Functional Analysis," Frederick Unger Publ. Co., New York, 1955.
- [79] W.M. Wonham, "On the Separation Theorem of stochastic control," SIAM J. Control, Vol. 6, 1968, pp. 312-326.
- [80] H. Koivo and E.B. Lee, "Controller synthesis for linear systems with retarded state and control variables and quadratic cost," Automatica, Vol. 8, 1972, 203-208.

#### BIOGRAPHICAL NOTE

Raymond Hon-Sing Kwong was born in Hong Kong on September 20, 1949. He attended Wah Yan College in Kowloon and matriculated in June 1968. He was school captain from September 1966 to June 1967. He entered M.I.T. in September 1968 and received the degree of Bachelor of Science from the Department of Electrical Engineering in June 1971. He was elected to membership in Tau Beta Pi, Eta Kappa Nu, and associate membership in Sigma Xi. While an undergraduate, he was active in various student activities, being President of the International Students Council in the academic year 1970-1971.

Mr. Kwong has been a graduate student in the Department of Electrical Engineering at M.I.T. since June 1971. He was a research assistant from June 1971 to June 1972, a teaching assistant from September 1972 June 1973, a research assistant from June 1973 to September 1973, a department fellow supported by the Grass Instrument Company fellowship from September 1973 to September 1974, and a research assistant from September 1974 to the present time. He was awarded the degree of Master of Science in June 1972.

During summers, Mr. Kwong has worked for Stone and Webster Engineering Corporation (1969), and Bell Telephone Laboratories at Whippany (1972).

Mr. Kwong is married to the former Shirley San-Yee Tsang.

-204-